'FINDING A VOICE'

THE EXPERIENCE OF MATURE STUDENTS IN A COLLEGE OF FURTHER EDUCATION

FERGAL HARDIMAN

Education Doctorate

National University of Ireland Maynooth Faculty of Social Sciences

Education Department

Head of Department: Dr Aidan Mulkeen

Department of Adult and Community Education

Head of Department: Josephine Finn

Supervisor: Dr Ted Fleming

May 2012

Acknowledgements

I would like to acknowledge most gratefully the teaching staff on the Ed.D programme for their professionalism, support and creativity. I would like to thank particularly Dr Anne B. Ryan and Dr Rose Malone. I am very privileged to have had them as teachers.

I was extremely fortunate also in my supervisor, Dr Ted Fleming, who was unfailingly insightful, supportive, encouraging and practical in his analysis. He inspired and encouraged me throughout this work.

My classmates were also a great support and many phone phone calls were made and very gratefully received that offered encouragement and advice. I learned a great deal from them and I am very grateful for their friendship.

I would like to acknowledge the honesty and generosity of the mature students who participated, directly or indirectly in this research. Their willingness to share their stories and reflections made the work fascinating and compelling. I am very grateful to each of them.

I would like to acknowledge my wife, Margaret and my children, Naoise, Saidhbhín and Lúach and my grandchildren who allowed me the time and space to do this work. Thank you to all.

Abstract

This study explores the experience of mature students in a Further Education college in Ireland. It looks at the development of the further education sector and its current role in Irish education. Further education is a major provider of education for adults as more than half the students in further education are now mature students. The system grew largely within the Vocational Education system and, it is suggested, has adopted many of the neo-liberal ideas and practices prevalent in society more generally.

The first research element was a survey and focus group, followed by nine qualitative interviews which explored with the students their reasons for returning to education, their experience while in further education and what they consider to be the benefits of participation. Using a constructivist grounded theory approach the data were generated and analysed through a systematic process of coding, categorisation and the development of core concepts grounded in the students' lived experience. Three significant concepts emerged from the analysis. The first concept of 'The latent self' describes the range of social, personal, structural and biographical influences and discusses how they interact and impact on the decision to return to education. The second concept, 'The Emerging Self' describes the negotiation of challenges faced by the students in the process of change. The third concept, 'The Revised Self', describes the changed self emphatically described by the students as resulting from their participation and characterised by greatly enhanced confidence and agency.

The conceptual label 'Finding a voice' was used to describe this 'Revised Self' as many of the participants spoke in terms of feeling able to speak and having things to say. It encompasses willingness and a sense of entitlement to participate and engage with others and a sense of having something to contribute.

The findings demonstrate that further education is a site of significant learning for adults. Although the purpose of the further education sector is to provide students with the skills needed in the workplace and many students came for vocational reasons, the actual learning went beyond the instrumental and resulted in many surprising benefits to the students and to society more generally.

CONTENTS

Introduction		1
CHAPTER 1	FURTHER EDUCATION IN IRELAND	
Introduction		7
Understanding Further Education		7
Background to Further Ed	lucation in Ireland	10
Vocational Education		12
Access Courses		13
VTOS and BTEA		15
Mature Students		16
Further Education Now		19
Conclusion		21
CHAPTER 2	CONCEPTUAL FRAMEWORK	
Introduction		23
Neo-liberalism		23
The European context		25
Neo-liberalism and Furthe	r Education in Ireland	27
Outcome based systems		29
The discourse of Further E	Education	35
Critical and transformative theorists		36
Habermas		36
Mezirow		39
Bourdieu		40
Conclusion		43
CHAPTER 3	REVIEW OF SELECTED LITERATURE	
Introduction		44
The McIver Report		45
The Post-Leaving Certifica	nte Sector in Ireland	46
Hidden Disadvantage		48
Where Next?		51
Ranlhe		54
College Knowledge		57
Conclusion		58

CHAPTER 4 METHODOLOGY

Introduction		59			
Epistemology Qualitative methodology Grounded Theory Research Procedure-data collection Questionnaire					
			Focus Group		68
			Interviews		69
			Ethical Issues	72	
			Insider research Data Analysis		
Initial coding		76			
Focused coding		78			
Theoretical cod	ding	79			
Memos		80			
Reflexivity		81			
Engaging with the literature		83			
Conclusion		84			
CHAPTER 5	FINDINGS				
CHAITER 3	FINDINGS				
Introduction		85			
The Questionnaire		85			
The Focus Group		91			
Profiles of participan	ts	95			
Conclusion		110			
CHAPTER 6	A THEMATIC VIEW OF THE FINDINGS				
Introduction		111			
Latent self		112			
The right time		112			
Always wanted to / missing out		113			
Needing a qual	lification	114			
Change		116			
Emerging Self		117			
Becoming a mature student		117			
Juggling comm	uitments	118 119			
	Fear of failure				
First assignment					
Keen to succeed		120			
Computer comp		121			
Benefits of bein	ıg older	122			
Support		123			
Enjoying it		123			

Revised Self- Finding voice Conclusion		125 127	
Conclusion		127	
CHAPTER 7	DISCUSSION		
Introduction		128	
The Latent Self		129	
Missing out		123	
Houle's Typology		132	
Barriers to participation		133	
Lifestage theories		135	
Moratorium		139	
Post-modern uncertainty		142	
Ahl on motivation		143	
Conclusion		144	
The Emerging Self		145	
Becoming a mature studen		146	
Home culture and college	синиге	147 148	
Juggling roles		150	
Am I up to it? Risk		150	
Attachment theory		151	
Recognition		156	
Walters' 3Rs		158	
Finding a Voice		160	
Conclusion		165	
Concression		100	
CHAPTER 8	CONCLUSION	166	
The findings		166	
Implications		168	
Limitations of the study		171	
Competing discourses		171	
A final thought		173	
REFERENCES		175	
APPENDICES		192	
TABLE OF FIGURES		204	

INTRODUCTION

I have been teaching in the Further Education (FE) sector in the City of Dublin Vocational Education Committee (CDVEC) for thirty years. The sector grew up within individual Vocational Education Committee (VEC) colleges from the early 1980s. Over the years it has become a significant part of Irish education and now has over 30,000 students, of which half are mature students. This means that FE is now a major provider of adult education. While it is generally recognised by teachers and managers in FE that mature students are different from more traditional students, little formal effort is made to accommodate their particular characteristics and needs.

This thesis sets out to explore the experiences of mature students in further education, why they come to FE and what they see as the benefits of participation.

In FE, a student aged twenty one or over is considered a mature student. Many younger mature students are similar in many ways to their colleagues a year or two their junior. This research focuses on mature students who have been out of full-time education for some years.

To help connect my experience with this exploration I will describe an incident that gave me a significant insight and made me think.

I teach media studies. In a media analysis class we were studying a print advertisement from the point of view of its construction, semiotics, narrative, audience and representation. The advertisement depicted a romantic scene and this led to a discussion of the nature of romance and whether romance was a natural or socially constructed phenomenon. We discussed different cultural, religious, commercial, ideological and historical perspectives and other mating practices and how these could be assessed. Later in the day I met a mature student from the class, Lisa, a local woman in her forties, and asked her how she was getting on. She said she loved being in the college but that she had found the class that morning 'very upsetting'. She went on to say that she had never thought about things like that before, just accepted them.

When I asked her if others in the class were upset, she said, 'Sure they're young, that wouldn't upset them.'

This encounter prompted me to consider how mature students are different from their younger classmates, why they come to a college of further education and what they get from it.

Despite the growth in FE in recent years there is little work done on the FE sector in Ireland. The one major review, The McIver Report (2003) was based on institutional analysis and examined 15 colleges from those with 150 or more students. This report outlined the range of provision in FE and called for the establishment of a separate FE sector. A Report in 2006, by the ESRI for the Department of education and Science (Watson, McCoy & Gorby, 2006) looked at Post-Leaving Certificate courses and Employment outcomes. There has been no attempt to explore student experience or views or to examine the implications of the growth in the numbers of mature students in FE. There are a small number of unpublished Masters' Theses that touch on various aspects of FE but none look at the experience of students, especially older students. This thesis is an attempt to address this imbalance.

There are a number of important studies on mature students in Higher Education in Ireland (Fleming and Murphy, 1997; Lynch, 1997; Inglis and Murphy, 1999) these studies predate the transformation of Ireland in the last decade. There are a number of more recent studies that have much to offer a study of FE which I will be looking at in another chapter.

Government policy in recent years has seen widening access to education as being central to promoting inclusion, employment and fostering national economic growth. There has been a commitment to widening participation in third level education and upgrading the qualifications of the workforce generally, especially among groups who traditionally were under-represented, including mature students. Further Education (FE) has been successful in attracting students from backgrounds traditionally under-represented in Higher Education (HE) and clearly has a role to play in providing a route back into education for those who were disaffected by their previous experience. FE offers progression, including access to HE to students whose lack of qualifications or participation in the Leaving Certificate Applied Programme (LCA) would have

precluded them from direct entry. For most, however, FE is an alternative to HE rather than a route to it, and its provision for mature students is an important part of attempts to combat inequality and exclusion.

This study is grounded in my own experience of working in FE for many years and my conviction that the curriculum accounts for only a small part of the learning that occurs, especially with mature students. I also believe that the emotional aspects of education are often neglected in considering how these students get on and how they facilitate the process of change that they often speak about.

The students in this study have come to further education at a time when the dominant discourse in education is instrumental and preparation for work is seen as the principal objective of FE colleges. Success is usually measured by the completion of awards and progression to employment or HE. This study explores with the students their reasons for coming to FE, their experience as mature students and what they see as the benefits of participation.

The students and teachers in FE operate in an environment with conflicting influences. On one side there is the neo-liberal context of marketisation, quality assurance and performativity and on the other the VEC history and tradition of inclusion, care and support. We are in a time of global uncertainty and economic recession with cutbacks in staffing and resources and more urgent demands being placed on education by increasing numbers of applicants and the expectations of government that we do more with less. In this climate it is important that we protect the values of the VEC system and affirm the contribution of FE to the education of adults. This thesis tries to contribute to that process. The research was carried out in the large college of further education where I work, referred to in the text as 'the college.'

I began my research with the general aim of exploring the experience of mature students in a Further Education College within the CDVEC.

There are a number of sub questions:

- Who are the mature students in FE? What is known of their educational socioeconomic and cultural background?
- Why do adults return to education? Why FE particularly?
- In what ways are they different to younger students?
- What helps or hinders their learning in FE? What can the FE sector and individual institutions do to promote effective learning and a positive experience?
- What do the students see as the benefits of participation?
- What are the implications of the findings for further education in Ireland?

The initial research involved a Focus Group and a questionnaire. The main focus was qualitative, using interviews with mature students and constructivist grounded theory methodology for data generation and analysis.

Structure of the thesis

The first three chapters establish the background and context of the work. Chapter one looks at the evolution of the FE sector in Ireland. It is suggested that the sector grew, not as the result of an overall plan, but in a haphazard fashion depending on the enthusiasm of individual principals and teachers. Vocational Education Committees (VECs) are the main providers of FE and their role in Irish education is an important factor considered in this chapter.

Chapter two considers the ideological climate in which FE and Irish education is operating. It suggests that in its practices and policies FE has adopted, by default, a neo-liberal agenda and considers some theorists who offer a way of understanding this development and addressing it.

Chapter three continues to set the context by looking at a selection of studies that contribute to an understanding of FE. This is done at this point 'to set the stage for what you do' (Charmaz, 2006, p.66). Their conceptual frameworks and methodological approaches were influential for this work and help to place FE in the broader context of Irish education. Engaging with a selection of the literature at this stage also contributed to a general theoretical sensitivity and awareness of broad issues and approaches. A more focused review of the literature was done when the research was carried out and categories and relationships between them emerged from working with the data.

Chapter four outlines the epistemological stance and the constructivist grounded theory approach of the study. It details the research procedure, the process of data generation and analysis and considers some ethical issues.

Chapter five looks at the findings from the questionnaire, the focus group and briefly profiles four of the participants and two others in more detail. Instead of dealing with the abstraction of the concept of 'mature student, these separate narratives introduce the individuals that make the concept real. Their experiences and feelings are the stuff of this study and their individual stories are, I think, vital for understanding mature students' experience.

Chapter six offers a more themed account of the participants' stories and outlines how the key ideas of Latent Self, Emerging Self and Revised Self emerged as core concepts.

Chapter seven engages in a discussion of the findings from a more abstract perspective and, in keeping with grounded theory procedures, engages with the literature relevant to these emerging findings.

Chapter 8 reviews the research questions and findings, looks at some of the implications and points out some of the limitations of the work and areas for further study.

Grounded theory.

The study set out to use grounded theory as a methodology for exploring the experience of mature students. The intention was to gain an understanding, not to generate theory. However, as the study progressed, theory emerged around the process of transformation through participation in education and the emergence of a 'revised self.'

CHAPTER 1

FURTHER EDUCATION IN IRELAND

Introduction

This chapter establishes the context of the study by examining the development of the further education sector in Ireland and outlining the diversity of its provision and its position as part of the Vocational Education system. It also looks at the increase in the numbers of mature students and some of the issues facing FE now.

Understandings of Further Education

The term 'Further Education', is somewhat problematic. It is often used in Ireland in conjunction with 'Adult Education' and in many documents and discussions the terms are interchangeable. This perhaps reflects the fragmentary evolution and the complexity of educational provision for those who have left compulsory schooling. The first formal recognition of Further Education was in the White Paper on Education from the Department of Education and Science (DES, 1995) which saw Further Education as a new category and proposed the establishment of a FE training authority 'to provide a coherent national developmental framework' (DES, 1995, p.81) and to have responsibility for the coordination of adult education, apprenticeships and Post Leaving Certificate education. Since then, FE has come to mean education and training 'which occurs after 2nd level schooling, but is not part of the third level system' (DES, 2004). Third level, or Higher Education, is made up of the universities, Institutes of Technology (ITs) and a number of other designated colleges, including Teacher Training Colleges. The White Paper on Adult Education (DES, 2000, p.27) defined adult education as 'systematic learning undertaken by adults who return to learning having concluded initial education or training'. A DES report on the state of the art of adult learning and education (2008, p.10) says 'the philosophy of Adult/Further Education in Ireland is to provide a range of education programmes for young people and adults who have either left school early or who need further vocational education and training to enhance their employment prospects and to enable them to progress their education up to a standard equivalent to upper secondary level'.

There is a degree of overlap in the definitions and practices which makes a clear separation difficult. The first Minister of State for Adult Education, Willie O'Dea, was appointed in 1997 which indicated a commitment by government to adopt a more focused approach to adult education. The new minister pointed out the lack of organisation and cohesion in the sector. Even though recognition for FE came only in 1995, post Leaving Certificate courses had been running for many years and the National Council for Vocational Awards (NCVA), established in 1991, recognised that 'an unofficial FE sector has, in effect, developed' (NCVA, 1992, p.33).

Until recently FE provision was the responsibility of two government departments: the Department of Education and Skills (DES) and the Department of Enterprise, Trade and Employment (DETE). The DETE delivered mainly through Foras Áiseanna Saothar (FÁS), the National Employment and Training authority established in 1967 as ANCO to provide vocational training, it now offers training programmes aimed at helping unemployed people re-enter the labour market. FÁS (now under the DES and re-branded as Solas) caters mainly for apprenticeships in a rather restricted number of areas. In Ireland the apprenticeship trades are traditionally 'male' occupations and the participation of women in apprenticeship is negligible (OECD, 2010, p. 8). In 2004 less than 0.5% of registered apprentices in the first phase of apprenticeship were women. Conversely, in 2003/4 women accounted for seventy two percent of participants in Post-Leaving Certificate courses, which cover many traditionally 'female' occupations such as hairdressers, beauticians and child care (Watson et al., 2006, p.8). There are other organisations engaged in sectoral training. These include Teagasc (agriculture) focusing mainly on farm training, Fáilte Ireland (tourism) and Bord Iascaigh Mhara (BIM) (fisheries). The DES provision is mainly through the 33 VECs which offer a range of programmes including the Back to Education initiative (BTEI) and Vocational Education and Training Scheme (VTOS).

The range of organisations offering Further Education and Training (FET) was complicated by the lack of a coherent qualifications structure. The National Council for Vocational Awards (NCVA), established in 1991, brought in a national system of certification. However, it was not broad enough to match the diversity of provision in FET and many providers continued to offer certification from UK and other bodies.

The Qualifications (Education and Training) Act, 1999 was an important development. It provided for the establishment of the National Qualifications Authority of Ireland (NQAI), the Higher Education and Training Awards Council (HETAC) and the Further Education and Training Awards Council (FETAC) to replace the NCVA. The NQAI was established in 2001 to set up and maintain a framework of qualifications and to promote and simplify access, transfer and progression (NQAI, 2003a, p.5). The National Framework Qualifications (NFQ) was introduced under the 1999 legislation to be 'a single, nationally and internationally accepted entity through which all learning achievements may be measured and related to each other in a coherent way and which defines the relationship between all education and training awards' (NQAI, 2003b, p.7)). It is based on outcomes – packages of knowledge, skills and competence. FETAC, which has responsibility for making awards from levels 1 to 6 on the NFQ, simplified the system of FET awards by taking responsibility for awards previously made by FÁS and a number of other bodies.

The National Qualifications Framework was a major part of European agreements, outlined in the Bruges-Copenhagen Process (2002) and reviewed every two years, to promote the European dimension in vocational education and training and the development of quality assurance mechanisms. It aims to establish a European Education and Training Area to include a European Qualifications Framework for Lifelong Learning to promote recognition of qualifications, mobility for working and learning, quality and transparency. Importantly for mature students, the European Framework of Qualifications and the Irish NFQ seek to promote the concept of lifelong learning and offer recognition of older awards and of knowledge and skills gained through experience or informally. This is seen by Connolly (2007, p.115) as a 'key solution to the elitism of the traditional qualifications pathways...'. It offered adult education a foothold in the formal system traditionally resistant to the needs of non-traditional students.

As many FE courses are certified by UK awarding bodies, the work of the NQAI in the recognition and alignment of awards on the NFQ is of particular interest. FE, for the most part, offers awards at level 5. However, there are a considerable number of courses offering awards at levels 6 and 7 and a small number at level 8 (Honours

Bachelor Degree). The provision of awards at levels 6, 7 and 8 in FE is a contentious issue as, under current DES guidelines, FE providers are not permitted to develop non-FETAC (Further Education and Training Awards Council) courses.

The number of mature students in FE has increased greatly in recent years. Older students have also increased their share of all levels of FETAC award. Those over forty years of age accounted for a fifth of all major awards in 2009, a third of minor awards and fifty-two percent of specific purpose awards. My focus is on mature students in full time FE courses in a college of the City of Dublin Vocational Education Committee which is the largest provider of FE and FETAC level 5 awards.

Background to FE in Ireland

The FE sector grew out of the Post-Leaving Certificate courses which emerged from the early 1980s. Many of the courses, practices, and assessment systems in this sector evolved as a result of pragmatism. As enrolments in traditional courses declined, schools looked for other ways of maintaining student numbers and teachers. O'Sullivan's point about the emergence of adult education is valid for FE also – that it was it 'depended very much on the level of interest and commitment on the part of a particular CEO, VEC committee member, school principal or individual member of a school's staff' (2005, p.518). Individual colleges were quick to recognise and provide for gaps in employment and training. Structures emerged within colleges such as Departments and Programme Leaders, which had no official recognition but met the needs of the institution. Many colleges re-named themselves Further Education Colleges to reflect the changing profile of the students, increasing numbers of whom were non-traditional students and many did not have a leaving certificate. For many years the sector used a variety of certifying bodies including the CDVEC's Curriculum Development Unit and a number of national and foreign examination boards and professional institutes. FETAC (the Further Education and Training Awards Council) now certifies one-year courses and most two-year courses are certified by EDEXCEL, an English awarding body, (formerly BTEC). Both of these systems are outcomes-based and modular.

Despite having significant numbers of students (30,500 in 2010, 11,000 within the CDVEC (DES, 2010) the self-styled FE sector still operates within the confines of the

second level system which have increasingly proved unsuitable. The doubling of student numbers in FE since 1991 is attributable, almost entirely, to mature students. The sector provides a range of courses not available elsewhere and accommodates students who could not access third level education directly as well as those who wish to study areas not catered for elsewhere. It also has an important role in providing 'second chance' education and facilitates a significant number of adults and non-traditional students both on designated 'return to learning' programmes and in mainstream courses. FE colleges now offer a large variety of courses including part-time day and evening courses, one and two year full-time courses and courses offering bachelors degrees awarded by Irish and UK universities. It blurs the distinction between second and third level education and is sometimes referred to as 'level two and a half'.

Most of the teachers in FE were initially trained for the second level system although many have undertaken re-training to adapt to their new and evolving situation. Very few have any training in teaching adults and many (especially those coming from industry) have no teacher-training at all. The system has established progression routes to third level colleges in Ireland and the UK. It is a feature of the fragmentary nature of FE provision that many colleges individually negotiate entry to the first, second or third year of third level courses with the colleges directly, while FETAC has also established progression routes to Irish HE colleges for holders of its awards. In 2009, eleven percent of all Central Applications Office (CAO) applicants were holders of a FETAC major award, fourteen percent in 2010 (Expert Group on Future Skills Needs, 2010). For many years there have been calls for the recognition of FE as a distinct sector within Irish education with appropriate structures and resources and particularly for the implementation of the McIver report (2003). This review of the sector called for an increase in funding to provide the type of resources required by FET providers. The measures recommended included a revised organisational and administrative structure, reducing teaching time and creating non-teaching roles for advisors, counsellors and programme managers. It addressed the teaching workload, upgrading buildings, facilities and learner support services and issues relating to teacher qualifications, induction and development. So far there has been much discussion but none of the recommendations have been implemented. Because of this, the operation of the new framework and quality systems are seriously hampered and the government's commitment to FE must be questioned.

Vocational Education

The VECs are by far the largest providers of FE. Historically, Vocational schools, established under the 1930 Vocational Education Act, and secondary school were part of a binary system in which students from working class backgrounds were encouraged to study for trades. Academic or liberal education was the preserve of the religious-controlled and often fee paying secondary schools or colleges. This dual system that emerged in Ireland fits with Bourdieu's contention that the education system is more an agent of social reproduction than of transformation in that it works to 'consecrate' social distinctions by cultivating ways of being that have the effect of reproducing social inequality. It has been suggested (Tierney and Clarke, 2007, p27) that the late recognition of further education and training in Ireland could be interpreted as reflecting negative perceptions of vocational education in a system that has historically been classically oriented. Anderson, Brown & Rushbrook, (2004) argue that the primary function that national governments attribute to vocational education and training is the provision of working class education. Those who lack cultural capital due to age, gender or socioeconomic background are more highly represented in this sector.

This history and ethos of Vocational Education is important to the country's VECs which created a broad range of courses including foundation level, degree and post graduate programmes, adult education, literacy, youth and community services, outdoor education centres and prisoner education. In some cases these take place in the same centre. VECs have a presence in every county and have been nominated to administer additional programmes on behalf of the state e.g. youth work and the higher education grants scheme under the new Student Support Bill (2008, p28). VEC colleges have responded quickly to perceived training and education needs and have worked with local groups to meet local needs. They are not selective in their intake and have a long history of accommodating non-traditional students and students with difficulties. Their history, structure and willingness to innovate meant that VEC schools were well placed to provide FE. The stigma attached to 'Techs' has not

disappeared completely and FE is sometimes seen as 'the poor relation' – there for those who couldn't make it into HE. However, perceptions are changing as colleges offering courses such as Film, Sound Engineering, Animation, Rock and Traditional Music have acquired a degree of 'trendiness'. These changes are indicative of the move from the technical to the knowledge economy and the growing status of these professions as well as the general national and European policy emphasis on employment and the VEC's adaptability and readiness to provide training for these new areas. VECs were also quick to recognize the economic and cultural importance of these disciplines and responded more readily than higher education institutions. This also reflects the blurring of the distinction between academic and technical knowledge and the shifting status of FE.

The history and culture of VECs is important also in looking at mature students in FE. The openness and flexibility that characterises VECs has been a significant factor in making returning to education attractive, especially for those whose previous experience had not been positive. This is something I will be returning to as well as the danger that the ethos of the VEC is being overshadowed by narrow neo-liberal instrumentalism. I will be considering the role of FE in addressing disadvantage and examining how disadvantage is conceptualized and the multiplicity of factors that impact on it.

Access Courses

In the UK Access courses, developed in the 1970s, were established 'to prepare mature students for entry into HE and provide the underpinning knowledge and skills needed to progress to a degree or diploma course at a university or college' (UCAS, 2004). Since then they have been key to widening adult participation in higher education. Before this, some institutions offered courses to prepare adults for entry to higher education. Otherwise they had to take 'A' levels or come to an informal arrangement with individual colleges or admissions officers. These arrangements were unsatisfactory, ineffective, unfair and lacking consistency. 'A' levels were considered by many to be unsuitable preparation for adult students because 'too little attention is paid to study skills or the more general problems of people returning to formal learning after what might have been a long absence' (West, 1996, p.3). As with the development of further education in Ireland, the development of Access courses in the

UK was often in response to the needs of individual colleges and the zeal of individuals within them. The convergence of other issues gave further impetus to their development. Inner city problems and race issues resulted in attempts to recruit more minorities into higher education and professions like teaching and social work. Increasing unemployment highlighted the need to improve the skills of the workforce and to widen access to higher education for those previously excluded.

As Warmington (2002, p.588) points out; 'Access programmes located themselves in opposition to the (gold) standard belief that those who would benefit from higher study could be identified and selected by the age of eighteen'. In the 1990s, the rationale for widening adult participation in FE and HE was firmly embedded in the discourse of the need for flexibility, re-training for the new technologies and new working practices. Warmington goes on to suggest that as work patterns have become characterised by impermanence, retraining is increasingly common. He cites Ainley (1993, p.87) who maintains that retraining consists not only of specialised occupational information but with 'generalised knowledge that allows individuals to be flexible and easily adapt to changing labour contexts. Similarly, in Ireland the Enterprise Strategy Group (Forfás, 2004, p.28) stated:

All occupations are becoming more knowledge-intensive, with a corresponding rise in the requirement for qualifications and technical skills. Employees will be required to acquire a range of generic and transferable skills and attitudes. In most cases, work is becoming less routine, with a requirement for flexibility, continuous learning, and individual initiative and judgment.

Access courses in the UK began as a pilot project scheme in seven local education authorities to prepare students who had 'special needs which cannot be met by existing educational arrangements' and who possessed 'valuable experience but lack the qualifications required' (DES, [UK] 1978). However, with the expansion of higher education they became a significant part of government strategy. By 1994 there were 900 formally validated Access programmes in England and Wales and such programmes were a 'major pathway to higher education for adult learners' (West, 1996, p.4). A caveat in the Deering Report (1997) has relevance for the Irish experience: 'There is a basic tension or contradiction in the whole Access enterprise. What is offered is a non-traditional pathway to a traditional provision; and a student-centered preparation for a subject-centered enterprise' (The Deering Report, (1997) V,

5:3). This is something identified by mature students in FE themselves and it has also been noted in relation to FE students who progress to HE in Ireland. FE students are perceived to get more 'hand-holding' and to be less equipped for third level study than students who completed dedicated access courses. This was also identified by Murphy and Fleming (2000) who point out that the learner-centered and experiential approaches of access courses are at odds with that put forward in university.

Although FE in Ireland does offer a small number of dedicated access courses with links to universities and some Institutes of Technology (ITs), access courses are not a common feature of provision in Ireland as they are in the UK. It is important to point out that many mature students in FE who progress within FE or to HE did not come to FE with the intention of progressing and would not have enrolled on a course offering access to HE. In most cases the idea of such progression, or participation at any higher level would not have occurred to them at the time (Fleming and Murphy, 1997, p.65).

VTOS and BTEA

The Vocational Training Opportunities Scheme (VTOS) is an educational opportunities scheme for unemployed persons which is funded by the Department of Education and Science under the National Development Plan 2007-2013 with assistance from the European Social Fund. The Scheme is operated through the Vocational Education Committees and provides a special range of courses in over 100 locations throughout the country designed to meet the education and training needs of unemployed people and is aimed in particular at unemployed people who are early school leavers (citizensinformation.ie).

Participants keep their benefits and secondary benefits, course fees are paid and books and materials are provided. Depending on individual circumstances participants are also eligible for other allowances, such as meal allowances, travel allowance and childcare and a training bonus.

To qualify, participants must be at least 21 years of age and be in receipt of an unemployment payment for at least six months. The aims of the scheme are to give unemployed people education and training opportunities which will develop their

employability, and to prepare people to go into paid employment or on to further opportunities leading to paid employment.

Participation in VTOS is in two modes - as a 'core' group of VTOS students participating in a group of up to twenty in a VTOS centre or adult education centre, or as 'dispersed' VTOS students participating in a group of students some of whom may be VTOS students and some of whom will be studying through other schemes/programmes or as part of the general student body. These 'dispersed' VTOS students generally attend Further Education Colleges. There are approximately 1,300 VTOS places in the City of Dublin VEC, the largest FE provider in the country, which are allocated among its colleges. The Back to Education Allowance (BTEA) is a similar scheme although operated somewhat differently. These supports are of crucial importance in allowing adults to return to education. In their document 'Government for National Recovery, 2011-2016, (2011, p.43) the Labour Party and Fine Gael promised that 'Vocational training for jobseekers will be a high priority. We will expand training options for jobseekers across the VEC, further and higher education sectors to facilitate upskilling of the labour force.' What this means in practice remains to be seen, perhaps extending the VTOS and BTEA programmes. It is worth pointing out that although there has been much welcome discussions about widening participation in HE, especially increasing the number of mature students, there has been no such discussion, apart from the election promise above, about widening participation in FE and its role in providing alternative routes to HE.

Mature students

In recent years there have been attempts to widening participation among sections of the population who were traditionally under-represented in HE. Widening participation is seen as central in promoting economic competiveness and becoming a 'knowledge economy.' HE has seen a huge increase in numbers progressing to third level in Ireland. In 1980, twenty percent of school leavers progressed to third level this had increased to fifty five percent currently (McCoy and Byrne, 2011, p.145). This expansion was, however, mainly in the 18-21 age group and disproportionately from managerial and professional classes (OECD, 2006). Participation levels of mature students is low by European standards, although this has increased significantly in

recent years, particularly in FE. There has also been a significant increase in the number of non-Irish students in HE and FE.

Various pieces of national legislation aimed at increasing participation were enacted since the mid 1990s. The 1997 Universities Act allowed for more places on HE courses and imposed a duty on universities to prepare and implement statements of their policies in respect of access to university education for all. The Equal Status Act (2000), the Disability Act (2005), and the Education for Persons with Special Educational Needs Act (2004) have created a framework for equal access and encouraged participation by under-represented groups. Financial and other support has been provided to support widening participation in higher education through the National Development Plans 2000-2006 and 2007-2013 and the HEA which have allocated considerable funds. The establishment of the National Access Office helped raise public awareness of access issues and establish targets and review progress. Initially the focus was on mature students, students from low socio-economic groups and on students with a disability (Conway, 2009).

Access officers in HE institutions have greatly supported entry by non-traditional students. The Disability Support Service in partnership with the National Learning Network provides disability support for limited number of colleges of Further Education in the City of Dublin VEC. This is funded by the European Social Fund and has been successful in helping people with a disability access courses in further education and supporting them while there. However, in further education, this support is available only to students with a disability and other disadvantaged groups are not offered similar help. This may be because the VEC sector, the main providers of FE, has traditionally welcomed students from disadvantaged backgrounds and other non-traditional students. The VECs still accommodate such students and many FE colleges and secondary schools are located in areas that are considered disadvantaged. However, the competition for some courses in FE is intense and the access issues can be very similar to those in HE.

The term 'mature student' is often used to differentiate older students from the 'traditional' students of 18 or so who have just left secondary education having done the Leaving Certificate exam. In FE mature students are those who are 21 years of age

or older. In HE in Ireland a mature student is 23 or older, in the UK a mature student is 21. The term adult learner is seldom used as almost all students in FE are over 18 and officially adults. The term 'non-traditional' student embraces many other subgroups. The report 'Access and Retention: the experience of non-traditional learners in Higher Education' 2008-2010 (RANLHE) defines 'non-traditional' as students who are under-represented in HE and whose participation is constrained by structural factors: including students from low income families, first generation entrants, students from (particular) minority ethnic groups, students with disabilities and mature students. All levels of education in Ireland have seen a dramatic increase in such nontraditional students in recent years with inevitable transformations and demands on resources and structures. The concept of the 'traditional student' is thus increasingly of limited use in FE and implies a homogeneity that does not take account of the complexity of individual histories and dispositions and, even as a cultural stereotype, has undergone rapid changes although the images of students as young, hard drinking, sex-obsessed party animals is perpetuated in many of the student notices on college walls.

There are, of course, shared commonalities of age and experience which clearly characterise student and other groups. However, a twenty-one year old single mature student will probably see themselves (and be seen) as very similar to an eighteen year old. Young mature students share most of the characteristics of students a few years younger. For this reason I will be concentrating, in this study, on the experience of students who have come back to full time education in the FE sector after an absence of several years.

Many twenty-one year old students in FE delayed continuing education or did not have the necessary qualifications to progress to HE. FE can provide an alternative route to HE or progression to vocational qualifications. Waller (2006) reports that in the UK 'non-disadvantaged' groups are using Access courses to enter HE. This he sees as supporting the contention (Ball, 2003) that the middle class eventually dominate any form of public service. In Bourdieu's terms, they have the necessary social capital to ensure this. The Points Commission, (DES, 1999, p.107) expressed concern about the potential of mature entry being seen as a fallback for already advantaged groups.

Government commitments to policies of lifelong learning and the demand for education to be of service to the economy and society has prompted great changes and increased the number of mature students in education. Many such students are coping with other obligations and demands and this forces a consideration of how the needs of such students can be accommodated and their contribution valued if learning in FE is to work for them. A small proportion of older mature students in FE have the resources, time and support, perhaps having retired from full-time work, and are pursuing their interests as a leisure activity.

In both FE and HE, modularisation, Recognition of Prior Learning (RPL), the introduction of the National Qualifications Framework, the adoption of European Credit Transfer System (ECTS) and new opportunities provided by technology for distance learning, offered the promise of greater flexibility in delivery, transfer and progression. There were also major implications for access, the establishment of structures more suitable for adult and part-time students and the recognition of informal learning and experience. In FE the promise offered by these innovations is yet to be explored.

FE now

Although the initial rationale of post-leaving certificate courses was to 'bridge the gap from school to work' the sector has changed significantly. The Term 'Further Education' has been adopted as recommended by the McIver report (2003) to reflect the changing nature of the courses and the profile of the student body. In summary:

- There are over 30,000 students in the FE sector. This year has seen huge growth in the demand for places. The DES has capped the number of FE students at 30,000.
- More than 50% of entrants to FE are mature students (over 21)
- Younger students tend to be concentrated in the middle range in terms of educational achievement. Older students more likely to have left school with no qualifications.

- FE an important route back to into education for those who did not complete second level.
- Seventy percent of the students in FE are female.
- FE is now an important alternative route to Higher Education and professional qualifications.
- There is no central application system admissions policies vary considerably.
 The Leaving Certificate Applied Programme (or equivalent) is acceptable for FETAC courses.
- Older and female FE students tend to come from less advantaged backgrounds than younger and male FE students.
- All FE students are less likely than average to come from professional backgrounds, especially younger females FE students.
- The changed profile of FE students has not been acknowledged officially and little effort has been made to consider the implications.

In my experience, within FE there are several, often conflicting and unstated, views that underpin practice in FE that reflect different positions on its ideology and role and form an important part of the background to the study of mature students.

• To what extent should FE offer second chance education?

This is reflected in individual approaches to student intake, especially in courses for which there is great demand. Certain courses are more welcoming of mature students. It is also reflected in the desire for 'success' as measured by course completion with a full award as opposed to anything else - 'Schools have little incentive to take those who will lower its performance profile' (Baker, Lynch, Cantillon & Walsh 2004, p.147). There is considerable variation in admission requirements and practices among FE colleges.

Vocational ethos v educational ethos

At present the focus in FE is on vocational development. This is particularly noticeable in debates in which the demands of the 'real world' are often contrasted with attitudes of education, care and personal development. This reflects the lack of a shared view of the purpose of FE.

Individual and community

There is little space in FE to include aspects of a broader social and global dimension. Some attempt is made to do this by individuals. This is also reflected in debates about attendance policy. Some see attendance and the meeting of outcomes to be entirely the responsibility of the individual student. Others see the process and communal aspect of learning to be of importance.

• Acceptance of outcome-based education.

There is little debate about learning needs or methods and their implications. This reflects the focus on the unproblematic transfer of skills, knowledge and attitudes in FE.

• Institutional status

Its second level status continues to be an issue as it greatly limits the kind of provision that can be offered. The lack of flexibility imposed on the sector means that arrangements are often made 'under the radar' without official approval. Under current DES guidelines the FE sector is not permitted to develop non-FETAC courses. In fact, thousands of students in FE gain level 6 awards from (and pay fees to) UK certifying bodies, mostly BTEC (EDEXCEL). The lack of a stated role for FE, the ad hoc nature of its development and the tolerance of provision that run counter to DES guidelines indicates ambivalence at the official level that merits investigation.

Conclusion

I have been working in the CDVEC for thirty years and I have been part of the evolution of the FE sector. As a practitioner I am well placed to observe how policy and practice interact and to see the real problems and the benefits of FE. The circumstances of its growth have meant that most attention was given to practical and immediate issues with little time for the development of a vision or philosophical base for FE. The increase in the number of mature students in education generally, and especially in FE, demands a reassessment of FE's role and function and the values that underpin it.

The strengths of the VEC system include its flexibility, its history and ethos. It has long experience of inclusive education and working with students from all backgrounds and circumstances including adult and compensatory education. The CDVEC, quite justifiably in my opinion, prides itself on its culture of care for its students and providing a safe, inclusive and supportive learning environment. This culture of care has prevailed even in the current climate. Adults are now a major part of the CDVEC's mainstream provision, and curriculum, pedagogy and practices in FE need to be reassessed and the value of public, accountable education reaffirmed. Individual colleges and the CDVEC management machinery deal, for the most part, with the immediate with little time set aside for reflection or planning. Providing time and structures for such a reassessment is a challenge I feel we must take on. Apple, (2005, p.228) speaking about public schools in the US says; 'How do we uphold the vision of a truly public institution at the same time as we rigorously criticize its functioning?

The next chapter considers the ideological context in which FE operates and the values it has adopted. It also looks at some critical theorists who offer a way of understanding this context and offer alternative perspectives.

CHAPTER 2

CONCEPTUAL FRAMEWORK

Introduction

This chapter sets out the conceptual framework for the work and looks at some of the significant ideas which I suggest have relevance for looking at mature students in Further Education. The concept of neo-liberalism is important as I suggest that FE in Ireland, in the absence of an underpinning philosophy of its own, has adopted this logic by default. This is the context in which I locate my study. Some critical theorists help illuminate this and offer alternative visions of FE. Ways of seeing mature students are broadened by using the lens of concepts such as habitus, class and disadvantage. Concepts are seldom uncontested and looking at their different interpretations has contributed to my understanding of the complexity of the issues and ways of addressing them.

Neo-Liberalism

There are diverse manifestations of neo-liberalism, many of the political directions and policy decisions in Ireland and elsewhere in recent times are easily recognizable as fitting into neo-liberal orthodoxies which celebrate individual responsibility and a radical reconfiguration of the role of the state. As well as being an economic model that became prominent in the 1980s (although its roots go back much further) it also describes an ideology, a set of ideas that inform a particular world view and how society should operate. The 'governmentality' based on this ideology is founded on a kind of economic thinking which favours the self-regulating free market over state intervention and is rooted in entrepreneurial values such as competitiveness, individual 'empowerment' and self-interest. Other characteristics include deregulation of the economy, privatization of state owned enterprises and liberalization of trade and industry regulation nationally and globally and the restructuring and downsizing of the public sector. Prominent also are tax cuts, reductions in welfare and social services, anti-trade unionism and the promotion of labour 'flexibility.'

Neo-liberalism is entwined with concepts such as globalization and the digital revolution which facilitated it. Increasingly mobile capital and the growth in global economic competition led to greater volatility in labour markets and the need for constant readjustment of structures and work patterns. Knowledge, skills and flexibility were seen as essential and education systems were harnessed to produce them. The development of a global market aided by the World Trade Organisation (WTO) and its design of free trade agreements also assisted the spread of the ideology of neo-liberalism and the commercialized culture that helped create and sustain it. In keeping with Marxist doctrine, the ideology of the dominant class became the dominant ideology. Large transnational corporations, financial elites and politicians were supported by public relations specialists offering convincing arguments for the benefits of their economic policies and, almost unnoticed, such thinking was adopted as normal, natural and inevitable. Finnegan (2008, p.58) describes neo-liberalism as a form of 'cultural hegemony' used 'to secure consent for the increasingly uneven distribution of power and wealth across the globe.' The weakening of public service media contributed to the decline of the public sphere and its invasion by the system thinking that characterise neo-liberalism. Government objectives were framed by the thinking and discourse of commerce and the more traditional pursuit of the public good and social justice were overshadowed by the imperative of meeting the demands of the market. The citizen was redefined as a client or customer and the state's relationship to concepts such as health and education were re-defined as marketable products. The individual became responsible for his/her own health and poverty is not something to do with social structures, but with the individual who has not acquired the training or learning to make him/herself employable. As Jarvis (2007, p.67) points out that the development of a market in education, and the personal responsibility to avail of this market, disadvantages those who are already disadvantaged.

As we have seen in this country, the power of global corporations and financial institutions have eroded the sovereignty of the state and the possibilities for political action; we *had* to support Anglo Irish and put billions into it and the other banks. The language and discourse of the market entered everyday parlance and now pervades many spheres, including education, and reflects the assimilation of a way of seeing these areas and their role in society.

Even though the present global financial turmoil confirms the inbuilt crisis of this model, the current public discourse in this country is seeking a solution within the established parameters. There are some indications that in other countries more radical solutions are percolating up from a local more popular discourse.

The European context.

Declining employment figures in the 1990s made education a matter of European concern. EU policy documents from the 1990s saw education and training as a key element of employment policy to enable its citizens to adapt to ever-changing technological developments and labour markets. Lifelong learning was seen largely in terms of keeping workers up-skilled and 'flexible' in the new globalised 'knowledge economy'. This is an important discursive shift in education discussion. The term 'lifelong learning' as opposed to 'lifelong education' put the emphasis on the individual who was seen as taking charge of his/her own learning, whereas 'education' placed the emphasis on the structures and the onus on the state to put structures in place as well as ignoring other economic and political factors (Coffield, 1999; Walters et al., 2004, Jarvis, 2009). An EU White Paper (CEC, 1995) on the Learning Society, proposed a number of short-term objectives including the necessity to encourage the acquisition of new knowledge and to bring the school and the business sector closer together. Although its meaning is contested, 'the Learning Society' has been accepted by most western governments as a major part of their education and economic policies. The reasons for this and the dangers for the individual and society have been outlined by Coffield (1999) who maintains that it is based on the acceptance of the unproblematic connection between a country's competiveness and the up-skilling of its workforce, and that there is no alternative (Bourdieu, 1998, p.30).

The National Qualifications Framework (NFQ), established under The Qualifications (Education and Training) Act, (1999) was part of major European agreements, outlined in the Bruges-Copenhagen Process (2002) to promote the European dimension in vocational education and training and the development of quality assurance mechanisms. The process aims to establish a European Education and Training Area to include a European Qualifications Framework for Lifelong Learning to promote recognition of qualifications, mobility for working and learning, quality

and transparency. Importantly for mature students the European Framework of Qualifications and the Irish NFQ seek to promote the concept of lifelong learning and offer recognition of older awards and knowledge and skills gained through experience or informally. This is of great importance as is offers non-traditional students a foothold in the formal system. Connolly (2007, p.116) points out that 'while policy development in lifelong learning has been driven by the emergence of the knowledge economy, the policy proliferation has created a fertile ground on which adult and community education can grow and flourish'. She goes on to caution for the need to be vigilant that adult education is not hijacked by another agenda which does not serve the public good.

A number of initiatives from the mid-1990s were designed to strengthen the links between the education and training systems and the enterprise and industry sectors. The cumulative effect of which was 'to highlight the awareness of the critical link between education and economic and social well-being' (Heraty et al., 2000). The two ideas of employability and citizenship underpin much EU policy on education and lifelong learning. Jarvis (2009) points out that the expansion of the European Community to incorporate many less economically advanced and culturally different countries made citizenship and social inclusion important priorities in EU education policy. Hogan (2010, p.23) suggests that there are good economic reasons for this as the consequences of social exclusion are disproportionately costly compared with the efforts that try to minimize it, and economic activity requires political stability. Deakin-Crick and Joldersma (2007) contend that social integration is distorted when reduced to its political or economic dimensions and education is a key vehicle in the process of social integration. The Lisbon Agenda (2000) set the goal 'to become the most competitive and dynamic knowledge-based economy...with more and greater jobs and greater social cohesion'. It underlined the central role of education and training in boosting EU competitiveness. The Irish Government also identified cohesion as one of six priority areas in the White Paper on Adult Education (DES, 2000). Education and training are seen to have a key role in promoting economic and personal development, social inclusion through the economic process of employment, and thereby inclusion in society and participation as an active citizen and 'the skills required for success in the market economy as the same skills necessary for active citizenship (Walters et al., 2004, p. 147). The concept of citizenship in the context of

an enlarged European Union and an increasingly globalised world is problematic. In Ireland, National Development Plans from the 1980s on, supported by international bodies, identify education and training as key to the development of the economy. In a context of globalization, many other countries around the world adopted similar policies and philosophies. Tony Blair (1998) said 'Education is the best economic plan we have'. More recently, the Prime Minister of Australia, Julia Gillard promised to "make education central to my economic agenda" (*The Age*, Melbourne, 15th July 2010). Hogan (2010, p.23) sees this recent appropriation of education to economic and political ends as having the character of a revolution as it occurred so rapidly (over the last thirty years) creating a 'new international uniformity'.

Neo-liberalism and Further Education in Ireland

The emergence of the Further Education sector in Ireland occurred largely in response to the unemployment situation in the 1980s and its growth has paralleled the transformation of Irish society by globalisation and neoliberal orthodoxies. FE claimed from its outset that it was responsive to the needs of the market, that it could identify gaps in training or provision and move quickly to develop programmes that would provide people with necessary skills. This early identification with market needs and demands was to characterise FE and be offered as one of its major justifications along with its flexibility and the practical, work-related nature of its courses. With the decline in traditional industries and the move of manufacturing to cheaper developing countries, Ireland, along with other western countries, focused on new and emerging industries such as childcare, hospitality, the 'culture industries' and occupations based on information and communication technologies. FE was at the forefront of many such developments.

I consider this context to be important in looking at FE in Ireland as I believe that the practices and ideologies currently dominant in FE have been heavily influence by the neo-liberal agenda with consequences for the way education is understood and for conceptualising non-traditional students.

The language and concepts of business now permeate FE – we frequently refer to students as stakeholders. Colleges compete with each other for students and VECs

now engage in expensive marketing campaigns. There are places for employers on college boards. People are recruited from industry to teach and staff are allocated to work with industry and employers. The practices and ideologies currently dominant in FE emerged by default and have resulted in the uncritical transfer of knowledge, practices, cultures and mindsets of industry with no interrogation of practices or exploration of alternatives. Certain ways of seeing have now become accepted as common sense and embedded in the discourse. The lifeworld has been colonised by the demands of the economy, the system world and the state. The system promotes an individual responsibility and autonomy in education

Finnegan (2008) and others have pointed out that as the market has become the dominant paradigm of the age, and activities like education, that were once seen as distinct from the market, are now 'commodified' and incorporated into binding multilateral agreements. In 2000 UNESCO estimated that education was a \$2 trillion 'industry' (Lynch, 2006a). The Government for National Recovery Document 2011-2016 promised to implement a strategy for the development of the 'Education Ireland brand.' There is also the demand for transparency and accountability in the expenditure of public money and a demand for 'value' measured in quantifiable outcomes and 'clearly defined societal benefits' (van Vught, 1994, p.6). FE in this country has recently been directed to become involved in Quality Assurance provision ('performativity' as measure of output). Since 2005 centres offering FETAC courses must have an approved Quality Document. Deakin-Crick and Joldersma (2007, p.83) point out 'This accountability move has profoundly reshaped the curriculum, the character of teaching and the nature of learning by markedly narrowing what constitutes knowledge, teaching and learning.' Much of the demand for Quality Assurance has come from the emergence of the international trade in education, (facilitated by WTO, GATS and the EU) engaged in by publicly funded as well as commercial organisations. The social benefits became identified exclusively with economic development and the provision of courses and graduates that can best contribute to it. The emphasis on narrowly defined 'quality' is a feature of newmanagerialism and is increasingly evident in FE as well as in other areas of education (Lynch, 2006b).

Outcome-based systems

The demand for quantifiable measures of education had led to the reliance on outcome-based models of assessment, which has implications for the kind of education provided in FE. It has been suggested (Burke, 2007, p.3) that this has led to an '...almost unstoppable trend towards thinking largely in quantitative terms about teaching/learning ...and the utilization of business language and concepts in education debates and policy documents'.

Competency-based systems and learning outcomes are a feature of vocational education in many parts of the world. Although understood somewhat differently in different countries, attempts have been made, and continue to be made, to further common understandings across Europe and elsewhere (CEDEFOP, 2008). A common feature has been the shift from a content-led curriculum to a learning outcomes approach which is often associated with a modular systems and attempts at European and national levels to develop systems of credit accumulation, transfer and international recognition (CEDEFOP, 2008). The European Qualifications Framework is such a system. This in turn is connected with student migration and the development of an international market in education. Encouraged by supranational organizations such as the World Bank and the OECD, many developing countries have embraced competency based systems (Anderson et al., 2004). Many developing countries are keen to establish similar international networks of qualification frameworks (Zambia and other southern African countries) which can be seen as an aspect of globalization and the development of education as a marketable commodity.

Outcomes-based systems were first introduced to enhance the employability of young people and improve the labour market relevance of vocational qualifications. Employers have a substantial input into the identification of skills required. This saw the student as someone functioning in the workplace that will need to be flexible and to have transferable skills of use to the market. The outcomes are based on standards that can be validated with transparent procedures to ensure quality at national and international level.

It has been suggested (Wheelahan, 2009, p.203) that this privileges employer perspectives and as such acts as a mechanism for social power and, by denying

students the underpinning theoretical knowledge, results in 'unitary and unproblematic conceptions of work'. Knowledge is limited to its application in a work context, but students 'are not provided with the means to relate it to its general and principled structure and system of meaning.' O'Brien and Brancaleone (2011, p.2) question the validity of the learning outcome paradigm and highlight what they call 'the central paradox of learning outcomes, that is to say, significant epistemological and pedagogical insights that remain hidden and inarticulate in the learning outcomes paradigm.'

Learning outcomes do not constitute a curriculum and teachers are free to develop their own teaching strategies that are appropriate to their situation to enable the outcomes to be met. 'The emphasis is on defining learning outcomes to shape the learner's experience, rather than give primacy to the content of the subjects that make up the curriculum' (CEDEFOP, 2008, p.9). Learning outcomes are separated from the process of learning and the duration and location of the course are not important as the focus is on what the learner can do not on how or where she/he learned to do it.

One of the putative advantages of outcome-based systems is that they are reported to recognize non-formal and informal learning. In this country there are, increasingly, requests from students in FE for Accreditation of Prior Learning (APL) and some colleges have a designated APL post. In some cases parts of an award, or indeed a complete award, may be claimed on this basis. However, there are considerable problems in implementing APL procedures¹ and the potential of the system has yet to be fully explored.

Although FE colleges have considerable control over how they organize their courses and their internal procedures, there are inevitable consequences of outcome-based systems in practice. As courses are modular and outcome-based, there is a great degree of flexibility in the duration of courses and the rate at which individuals can show themselves to have met the outcomes. Traditional timetabling and the standard of the academic year may no longer be seen as appropriate. Attendance may not be considered necessary, especially if part of the course may not be relevant or the

-

¹ These difficulties are often to do with not having staff available to do this. FE operates within the second level system and has very limited flexibility in allocating teachers to non-teaching roles.

outcomes may already have been met. This has implications for the kind of teaching and learning that happens and it promotes more of an emphasis on individual achievement and less on the communal aspects of participation and the kinds of learning this promotes. Outcomes based learning is appropriate in many aspects of vocational education, but an exclusive reliance on it in FE is limiting.

Promoting a broader range of access routes to HE has been a policy objective for some time (National Plan for Equity of Access to Higher Education (HEA) 2008-2013). However, the requirement of many HEIs for FETAC applicants to have achieved all the components of a level 5 award in the same year appears to be negating the advantages of a modular system and denying flexibility to non-traditional candidates.

Real vocational preparation is, of course, important, and measuring knowledge and skills for accountability and other purposes is legitimate. However, my concern would be that the exclusive reliance on such approaches, and the thinking behind them, reflects an extremely limited conception of education that has evolved by default, and reflects the hegemony described by many critical theorists (Giroux, 2004; Harvey, 2005; Apple, 2006). It underlines the need for a consideration of the nature and purpose of further or vocational training and education, the requirements of the individual and the nature and demands of society.

Habermas (1969, p.3) suggests that the university, along with its role in transmitting and developing the technical knowledge and cultural tradition of a society, also forms the political consciousness of its students. This can be done without deliberate intent as systems of knowledge and education are not neutral. Systems that are not reflexive, and aim to pass on established knowledge and practices uncritically, reproduce existing systems and ways of knowing. This is not in the best interests of the student or society. Burke (2007, p.4) cites Freire: 'all educational practice implies a theoretical stance on the educator's part' and asks what is the philosophy underlying the competencies-based standards approach to education? Such systems are based in behaviourist psychology (Fleming, 2009) and are attractive in that outcomes can be identified and measured and are effective in teaching in many of the skills-based areas taught in FE. However, an exclusive reliance on them is limiting as it can separate

performance from understanding and seeing the wider context in which the competency is applied.

The FETAC system is based on such measureable outcomes and the National Qualifications Authority of Ireland (NQAI) is part of an attempt to standardise awards, facilitate the recognition of other national qualifications and facilitate international transfer in keeping with the Bologna and subsequent EU agreements. Training for FETAC awards is now offered by an increasing number of private providers whose fundamental agenda is commercial rather than educational. VECs now tender to provide educational services to employers and in many ways have adopted the ethos and priorities of their commercial competitors. This has implications for the kind of education provided in FE. It has been suggested (Burke 2007, p. 3) that this has led to an '...almost unstoppable trend towards thinking largely in quantitative terms about teaching/learning ... and the utilization of business language and concepts in education debates and policy documents'. We frequently refer to 'stakeholders' and see students as clients who are surveyed to establish their satisfaction with the product, which the provider can then improve, to better meet the demand. Colleges, even within the CDVEC, are competitive and spend large amounts of time and money on marketing. Pressure to 'keep the numbers up' is felt by everyone in the system. (The rationale that was offered for this in the past was that increasing numbers meant that the college could maintain or increase its number of teachers. Changing economic and employment conditions, such as the huge reduction in the number of part-time teachers, have made this reasoning much less relevant). It is generally accepted that student numbers are a measure of the success of the institution. Assignments for the EDEXCEL awarding body, (wholly owned by Pearson PLC, the international media and education company) widely used in FE in Ireland, include a 'scenario' to simulate the workplace and make the material relevant. Employers have reserved places on the college's board of management and are frequently consulted on course development. People are recruited from industry to teach and their practical experience of a vocational area is greatly valued by the institution and students, often much more than the contribution of the traditional teacher.

Although this approach can offer many benefits to vocational training, the economy and society more generally, the uncritical transfer of knowledge and industrial and commercial attitudes allows for no interrogation of practices or exploration of alternatives. Most colleges have a post dedicated to working with 'industry' and college walls now carry commercial advertising. The General Secretary of the Irish Vocational Education Association (IVEA) said in 2006: 'The world of FE should be in the business of servicing the world of work'. Finnegan (2008) sees the politics of neo-liberalism as a powerful and complex form of cultural hegemony and cites Bourdieu, Giroux and Harvey who see it as a world view that promotes a certain conception of human beings as self-interested, calculating and individualistic. The incorporation of the language and concepts of the market into education and civil society is testimony to this hegemony 'when the commercialisation of life is normalised in educational bodies, its values are encoded in the educational system, generally without reflection' (Lynch, 2006a).

In my own daily practice in FE I can identify the relevance of these observations. We frequently hear that we have to live in the real world and prepare our students for it. As Finnegan points out, the argument that 'there is no alternative' serves to 'stifle dissent and narrows the social imagination' (Finnegan, 2008, p.59). Many of the practices that have evolved in FE reinforce a particular view of education as a commodity and the student as an individual consumer. There is no concept of a communal aspect to education where ideas are formed and exchanged through discussion or 'problem posing' but rather a promotion of individualism and a narrow focus on discrete measurable outcomes in separate modules. FE has adopted the neoliberal logic of education as a matter of consumer choice and as a private commodity quantified in internationally recognised units to be traded in the market place for individual and organisational profit. The emphasis on individual autonomy in adult education has been identified by Inglis (1997) and Grummell (2007) as being inherent in a neoliberal logic which makes the individual responsible for their own learning and readiness for employment. It lacks a social and critical dimension. Students are empowered to work in the system not 'emancipated' from it (Inglis, 1997).

In the context of globalisation and neo-liberalism, FE, in its content, practices and ethos, seems to have become primarily a resource for economic policy and the development of mind-sets that unquestioningly accept dominant orthodoxies. 'Colleges of higher and further education become producers of commercially-oriented professionals and skilled workers rather that public interest professionals' (Hanlon, 2000, cited in Lynch 2006a). Reid (2005) stresses the importance of public education systems in the context of challenges to democracy posed by economic globalisation. As Colleran et al. point out (2006) that education systems that focus on the needs of the knowledge economy do not necessarily serve the public good. Since 1930, VECs in this country have been providing education as a public good with none of the affiliations of other education providers. Hanlon (2005) has commented on how new ideological climates can affect the culture of public service organisations as well as the individuals working within them, how they perceive themselves and their careers and how they can be reshaped politically. This has direct relevance for teachers working in the FE sector at this time as many of the practices he describes are evident in Ireland also: casualisation of employment in colleges and the encouraging of flexibility, competition, the development of markets and quasi-markets, mechanical adherence to often inappropriate guidelines. These new practices, Hanlon argues, create uncertainty in the minds of public service professionals about their role and how they see themselves and how others perceive them. He also identifies a dissonance between different generations of public sector professionals as new regimes are accepted by recent entrants (especially if they wish to be promoted) and soon seen as normal as 'natural wastage wipes out the dinosaurs and new professional identities take hold' (Hanlon 2005). In my own college such divisions are noticeable, especially among those recently recruited, who know no other way of working, or part-time teachers still working in industry. Disparaging comparisons are often voiced about the public service and the 'real world', which contribute to the erosion of the educational ethos and practices of the CDVEC. The demands of a system that prepares students for employment and more recent policies and work practices have contributed to a 'culture of silence' (Freire, 1970b) and contributed to a lack of questioning in vocational preparation in FE.

The discourse of Further Education

The concept of discourse is useful in looking at how neoliberal ideologies have become normalised in the field of FE. Foucault (1980) maintains that discourse establishes the basis on which we make sense of the world and process our experience. It influences the ways we can think and act and produces our assumptions about the way things are. As Ryan puts it: '...discourse language and visual imagery do not simply 'reflect' or describe reality, but play an integral role in constructing reality and experience, the ways we know and understand the world, and what we assume to be natural or normal' (Ryan, 2004, p.23).

Fairclough (1997) supports Habermas in his contention of a progressive colonization of the lifeworld by the economy and the state, entailing a displacement of communicative practices by strategic practices. In making the point that contemporary society has been characterized as 'promotional' or 'consumer', Fairclough critically examines the discourse of the prospectuses of some English universities which he notes were informed by market research with promotion as their primary function. He comments that the discourse of such organisations can illuminate the decline in their implicit authority and their transformation into the construction of more entrepreneurial institutional identities. Similar shifts are identifiable in the promotional literature of FE colleges in this country with similar implications for power, ideology and hegemony. The language and practices of New Managerialism have become unconsciously assimilated into discussions of education. Jacinta Stewart, CEO of CDVEC recently asked 'who else has our market penetration?' (2008). All CDVEC colleges now put considerable resources into their brochure, website, student handbook, 'brand image' and slogan '...what were formerly seen as rightest policies have now become "commonsense"....'. (Apple, 2005, p.209). Habermas (1973) saw the necessity of unmasking power relations as embedded in linguistic convention in order to challenge them. The discourse of business and economics is now dominant in debates about education and alternative discourses and possibilities are excluded. Pring (1996) argues that the current tendency to use business terminology in education parlance is not innocent, but rather entails a fundamental re-definition of education itself. Outcome-based practices emphasise the 'product' of education, whereas, as

Palmer (1998) points out 'good education focuses on the process rather than the product. He also notes that the process may be uncomfortable with results not seen immediately. This imposes great constraints on education in a system dominated by commercial concepts of success and immediate customer satisfaction (Brookfield, 1987, p.234).

There is a lack of debate in FE about the changes in practices, systems and the thinking that underpin them. Connolly (2007, p.119) identifies the dearth of funding for research as being crucial in maintaining this silence.

Critical and transformative theorists

Real vocational preparation is, of course, important, and measuring knowledge and skills for accountability and other purposes is legitimate. However, my concern would be that the exclusive reliance on such approaches, and the thinking behind them, reflect an extremely limited conception of education that has evolved by default, and reflects the hegemony described by many critical theorists. It underlines the need for a consideration of the nature and purpose of further or vocational training and education, the requirements of the individual and the nature and demands of society.

It would seem appropriate that any conception of education in FE should incorporate an examination of the underlying assumptions and the vocational, economic and political framework for which it aspires to prepare its students. Freire's notion of 'conscientisation' is a useful idea in this context, 'in which men (sic) not as recipients, but as knowing subjects, achieve a deepening awareness both of the socio-cultural reality which shapes their lives and of their capacity to transform that reality (Freire, 1970a, p.27).

Habermas

Habermas suggests that the lifeworld has been colonised by the economy and the state. The lifeworld is 'a storehouse of unquestioned cultural givens from which those participating in communication draw agreed-upon patterns of interpretation for use in interpretive efforts' (Fleming, 2008a, p.6). Our perceptions and assumptions about the world are unquestioned and limiting. When the lifeworld is taken over (colonised) by the demands of the system world of the state and by the economy, the capacity to

make meaning is distorted and limited. Real need cannot be identified. 'The colonised lifeworld sees those things that are supportive of and consistent with the imperatives of the economy as common sense' (Fleming, 2008a. p.7). Evidence of the colonisation of education by the economic system is seen by Deakin-Crick et al. (2007, p.82) in the current perception of the purpose of education as being the production of individuals 'capable of maintaining their own economic well-being and who will participate in the economy as workers and consumers'. With the decline in critical investigation of assumptions, students view schools less as learning institutions and more for 'credentialising'.

A recurring theme in Habermas' work is the decline of the public sphere where issues are discussed informally. Such discussion is essential for democracy. He sees individualism and the focus on the self as being a consequence of this decline and the decline in the critical investigation of assumptions the resulting 'structurally depoliticised public realm' (Habermas, 1975, p.37) which promotes a privatised attitude to life and minimises challenges to the dominant order. Similarly, the focus on employment in FE and the assessment practices promote a narrow and uncritical individualism that precludes a consideration of alternatives.

Habermas (1975, p.37) identifies 'a learning project at the centre of democratic society that will rescue reason from being taken over by money and power'. For him, the realisation of the conditions for democracy is similar to the conditions necessary for adult learning (Fleming, 2008a p.13). Developing a critical reasoning necessary for democracy is a key task of adult education, as reason has become distorted under capitalism with the decline of civil society and the public sphere and the colonisation of the lifeworld. Habermas believed that adult learning happens primarily through speech. Free open discussion is central to his thinking on how to question everyday practices and assumptions and decolonise the lifeworld. He identified four conditions, or validity claims, for mature, rational authentic discussions: comprehensibility, sincerity, truth and legitimacy. Fleming (2008a, p.11) drawing on Habermas, says 'The role of the educator is one of creating classrooms that encourage the fullest participation in discourse, assisting students to assess critically the validity of their ways of making meaning and seeking perspectives that are more open to change'. For Habermas, this entailed communicative action, which '...involves critical reflection on

assumptions that underpin beliefs, a discourse to justify what we believe'. Such discussion must meet the conditions mentioned above and the ability to participate in it is a learned ability. For Habermas it is *the* learning task for adults. Brookfield (2005, p.251), again drawing on Habermas, says, 'learning systems can ...be created that institutionalise norms of critical scepticism as does ...the concept of critically reflective education for professional practice.' Andreotti (2010) and Hodge (2010) also point out that preparation for the jobs market does not rule out critical or transformative approaches to Vocational Education and Training (VET). Pring (1995, p.14) seems to be making a similar point when considering incorporating vocational values in academic education.

Such an education would develop self-reflection – the capacity to examine one's own motives and ideas, to understand oneself as a person, to examine the values which guide one implicitly and to ponder alternative goals....to develop....a social and political awareness....'.

Andreotti (2010, p.9) suggests that a new epistemology 'to decolonize the imagination and to pluralise the possibilities for the future by pluralising knowledge in the present in order to enable dialogue, relationships of solidarity and ideally, the collective creation of non-hegenomic systems.' She suggests the 'pluralisation of epistemological frameworks' and an emphasis on social and historical relativism as a way of opening more possibilities for the future. Part of the teacher's role, for her, is to see the world from different perspectives and 'to support learners to learn to unlearn' and to think 'otherwise'

Reflexive learning involves examining and challenging everyday practices and discussing their justifications. This kind of practice for Habermas, involves 'the mix of credible practical experience and socio-political awareness' (Brookfield, 2005, p.251). Such conceptions of education avoid 'the dualisms between education and training, between thinking and doing, between theory and practice...'(Pring, 1995, p.183). The separation of academic and vocational education reflected a particular idea of society based on dominant orthodoxies and a narrow conception of education. Dewey (1916) strenuously argued against the separation of theory and practice, thinking and doing. As Pring (1995) points out, the liberal/vocational, education/training dichotomies are false. The dominant idea behind vocational

preparation is the acquisition of instrumental knowledge or 'competence', and while competence as a goal might be limiting, it need not be. Further Education can develop a capacity for critical professional practice and transformative learning which combines being skilled in one's vocational area but also being able and willing 'to resist politically the dubious functional application or control of knowledge that one practices' (Habermas, 1969, p.47).

Mezirow

Drawing on Habermas, Mezirow's 'Transformation Theory' (2007) was based on research into 'What happens when adults return to education?' This is an explanation of how our frames of reference influence the way we make meaning and how they may be transformed to empower adult learners. 'A frame of reference is a structure of assumptions or taken for granted beliefs we have about reality' (Mezirow, (1995, p11). This is how we interpret the world and create experience. 'Transformative learning experiences are emancipatory in that they free learners from the constraints and distortions of their worn frames of reference' (Mezirow, 1995, p.11). A more fully developed and dependable frame of reference is one that is more inclusive, differentiating, more open to alternative perspectives and more integrative of experience' (Mezirow, 1995, p11). Central to Mezirow's transforming frames of reference is the notions of critical reflection and collaborative discourse, which questions the basis for assumptions. Collaborative discourse requires freedom, trust, participatory democracy, tolerance, solidarity, and respect for other viewpoints. It is the role of the educator to create the conditions which allow this discourse to occur. For Mezirow, 'the professional goal of the educator is to foster the learner's skills, habits of mind, disposition and will to become a more active and rational learner. This involves becoming more critically reflective of assumptions supporting one's own beliefs and those of others...' (Mezirow, 2003, p.5). Being critically reflective of assumptions is necessary to fully understand communication and where we and others are 'coming from', given the complexity of communication and the psychological, socio-linguistic, contextual, emotional and epistemic factors involved and allow transformation in frames of reference. This freeing from conditioned assumptions about the world, others and ourselves Mezirow (1998, p.6) sees as 'essential in the world of work, in functioning as a citizen in a democracy and in making responsible moral decisions in fast changing societies...'.

I found it helpful to consider the concept of attachment (Bowlby, 1979; Fleming, 2007) in the context of transformative learning. It has been acknowledged (Cranton, 1994; Andreotti, 2010) that questioning assumptions or long-held ideas can result in considerable discomfort. The idea of the 'strange situation' appeals to me as it captures the unease I have often noticed in students, especially mature students, when starting out on their college career, engaging with new perspectives or examining certainties. I think it offers a useful conceptual tool for looking at mature students in FE.

Bourdieu

Bourdieu offers concepts such as capital, field and habitus as a way of looking at education and its role in addressing inequality or reinforcing it. The habitus can be understood as '...the values and dispositions gained from our cultural history that generally stay with us across contexts...' (Webb et al., 2002, p.36.). The social and cultural messages that constitute the habitus often unconsciously influence our thoughts and actions. Social groups may have what Bourdieu calls a 'class habitus' which may influence educational choices and contribute towards a socio-cultural understanding of under-representation. The concept of institutional habitus is significant for education which was, for Bourdieu, the main institution through which class difference was maintained as the knowledge and experience of the dominant groups are favoured in educational institutions.

Bourdieu (1986) talks about three types of capital: economic capital which refers to economic resources – money etc. which can be converted to other kinds of capital, to buy educational resources for example. Social capital exists as sets of lasting social relationships, networks and contacts. Cultural capital is made up of forms of knowledge, education, skills dispositions and attitudes. Cultural capital allows dominant groups to legitimize the maintenance of status and power.

The field of education has the capacity to confer capital, especially cultural capital, which plays an important role in the reproduction of social structures and relations. In Ireland, historically, students at vocational schools came largely from the lower socioeconomic groups. Only a minority of students who finished primary school

progressed to the secondary system. It was especially difficult to progress without having passed the Primary Certificate exam, introduced in 1929. Many went straight into employment, or became unemployed. Others, mainly children from working class families, went to vocational schools, where having a Primary Certificate was not a requirement. The separation of vocational and general academic education was reinforced and remained until the late 1960s and the establishment of Community and Comprehensive schools.

Ireland was not the only European country in which the education system was structured in such a way as to encourage students from working class backgrounds to study for trades and to value the knowledge associated with academic or liberal education over other kinds of knowledge. The binary system that emerged in Ireland fits with Bourdieu's contention that the education system is more an agent of social reproduction and stratification than of transformation in that it works to 'consecrate' social distinctions by cultivating ways of being that have the effect of reproducing social inequality. The academic (often fee-paying) secondary schools were a means of transferring or acquiring the cultural capital that allowed for the perpetuation of social divisions and dominance and exclusion. The two systems operated side by side at post primary level in almost every town in Ireland. The 'College' for academic education and the Vocational school or 'Tech' for the others – similar to Bourdieu's Grand Ecoles and Petites Ecoles (1996). Considerable snobbery existed (and to a degree still does) which perceived the 'Techs' to be for the students who weren't considered suitable for the higher status academic education.

The structure of social space as observed in advanced societies is the product of two fundamental principles of differentiation – economic capital and cultural capital – the educational institution, which plays a critical role in the reproduction of the distribution of cultural capital and thus in the reproduction of the structure of social space, has become a central stake in the struggle for the monopoly on dominant positions.

(Bourdieu, 1996, p.5)

The habitus is key to the reproduction of social relations. It is as much unconscious as it is conscious and is incorporated into a way of being, habits of mind and attitudes. It is the way that values and cultural capital are passed on. This is a useful way of looking at how vocational education passes on the dispositions of a vocational area

(cultural and social capital) as well as the technical skills. It also helps in examining the vocational learning cultures and different kinds of capital associated with various strands in the education system – HE and FE and even with different courses within the one system or college.

Colley et al. (2003) suggest that learning in vocational education is not just a matter of acquiring a determined set of skills and competencies rooted in outcomes referenced models of curriculum, but that there is a hidden curriculum of adjustment and conformity to the dispositions and attitudes of a vocational area. It is a process of identity transformation and the development of a workplace habitus. The acquisition of the emotional capacities and personal qualities for the role are the key aspects of the training. It is 'a process of becoming' (Colley, et al., p.471) and of structuring a vocational identity that is of paramount importance is establishing a legitimate position within the field. Habermas (1969, p.2) speaks of acquiring the 'extrafunctional abilities' which he describes as 'those attributes and attitudes relevant to the pursuit of a professional career that are not contained per se in professional knowledge and skills.' The 'guiding ideology of practice' (James, 1989, p.25) underpins a vocational culture and is an important aspect of the field. Colley (2003) maintains that a central aspect of the student's learning is an orientation to a particular identity, a sense of what makes 'the right person for the job'. This vocational habitus dictates how one should properly feel, look and act as well as the values, attitudes and beliefs one should espouse. As many of the tutors in FE come from 'industry' and are immersed in the vocational culture, there are limited opportunities for interrogating that culture. Part of the education process may be to filter out candidates who are seen as not fitting in or not having the right attitudes (Bates, 1991). For Bourdieu, it is through the habitus that social reproduction takes place. The concept of institutional habitus may be useful in the context of retention in HE institutions, especially with reference to mature students and other non-traditional groups.

In this research I see these concepts as offering a way of seeing participation in education, especially by non-traditional groups. Archer (2003) and Reay et al. (2002) emphasise the difficulty inherent in concepts such as 'class' since the meaning of the terms are fluid, changing and limiting. Bourdieu offers other understandings by illuminating the way cultural norms shape the educational expectations and decisions

of social groups and how they contribute to patterns of participation and exclusion. This also contributes to understanding the concept of 'disadvantage' (Thomas and Quinn, 2007) and the factors that are seen to constitute it and inform attempts to remedy it. It focuses too on the way different kinds of capital may be necessary to successfully participate in education and how the distribution of this capital serves to exclude some groups. It prompts an investigation of how those disadvantaged by lack of cultural capital can be supported and the role of institutions, policies and individuals in doing so.

Conclusion

The first section of this chapter outlined the neo-liberal context in which FE in Ireland operates. It is suggested that FE has, by default, assimilated neo-liberal attitudes in its policies and practices and now needs to re-evaluate its role. Brookfield (2005, p. viii) defines critical thinking as 'the process of unearthing, and then researching, the assumptions one is operating under, primarily by taking different perspectives on familiar, taken-for-granted beliefs and behaviours.' This offers a way of understanding and questioning the current position and considering alternatives that may offer ways of addressing disadvantage and inequality.

The theorists mentioned in this chapter have had much to offer in illuminating my experience in FE and its practices and in contributing insights that may contribute to a way forward. They offer too, a vision of further education which, along with vocational preparation, includes a focus on the personal and the social and a questioning of established practices and systems. Whatever political aspirations motivate it, cultivating an 'ability of individuals to disengage from the tacit assumptions of discursive practices and power relations in order to exert more conscious control over their everyday lives' (Kincheloe, 2000, p.24) is a key task of adult education, especially as the discourses of vocationalism and new managerialism have colonized FE. Learning for living and learning to make a living need not be incompatible. We need to consider the balance.

CHAPTER 3

REVIEW OF SELECTED LITERATURE

Introduction

Although Further Education (FE) has been a significant sector of Irish education for many years, very little research has been done on the area. FE in Ireland emerged in an uneven fashion usually through the efforts of individual principals, teachers or institutions and it is only in more recent times that there has been a degree of cohesion in provision and awards.

Irish FE is significantly different to FE provision in other countries and while studies of other systems have much to offer, there are important differences, including differences in structures and training of staff, which somewhat limit their immediate usefulness for exploring the experience of mature students in FE in Ireland. The similarities however, are noticeable and most systems are dealing with the same macro-political and ideological influences as we are. Mature students have similar experiences and cultural and institutional differences help to illuminate the situation in this country. From a methodological point of view, the approaches taken by other studies have proven to be extremely useful and have influenced the decision on a methodology for this study.

Mature students in FE in Ireland have been somewhat neglected in research. There are, however, a number of important studies on mature students in Higher Education (HE) in Ireland (Fleming and Murphy; 1997, Lynch, 1997; Inglis and Murphy, 1999), however, the FE sector is significantly different to HE and these studies pre-date the transformation of Ireland which occurred in the last decade.

I am going to look at the two major studies on FE in Ireland: A Report in 2006, by the Economic and Social Research Institute (ESRI) for the DES (Watson et al.) which looked at Post-Leaving Certificate courses and Employment outcomes, and the McIver Report, (2003). Neither study sought to explore student experience or views or to examine the implications of the growth in the numbers of mature students in FE.

There are a small number of unpublished Masters theses that touch on various aspects of FE but none look at the experience of students, especially older students.

I will also be looking at some more recent studies. 'Hidden Disadvantage' (McCoy et al., 2010), a study of the low participation in Higher Education by the Non-Manual Group', and 'Where Next? (2010) A Study of Work and Life Experiences of Mature Students (incl. disadvantaged) in Three Higher Education Institutions, by Fleming et al. for the Combat Poverty Agency, have much to offer to a study of mature students in FE.

The McIver Report

The McIver Report (2003), the one major review of FE in Ireland, was based on institutional analysis and examined 15 colleges among those with 150 or more students.

The review had come about because of the commitment in the White Paper on Adult Education: 'Learning for Life' (2000) and the recognition that a FE sector had been operating within the second level system for many years. Increasingly it was pointed out that these structures were unsuitable and the FE system could exist only by creatively stretching the second level regulations. The situation was extremely unsatisfactory. The report was to examine and make recommendations regarding the organisational, support, development, technical and administrative structures and resources required in schools and colleges with large scale PLC provision having regard to good practice in related areas across the system and in other countries.

The McIver report's detailed recommendations were a blueprint for the establishment of a well resourced, vibrant FE sector, separate from second level with several changes in structures and staffing to reflect the distinction between the needs of staff and learners in FE and second level education. It called for an increase in funding to provide the type of resources required by Further Education and Training (FET) providers. The measures recommended included a revised organisational and administrative structure. It recommended reducing teaching time and creating non-teaching roles for advisors, counselors and programme managers. It addressed the teaching workload, upgrading buildings, facilities and learner support services, and issues relating to teacher qualifications, induction and development. The methodology

encompassed a questionnaire for the 15 colleges, detailed site visits, visits to other similar institutions in Ireland and abroad and a review of the literature.

The McIver report was extremely important in many ways. It was the first study of the area and collected data and detailed the scope of the work done in FE and made a strong case for the establishment of a separate FE sector. The vision was of a modernised sector with improved facilities and childcare whose staff would be trained and properly resourced to respond to adult learning needs. Those working in the area were pleased with the report and encouraged to think that changes would be made as a result. Since 2003, despite continuous calls for its implementation, nothing has been done. Staff hours and staff-student ratios are still high and the structures and funding still does not reflect the complexity of FE or the range of needs it tries to meet.

The report used both qualitative and quantitative methods and was effective in fulfilling its terms of reference. It was limited in that it examined only 15 colleges from those serving 150 or more students and excluded all second level schools with PLC students. Although there was an element of student consultation, it was, of necessity, very limited. It drew together information on a very disparate sector that provided a good base for further study.

The Post-Leaving Certificate sector in Ireland.

One such further study was the 2006 'The post-Leaving Certificate sector in Ireland: A multivariate analysis of educational and employment outcomes', by Watson et al. This is a mainly quantitative study. It recognised the relative neglect of the area in the research literature and sought to address some central and key issues surrounding the role of post-Leaving Certificate (PLC) courses: to what extent the levels of participation in PLC courses have changed over time, the profile of PLC participants in terms of gender, age, educational attainment and regional location, the impact of PLC participation on progression to further study and how PLC leavers fare when they enter the labour market. The report drew predominantly on a supplemented sample of the PLC sector, weighted to ensure representativeness of gender, programme year and age group.

From the point of view of this study, one of the most striking findings was that since the early 1990s there has been a steady increase in older participants (over 21) who now make up roughly fifty percent of FE students. This dramatic change in the student profile indicates the success of these courses in bringing people back to the educational system and the need to consider the implications of this change something the report does not address. It makes interesting points in terms of the family background of FE students. All students are less likely than average to come from professional backgrounds. Young male and female students are particularly likely to come from intermediate non-manual backgrounds. This is significant as these groups have relatively low rates of entry to higher education. This is the subject of another report, 'Hidden Disadvantage?' (2010), considered below.

In terms of performance at Leaving Certificate, most FE students are in the middle range. However, older students are more likely to have no qualification, which again shows the importance of FE in providing a route back into education for those who did not complete second level. Although a considerable number of students progress to third level education, the report maintains that FE is more likely to be an alternative to HE. It raises the issue of the marked regional differences in participation in courses and in the relative numbers of older and traditional participants, and suggests that this may be accounted for by the different approaches of various VECs. This points to the unstructured and haphazard development of the sector, which is not developed in the report.

Some of the information in the report was hard to follow for the non-technical reader and, perhaps because its brief was limited, it did not thoroughly pursue the implications of its findings. It was a good general picture but did not capture the practice on the ground or deal with any of the problems facing FE. It considered the employment outcomes and progression to HE, but did not consider any of the other personal or social benefits of participation.

From a practitioner's perspective, these two reports, those of McIver and Watson et al. gave an overview of the sector and confirmed many of the ideas about the broad social and academic backgrounds of FE students we had from observing our own situations locally or picked up anecdotally. They also made us aware of the variations in

provision, student profile and progression and suggested areas that need further investigation.

Watson et al. (2006) use the traditional social class and socio-economic approaches to categorise students which fails to explore more complex aspects of difference or significant differences within groups. McCoy et al. (2010) and Thomas and Quinn (2007) point out the limitations of considering disadvantage in such a confined way and emphasise the need to include a consideration of cultural and social capital as well as economic capital. Archer (2003) highlights the difficulties inherent in the traditional conceptualisations of class which rely on economic or employment status and do not take account of how class may interact with other social identities such as race or gender. This is particularly relevant when considering mature students who are especially difficult to fit into established categories.

Although both studies together give a good indication of the complexity of FE provision and the broad patterns of participation, they do not discuss the differences between course type: one or two year, level of the course and the differences in the characteristics and expectations of the students. The student experience is absent from these studies. Watson et al. do not deal with some of the major issues currently facing FE, and which existed at the time the reports were written, such as the reliance on non-national awarding bodies for awards above level 5 and the refusal of the DES to allow HETAC awards in FE. However, they provide a useful platform for further investigation of FE.

Hidden Disadvantage?

The report 'Hidden Disadvantage? A study of the Low Participation in Higher Education by the Non-Manual Group', (McCoy et al., 2010) by the Economic and Social Research Institute (ESRI) for the Higher Education Authority (HEA) is more nuanced in its perception of social class and the factors contributing to low participation in Higher Education. It raises the inadequacy of a categorical approach, specifically in this case the classification 'non-manual', which it says is not homogenous but made up of two distinct groups – the intermediate and other non-manual with different characteristics. It fundamentally questions the traditional definition of under-represented groups in HE as it points out that participation by

young people from lower white collar backgrounds is no higher than those from lower blue collar backgrounds. It maintains that the merging of these two groups, intermediate and other non-manual, has concealed a dramatic picture of educational disadvantage among young people from other non-manual backgrounds.

The study combines quantitative analysis of trends with qualitative research exploring attitudes, experiences, aspirations and expectations of young people from the non-manual groups. The combination of in-depth open-ended interviews and quantitative data provided a thorough overview as well as uncovering the complexity of non-participation and the opinions and feelings about post-school decisions. Methodologically the combination of quantitative and qualitative methods was effective in providing context and revealing the experience and thinking of the participants.

The report has little to say about mature students but has several interesting things to say about FE. It points out that, although participation in FE courses among school leavers declined over the 1997-1998 and 2002-2004 periods, the take-up of such courses among older people meant that numbers in the sector overall increased. It also shows the relatively high level of entry into FE among young people from other nonmanual backgrounds and skilled manual backgrounds, the groups whose low levels of entry into HE are the subject of the report. (There is, however, no such information on mature students in FE.) This highlights the role of FE in providing a route back into the education system for those who were disaffected by it at secondary level. FE offers clear progression routes, including access to higher education. FE students from non-manual and manual backgrounds have relatively high levels of participation in the LCA programme which would preclude them from direct entry to HE. The report by Watson et al. acknowledges that a considerable number of FE graduates go on to higher education. FE has been very successful in encouraging second chance and nontraditional students and in redressing previous negative experiences of education, which this report highlights is an important determinant of post-school choices. This, and the use of concepts of social and cultural capital, helps to locate the reasons for non participation in broader societal inequalities and not in individual or family deficits.

The report identifies as important, but does not go into detail about, issues to do with the opportunities to progress from FE to HE. There are significant difficulties about this that are not mentioned including the number of places allocated to applicants with FETAC or other awards, the recognition of non-national awards and the requirement of many HEIs for applicants to have achieved all the components of a FETAC major award in the same year. These issues need further exploration as they impact seriously on progression to HE from alternative routes and facilitating access by non-traditional groups.

Other issues to do with decisions to progress to HE are identified. Many are relevant to mature students in FE. Information, guidance and school expectations and support are considered key, especially for groups with little family involvement in HE. Financial matters play a major role in decision making and availability of grants and information about them is vital. The availability of jobs in recent years was a disincentive for some young people to go to college, while for mature students, becoming unemployed can start a process that results in a decision to return to education.

The report also examines some of the other socio-economic disparities in participation in FE courses. Nationally, females make up more than seventy percent of FE students, which may be explained by males taking up apprenticeships.² The type of school attended is also a key determinant of participation; students who attend a vocational school are over two and a half times more likely to participate in a FE course. This may be partly accounted for by the fact that almost all FE courses are run by VECs and students attending vocational schools would therefore be more familiar with FE, especially when FE shares staff and a building with the second level school. To put it another way, the institutional habitus of FE colleges would be familiar to these students. Other connections could be made here between the status of FE and the historical and current perception of vocational education in the Irish context. (It would be interesting to know if socioeconomic and other disparities exist in course choice within FE.)

_

² It is likely that this balance has changed. The recession and the decline in the building industry have led to a huge reduction in apprenticeships.

From a theoretical point of view, the cultural and rational action perspectives were helpful in illuminating the complexity of the issue and how cultural norms and beliefs contributed to the shaping of expectations and decision-making. The concepts of habitus and cultural capital are useful in understanding many aspects of educational inequality. Likewise, the idea of 'distance from social origin' and the importance of cultural factors offered ways of understanding aspects of education choice. Unlike the other studies, 'Hidden Disadvantage?' used the voices of students which brought the quantitative data to life and gave a very rounded picture.

Where Next?

Another report which is relevant to this study is 'Where Next? A Study of Work and Life Experiences of Mature Students (incl. disadvantaged) in Three Higher Education Institutions' by Fleming et al. for the Combat Poverty Agency (2010). The report set out to examine how mature students and 'disadvantaged' mature students view and value higher education after graduating. While much has been done to improve access to HE for mature students, this report attempts to address a gap in the research and focus on what happens to such students while in HE and particularly after they graduate. It maps their destination after completion of their first degree, explores their transition into the workplace or other arenas and investigates their reflections on their motivations for entering HE, their expectations and experiences. I found this HE study very interesting as it is very relevant to an investigation of the experiences of mature students in FE.

Like 'Hidden Disadvantage', 'Where Next?' used a mixed methods approach and used questionnaires and in-depth semi-structured interviews to identify and analyse broad trends while attending to the lived experience of the graduates. The literature review which outlined the context for the research emphasised the government commitment to widening participation in education for groups that have been under-represented, including mature students. This is seen as key to overcoming disadvantage and solving a range of social problems. It is also seen as vital for up-skilling the workforce and maintaining economic competiveness. The low levels of educational attainment of Irish adults are to be addressed by targeting mature students and encouraging their participation in HE. This policy context is important in looking at FE as well as the findings of this report, which point out that despite many initiatives, there are still

considerable inequalities in access to HE and relatively few mature students participating. The report points out that the lack of disaggregated data has limited research, which is the case in FE also. It also maintains that there is little relevant research on mature students and a dearth of qualitative research about their experience.

Interestingly, the concept of 'disadvantaged mature student' was found to be of limited usefulness as the research findings were not significantly different whether one was a mature student or a disadvantaged mature student. The unpacking of the concept of disadvantage was interesting as it provided a socio-cultural as well as an economic understanding. The report looked at the social and personal benefits of participation in HE as well as the economic benefits. This is something which I think should be explored with FE students also as it gets away from seeing education in purely instrumental terms in a system whose *raison d'etre* was to prepare students for employment. The report found that, for the students, 'destination' included a broader range of concepts than employment or progression and included self esteem, confidence and identity. This was a 'sensitising concept' I brought into my study.

Overall, this study was extremely useful and interesting as it focused on mature students, and although it dealt with HE, many of the issues are relevant to FE also. It identified the global and national context of education policy as being informed by an instrumentalist and work-oriented imperative which is also important to our understanding of FE also. Extending the concept of destination and finding that the benefits of participation in HE went beyond financial or career rewards confirms my experience of FE students and is something that is explored in the current study.

It suggested different understandings of mature students and prompted questions about the role of FE in addressing disadvantage and encouraging mature students and others considered to be disadvantaged, into education. Although this is something VECs pride themselves on, often justifiably, there is no co-ordinated approach or targets. With the current economic downturn and student demand in FE exceeding supply, it is likely that non-traditional students will be less welcome. Looking at mature students through the lens of equality, addressing generational poverty and exclusion gives a

focus and imperative for FE to re-think its role and implement real policies in this area.

The report mentions the vital role of access offices in HEIs in ensuring the participation and success of mature students. FE does not have access offices and this work is done (when it is done) informally and on an individual basis by teachers or others in the system. FE operates within the second level system and has a limited, number of teachers with posts of responsibility it can appoint to non-teaching roles. The recent cuts and the decision not to replace posts of responsibility when an incumbent retires have greatly impacted on the non-teaching support FE can offer. Applicants are often refused on the basis that 'we don't have the supports they need'. It provides much of the services it does through good will, juggling budgets and the very flexible implementation or ignoring of rules. There is a disincentive to accept students who require extra support of any kind. This illustrates the importance of the demand for appropriate structures and supports called for in the McIver report.

Access courses were identified as a very important factor in mature students' progression to HE and the importance of developing access routes from FE was mentioned. This is clearly an area where providers of FE (especially the CDVEC which is the largest provider) could usefully work with HE especially as unskilled and semi-skilled workers are poorly represented in HE but young people whose parents are in these categories constitute a large proportion of FE students. (There is no such information specifically about mature students in FE). FE is well placed to encourage the most disadvantaged, marginalised or impoverished sections of Irish society who are not represented as mature students in HE.

The issue of financial supports is common to mature students in FE and HE. It was interesting to see how important they are to both sectors. The support of family is also common and the report identified the trickle-down effect in encouraging an acceptance and normalising of learning in the families of the students. These 'vertical and horizontal' supports need greater attention in FE.

I believe the report entitled 'Where next?' went beyond the McIver report and the report by Watson et al. on the PLC sector by framing its study of mature students in

the context of poverty and equality. Its use of qualitative as well as quantitative data enabled it to give a comprehensive sense of the issues affecting real people as well as providing an overview and a context.

RANLHE

Another work on a similar area to mine is the literature review of the ongoing 'Access and Retention: The experience of non-traditional learners in Higher Education 2008-2010' (RANLHE). Although concerned with HE, its subject matter, conceptual framework and methodology are of great interest as FE shares many of the concerns and characteristics outlined. This European project has eight partners from England, Poland, Germany, Ireland, Scotland, Spain and Sweden. It is set in the context of a huge increase in student numbers in HE globally and increased recognition of the benefits of participation for social and economic development. It recognises that despite the increase in student numbers, there are still inequalities in participation which many countries are taking measures to address. Its overall aim is to examine issues of access, retention and non-completion in relation to non-traditional students, including mature students, in HE on a comparative European basis. It recognises, as does 'Where next?' that, while historically there has been much discussion of issues of access and recruitment of non-traditional students, recently there has been more focus on the importance of outcomes and retention in HE. These issues are important for the individual student, the institutions and the state and its investment in education. This is of interest to FE in Ireland which has high non-completion rates and is an area which requires attention.³

The report takes a biographical research approach and defines 'non-traditional' as students who are under-represented in HE and whose participation is constrained by structural factors: including students from low income families, first generation entrants, students from (particular) minority ethnic groups, students with disabilities and mature students. The report claims that the biographical research methodology allows for a more complex understanding of non-traditional learners with its focus on the historical and socio-cultural as well as the psycho-social context of their lives. As an international project with eight partners, it seeks to understand international

-

³ Based on one study in a large FE college, anecdotal evidence and experience. The CDVEC does not compile statistics on its FE colleges.

patterns and contexts and to look at different theoretical approaches and, finally, to identify key issues.

Although there are significant differences, the concept of higher education is undergoing change in all participating countries characterised by more diverse provision and neo-liberal influences. The distinction between further and higher education is becoming increasingly blurred. In England, an increasing proportion of HE is now being taught in FE colleges as a partnership or franchise. Irish FE is different in many ways from its counterpart in England but these trends are noticeable in Ireland too. FE colleges offer degrees in association with Irish and UK universities, although there are clear indications that the DES is anxious to curtail this kind of development.

Some of the difficulties associated with using terminology in an international study are addressed. Definitions of social class, ethnicity or low income are difficult to use in comparative international research. The review refers to the international research done by Thomas and Quinn (2007) which also acknowledges this difficulty and uses the idea of 'first generation entrants' but points out that this term overlaps other categories of non-traditional student. All the countries showed significant inequalities in HE participation and understanding what is meant by 'non-traditional' and identifying the complex and inter-related factors impacting on inequality is important. The situation is complicated further by the limited availability of reliable data and agreement on terminology.

The difficulty of accessing accurate data on retention from institutions in Ireland was mentioned. Interestingly, one study showed lower completion rates in Institutes of Technology (ITs) than in universities which was considered significant as ITs historically have attracted a greater proportion of students from a working class background and are seen as being on the front line in widening participation. There is some evidence that FE, which also caters for a high proportion of students from the lower socioeconomic groups, has a high rate of non-completion, especially in one-year FETAC level 5 courses. This is significant as FE has an important role in widening participation and progression to HE. However accessing accurate data on the FE sector is a problem.

The review's theoretical perspectives on retention were interesting and had much to offer FE. Tinto's model is widely used and although it does not focus specifically on non-traditional students is useful in considering issues of retention. Tinto (1993) identified five conditions for student retention: expectations, support, feedback, involvement and learning. This model has been criticised for its emphasis on integration and says little about factors outside of college that affect the student.

Bean and Metzner (1985) maintain that for adult students the traditional social environment of the campus is not nearly as important. Mature students are more focused on the academic side of college and interactions in college do not matter as much as they do for younger students. Outside concerns and supports, especially from family, are much more important for mature students. This is supported by Yorke (2004) who suggests that Tinto's model does not take enough account of these factors. This emphasis on the importance of family support for mature students was mentioned in the two previous reports also.

Other useful concepts used to understand access and retention issues are cultural capital, habitus and field. Drawing on the work of Reay et al. (2002; 2005) the report refers to the idea of non-traditional entrants being like 'fish out of water' while HE entry for traditional students is very much part of their habitus and often seen as a natural progression. Cultural capital and the institutional habitus are seen as key elements in retention. Such socio-cultural theories require an examination of the complex ways that class and culture function and interact with institutions and pedagogy. The concept of social capital or habitus is a useful way of looking at the comparative status of FE in post compulsory education and even of different colleges or courses within FE. It is also helpful in thinking about the perception of VECs and further education colleges and why they might be more accessible centres than universities or other HE institutions for non-traditional students. FE colleges, as VEC institutions, are usually long established and well recognised in the community. They are familiar and have none of the connotations of elitism that often attaches to universities. This can be important in the decision making processes of applicants.

Looking at overall trends, key factors and the development of various 'typologies' are useful ways of generalising and give a broad picture of retention and drop out.

However, it is only when concepts are unpacked that the complexity of understandings and their meaning in individual lives is revealed.

The next section of the review gives an overview of the approaches identified in the literature to improve retention and success, especially of non-traditional students. In this it distinguishes between national, institutional, faculty and departmental approaches. Many of the ideas resonated with my own experience and I found they offered extremely useful ways of engaging with the issues at several levels.

The review had much to contribute to the ideas I brought to the current study. The international nature of the project highlights the similarities and suggests some differences in circumstances and approach. Agreeing terminology and matching data sets, when they exist, forces a closer examination of terms and concepts and their use in different contexts. The biographical research methods used in the report allow a greater appreciation of the importance of the historical, cultural and social context which I think is necessary for a full understanding of students' experience. I found it useful to consider how the concept of 'mature student' might overlap and interact with other aspects of 'non-traditional' student.

College Knowledge

College Knowledge by Fleming and Murphy (1997) looked at the experiences and characteristics of mature students in NUIM. Although quite an old study now, it proved helpful by setting the increase in numbers of mature students in the context of national and international developments in HE at the time and it highlighted many of the issues affecting mature students (including financial, academic and personal) and how they interacted with motivation, learning and success. It points out that while access is an important issue for equality and combating disadvantage, what happens after the student gets into college is equally important. A key finding is the significance of how the student negotiates the difference between their experiential or common sense knowledge and the knowledge of the university, 'College Knowledge'. I was interested to see if there would be similar issues with students in FE.

The study used mainly qualitative methods but included some quantitative information and, unusually for the time, included the voices of the students themselves from semi-

structured interviews and focus groups. This is the approach adopted by the current study.

Conclusion

Overall, the recent literature reflects agreement on the desirability of upgrading the education levels of the adult population and promoting access, especially for non-traditional groups. There is also a focus on retention and ways of understanding the cultural, social and personal factors that impact on it. The literature also offered a way of seeing mature students in FE in the broader context of disadvantage and exclusion. Recently there have been many initiatives in Ireland and Europe to encourage greater access to higher education, yet despite the large number of students it attracts and its role in adult education, Further Education in Ireland seems to be neglected.

These studies helped to develop 'theoretical sensitivity' for the present work and meant that I brought into the study ideas about habitus, social and cultural capital as well as traditional definitions of social class and disadvantage. The concepts explored in 'Where Next?' 'Hidden Disadvantage' and 'RANLHE' contributed to my understanding of the complexities of the issues and confirmed my choice of qualitative interviews for investigating the experience of students and the meaning of education in their lives.

Chapter 4

METHODOLOGY

Introduction

I began this research with the general aim of exploring the experience of mature students in a large Further Education college. This chapter describes the epistemological stance and the overall methodological approach taken. The decision to use a questionnaire, focus group and qualitative interviews and the specific strategy of constructivist grounded theory are explained. The second part of the chapter outlines the sampling procedure and the data collection and analysis process which were done following constructivist grounded theory guidelines.

Part one

Epistemology

Thinking of this research and how to go about it inevitably prompted reflection on my own epistemology. My guiding assumption would be that social reality is not 'out there', independent of the knower but is constructed by and between personal perspectives based on experience and may change over time (Gergen, 1999). The constructionist position would therefore have considerable appeal. As a consequence, I would seek to understand the 'complex world of lived experience from the point of view of those who live it' (Denzin and Lincon, 1998, p.221). I would consider it important to be aware of the important cultural, social and historical contexts that help to shape the construction of the experience of reality and the concepts that are used to understand it. I would, therefore, favour methodologies that value the subjective and recognise the importance of context and social interaction in the construction of meaning.

Often these subjective meanings are negotiated socially and historically... they are not simply imprinted on individuals but are formed through interaction with others and through historical and cultural norms... (Cresswell, 1998, p.8)

A constructivist approach looks at how sense is made of the world and how it is interpreted by those experiencing it. This facilitates the exploration of individual positions while, at the same time, allowing common factors to be identified. This is a

useful way, I think, to look at people having, on the face of it, a similar experience (mature students), but for whom the meaning and effects could be very different. This approach recognises that our individual circumstances affect our perception and construction of 'reality' and that meanings may be varied and complex. It acknowledges too the role of social, historical and institutional factors in the development of experience.

Language and discourse are an important focus of critical, constructivist enquiry which recognises that the concepts used to understand the social world are themselves socially constructed and invested with meaning that is contextual and temporary. Foucault (1980) maintains that discourse establishes the basis on which we make sense of the world and process our experience. It determines the ways we can think and act and produces our assumptions about the way things are. As Ryan (2006, p.23) puts it: '...discourse, language and visual imagery do not simply 'reflect' or describe reality, but play an integral role in constructing reality and experience, the ways we know and understand the world, and what we assume to be natural or normal.' Constructivists would see the dominant discourse as just one of many possible and ask how such a situation came about and what it says about the nature of power.

Qualitative methodology

As this inquiry involves investigating the experience of mature students it is exploratory and open-ended. The voices of the participants are central. From a constructivist perspective, a qualitative strategy of enquiry seems appropriate to attempt to capture the uniqueness of the individual and quantitative methodologies are more constrained in what they can contribute to understanding experience. However, quantitative methodologies are useful for providing a view of the broad context and used with a qualitative approach contribute to a more complete understanding of individual experience.

Qualitative research allows researchers to attempt to get at the inner experience of participants, to determine how meanings are formed through and in culture, and to discover rather than test variables. (Corbin and Strauss, 2008, p.12)

This echoes Mason's (2002, p.3) view of qualitative research which she says is '...broadly "interpretivist" in the sense that it is concerned with how the social world is interpreted, understood, experienced produced, or constituted.' Although qualitative research is influenced by a range of philosophies and traditions, Mason (2002, p.3) identifies a number of common factors: it is grounded in a broadly interpretivist philosophy, it is based on methods of data collection that are not standardised but are flexible and sensitive to social context, it is based on methods which involve understanding complexity, detail and context. Bryman, (2008, p.386) similarly, sees the main preoccupations of qualitative research as: seeing through the eyes of the people being studied, description and the emphasis on context, emphasis on process, flexibility and limited structure, concepts and theory grounded in data. Cresswell, (1998, p.17) suggests 'compelling reasons' for undertaking a qualitative study which seem to fit what I propose to do. The research question is exploratory and open-ended. I intend to present a detailed view of the topic and study participants in their natural setting.

Grounded Theory

Having established that a qualitative research framework is appropriate, I decided that a grounded theory methodology would be the most suitable for my research question. In grounded theory the researcher does not set about testing an existing theory but tries to allow the theory to emerge from the data. Grounded theory has been understood in many different ways since its original development by Glaser and Strauss in 1967. They were dissatisfied with what they saw as an over-reliance on quantitative testing of propositions derived from grand theories that had little relevance for a particular 'substantive' area. They aimed at generating more contextual, local theory that would work and be relevant to those being studied (Richardson, 1999, p.76). Qualitative research at the time was considered 'impressionistic, anecdotal, unsystematic and biased' (Charmaz, 2006, p.5). Even researchers who favoured qualitative methodologies acknowledged the absence of systematic guidelines that would give it rigour and greater credibility. Glaser and Strauss attempted to provide practical guidelines for the construction of theory from data. 'Grounded Theory' implied that the theory is grounded in the data and employs an iterative approach involving continual sampling, comparison and analysis. The

approach is inductive rather than deductive and participants' own accounts and their social contexts are given great attention.

The evolution of grounded theory has been described (Mills et al., 2006) 'as a methodological spiral that begins with Glaser and Strauss' original text and continues today.' Strauss and Corbin (1994, p. 283) recognize that 'a child once launched is very much subject to a combination of its origins and the evolving contingencies of life. Can it be otherwise with a methodology?'

I will be using the constructivist version of grounded theory proposed by Charmaz: '...a constructivist approach places priority on the phenomena of study and sees both data and analysis as created from shared experiences and relationships with participants' (Charmaz, 2006, p.130). Charmaz (2006, p.2) defined constructivist grounded theory as '...a contemporary revision of Glaser and Strauss's (1967) classic grounded theory that:

- assumes a relativist approach
- acknowledges multiple standpoints and realities of both the grounded theorist and the research participants
- takes a reflexive stance toward our actions, situations, and participants in the field setting, and constructions of them in our analysis

Charmaz refers to explicit yet flexible guidelines and views grounded theory methods as 'a set of principles and practices, not as prescriptions or packages. (Charmaz, 2006, p.9). However flexibly applied, grounded theory involves an inductive approach. It starts with the data, involves theoretical sampling, coding, comparison, writing memos and successive levels of analysis building levels of abstraction.

Coding

Coding is one of the fundamental processes of grounded theory. In coding, according to Bryman (2008, p.542), 'data are broken down into component parts, which are given names.' It is a way of organising large amounts of data and organising it in more manageable, conceptual pieces. 'Coding means naming segments of data with a

label that simultaneously categorizes, summarizes and accounts for each piece of data' (Charmaz, 2006, p.43). Charmaz sees at least two phases of coding: initial coding which involves becoming intimately involved in the data and being open to all theoretical directions and giving each segment of data a name or label. These labels are provisional and may point to gaps that need to be addressed in subsequent data collection. The second phase of focused coding is more selective and serves to 'pinpoint and develop the most salient concepts in the data into analytic categories'. It 'requires decisions about which initial codes make the most analytic sense to categorise your data incisively and completely'. Theoretical coding, a third phase of coding, specifies 'possible relationships between categories you have developed in your focused coding' (Charmaz, 2006, p.63). Theoretical coding leads to the development of core categories that integrate other categories and leads to theory development. Overall, the process moves from initial coding, which stays close to the data, to more abstract ways of conceptualising the phenomenon being investigated.

Constant Comparative method

The use of 'constant comparative methods' (Glaser and Strauss, 1967) is another fundamental feature of grounded theory which involves constantly analysing the data and codes looking for similarities, diversities and variations. This comparison is done as the data is collected and may prompt further data collection. This iterative process continues throughout the research process and concludes only when categories are saturated. 'Categories are 'saturated' when gathering fresh data no longer sparks new theoretical insights, nor reveals new properties of these core theoretical categories' (Charmaz, 2006, p.113). The concept of saturation is somewhat disputed. Dey, (1999, p.257) considers it an 'unfortunate metaphor' because he considers it imprecise and prefers the term 'theoretical sufficiency'.

Memos

Memos are informal analytic notes that the researcher makes during the research process. Writing memos is not just a method of recording thoughts but a process of interacting with the data, thinking about it and exploring relationships and emerging connections and ideas. Memos are written in 'informal, unofficial language for personal use' (Charmaz, 2006, p.80) and can serve to unblock and develop ideas, identify gaps and indicate the development of the researcher's thinking as the process

evolves. Memo writing forms 'a space and place for exploration and discovery.' They can assist and trace the development of more abstract categories and direct further analysis and coding.

Theory

Discussion of what constitutes theory inevitably impinges on epistemological issues and highlights some of the different approaches among grounded theorists themselves. Grounded theory has moved from its more positivist beginnings. Glaser and Strauss talk about theory being 'discovered' in the data, while Charmaz prefers to talk of the 'construction' of theory and emphasises the involvement of the researcher and the researcher's perspectives and history in the process of representation and interpretation.

Positivist theories aim to establish causation, to predict and to emphasise generality. Interpretative theories seek to gain an understanding, which it acknowledges is an interpretation. 'Interpretative theory calls for the imaginative understanding of the studied phenomenon....assumes emergent, multiple realities; indeterminacy; facts and values as linked; truth as provisional' (Charmaz, 2006, p.126). Objectivist grounded theory, in contrast to the constructivist approach, sees the data as independent of the social context of their production and the researcher. The differences have implications for how theory is understood. Charmaz sees theorising as an activity inseparable from the researcher's sensitivity that can address local worlds and see them in the context of larger social structures. It involves making connections, developing insights and ways of understanding. Grounded theory studies generally produce substantive rather than formal theories. A substantive theory seeks to uncover 'the basic social-psychological or social structural processes that are used by persons or social groups in response to specific social problems,' while a formal theory is 'a broader based and more generalized process that occurs in a variety of distinct, yet theoretically similar, social situations (Kearney, 1998, p.181).

My aim in this enquiry is not to discover an objective 'truth,' but to explore the experience of mature students in FE and to untangle, with the participants, some of the complexities and consider how they can be interpreted and understood. My initial intention was to use Grounded Theory methods as exploratory tools in the

investigation. It was later in the process, when theoretical sampling reinforced the implicit meaning of many of the themes and connections between them that abstractions and theory emerged.

A deductive approach, i.e. the testing of theory by proposing hypotheses for evaluating it, was not considered a suitable option for this work. An inductive approach was more appropriate. 'By using an inductive approach, the researcher can attempt to make sense of a situation without imposing pre-existing expectations on the phenomena under study' (Mertens, 1998, p.160).

A number of considerations in respect of adopting an inductive approach needed to be honoured:

- Not treating people as 'objects' as in some experiment;
- Showing respect for peoples' points of view and valuing peoples discrete perceptions based on individual experiences;
- Recognising that there are multiple realities based on individual perspectives;
- Realising that the researcher is not detached from the research process but is an integral part of it, as she/he is the interpretative 'instrument';
- Seeking to ensure that the interaction between the respondents and the researcher is dialectic in nature; the better to examine the multiple realities of the situation;
- That understanding may be achieved through a process of interpretation in which successive levels of generalisation and abstraction are constructed and evaluated:
- The use of inductive logic implies going from the particular evidence to a
 general understanding by making sense of what may be disclosed by an
 interpretation of the data.

I would consider all of the above considerations to be important and consistent with the epistemological position outlined. This process of analysis demands that an emergent stance is adopted both in the generation of data and its subsequent analysis. The utilization of the grounded theory method supports this approach and, although it offers a set of guidelines, it also calls on the researcher to be creative and sensitive.

Sampling

Purposive sampling (Ryan, 2006, p.84), unlike random sampling, aims to sample data in a strategic way according to their relevance to the research question. Theoretical (or purpose) sampling, according to Glaser and Strauss (1967, p.45) is:

the process of data collection for generating theory whereby the analyst jointly collects, codes and analyzes his data and decides what data to collect next and where to find them, in order to develop his theory as it emerges. The process of data collection is controlled by the emerging theory...

Data collection is an ongoing process and not a discrete and separate phase of the research process. Data collection and analysis is an iterative process, with further data being collected on the basis of the analysis of previous material. The relevance of the cases is important, not how representative they are. Unlike statistical sampling, the extent of the sample is not known in advance and sampling is complete when theoretical saturation has been reached rather than when a whole pre-determined sample has been studied. 'Sampling is an issue throughout the research and analysis process as decisions have to be made constantly about what sources or techniques to use or what material to interpret' (Flick, 1998, p.62). My initial sampling was the start of the general process of exploration. Later, when categories began to emerge, theoretical sampling helped test hunches and fill gaps.

Part 2

Research Procedure – preliminary data collection

My interest in this study is mature students and their experience in FE. I believed that the students themselves would provide the richest data. The college in which I teach provided a good location for research as it has a large number of mature students and I am very familiar with the environment. Initially, in order to get an overview of the mature student population in the college and to see what issues arose, I decided to administer a questionnaire and to carry out a focus group with a small group of students drawn from different areas of the college.

Questionnaire

I decided to conduct a questionnaire to gain an overview of the characteristics of the mature student population in the college and to identify issues to be explored more thoroughly in later interviews. It also allowed me to identify students willing to participate in interviews. Having first carried out a pilot questionnaire (Appendix 1), I administered a questionnaire (Appendix 2) to 100 mature students in the college. The sample surveyed by questionnaire was not representative but I did include students from each of the college's broad departmental areas. It fitted the definition of 'convenience sample' as defined by Bryman (2008, p.183) as I chose students who were available to me and whose tutors were available and willing to facilitate me. In most cases I distributed the questionnaire and waited while it was completed, having arranged with the group's tutor to have this time at the end of a class when the other students had left. This ensured a high response and no waiting for forms to be returned. Very few students opted not to complete the survey and there was no difficulty in reaching the target of 100. This approach ensured a 100% response and meant that I was present if any clarification was needed. This was especially useful during the pilot phase. It was useful too in that it provided an opportunity for discussion about the nature of the research and in some cases became an informal focus group. I took notes immediately after such discussions which contributed to the research process. In some cases, I gave the questionnaires to the tutor who asked students to complete them. To some extent I had a captive audience and as some of the students would have known me or my role in the college, and I was especially careful to emphasise the purpose of the questionnaire, its confidentiality and the completely voluntary nature of participation. I also sought permission to carry out the research from the college principal and undertook not to disclose the identity of the students, the college or the staff. In the whole process my position as an 'insider' and my personal contacts made it easy to work in the environment. McCracken (1988) points out that familiarity with the culture being studied has the potential to dull the investigator's powers of observation and analysis. It also offers the huge advantage of an intimate knowledge of the area. There was clearly a need to 'manufacture distance' and create a certain detachment to try to 'make the familiar strange.'

The development of the questionnaire took some time. There were several revisions in content and format as I was anxious that it should be as clear as possible and easily

understood (Bryman, 2008, p. 222). It was also important that it could be completed in a reasonably short period and not appear daunting. It consisted of three A4 sheets with an explanation of the purpose of the research and my contact details at the top. I would have preferred to have a single double-sided sheet, but reducing the font size and the spacing made it look cramped and unattractive. The pilot questionnaire generally went well but it did indicate that a number of questions needed to be refined. For example, the question on marital status did not initially include the category 'cohabiting'. Also on the basis of the pilot, I summarised the most common replies to two questions and asked the respondents to tick one or more from a range of options. In these cases I included an option of 'other' and invited comment. The completed questionnaires were generally satisfactorily done although relatively few were very forthcoming with comments in any of the responses where they were invited.

Although the data provided by the questionnaire was very helpful and the convenience sample makes no claims to be representative, it resulted in a disproportionate number of students on two year courses. For my purposes this was not a problem. Many of these students had already done a one year course and some would have done a preliminary return to learning one year course as well. There were no observable differences in the responses of students on one year or two year courses. As I had hoped, the questionnaire produced useful background information and ideas to follow up in interviews. The results are discussed more fully in Chapter 5.

Data collection - Focus Group

The focus group was part of the initial exploration and was made up of students I knew or who were recommended to me for this purpose. Although I had a number of issues I wanted to discuss, gleaned from my own experience, reading and informal discussions with individual mature students, they were general points to initiate discussion not lead it. I explained that I was embarking on this research and I was keen to get their ideas and insights. I was aware of the power differential between us implied by our roles as students and teacher. I had taught two of the participants a year earlier on separate courses. I was not teaching them at the time of the focus group and was unlikely to be teaching any of them in the future. We had remained friendly and frequently had informal chats. They had expressed an interest in participating in the research and I was confident that our relative positions would not

be a problem. As a teacher, I was comfortable with the focus group format and I knew that the interaction between participants can produce very useful ideas and sometimes different views. I was aware also of the danger of one participant dominating the discussion and the possibility of what Bryman (2008, p.489) refers to as 'group effects' and the tendency of members of such groups to agree with each other rather than disagree.

I had arranged for five participants. However, on the day two could not attend. I knew two of the participants but they did not know each other. There were very sociable and there was no awkwardness. I explained over an initial cup of coffee what the research was about, the timescale and how I saw the session going. I discussed how the material would be used and promised anonymity. The discussion was lively, good humoured and frank with everyone participating well but one was inclined to talk more than the others. I felt that the session provided me with several themes to pursue later in interviews. I was surprised by the strength of feeling expressed about younger students (although this was lead particularly by one individual and was not borne out in subsequent one-to-one interviews) and by how much the participants felt they had changed as a result of coming back to education. I was reluctant to ask about personal issues or potentially sensitive areas in this context which confirmed my belief that semi-structured interviews were the most suitable means of generating further data. The discussion lasted about an hour and a half. I later transcribed the session to allow for closer analysis.

Data Collection - Interviews

I decided to use face-to-face qualitative interviews as the main method of data generation. The interview allows for topics to be explored differently and in more detail than a focus group and can provide data that is not accessible in any other way. The information extraction model of interview (Ryan, 2006, p.74), where the objective interviewer seeks answers to straightforward questions, is unsuitable for exploring the complexity and nuances of lived experiences and assumes an uncomplicated relationship between question and answer. Mason (2002, p.52) uses the term 'data generation' in preference to 'data collection' to emphasise the researcher's role as someone who actively constructs knowledge. Instead of just

finding data 'out there,' the researcher works out how best to generate it from the chosen sources.

A constructivist approach acknowledges that the participants are not just a source of data waiting to be extracted and that the interaction between the interviewer and the interviewee is crucial in the process of constructing meaning. It is important to remember that the interview is a construction or reconstruction of a reality and not an uncomplicated retelling of a prior reality (Charmaz, 2006, p.27).

Mason (2002, p.62) suggests the common characteristics of qualitative interviewing include: the interactional exchange of dialogue, a relatively informal style, a topic-centred, narrative or biographical approach with a fluid and flexible structure to allow unexpected themes to be developed, an acknowledgement that knowledge is situated and contextual, meaning and understanding are created in an interaction between the interviewer and the participant. The idea is to get an insight and understanding of the participant's perspective and that this may include 'the texture and feeling as well as the facts' (Franklin, 1997). The interview in this approach is an interpersonal situation and the characteristics, skills and sensitivities of the interviewer and his/her ability to establish a relationship of trust are clearly of considerable importance.

I was aware that semi-structured interviews demand careful planning (Mason, 2002, Cresswell, 1998) to ensure that relevant data are generated. With this in mind I formulated a general interview guide (Appendix 3) to ensure a focus on the research questions. This was, however, thematic and included developing a feel for responses that might be significant even if unexpected. It was informed by the insights from the focus group and questionnaire and my own sensitivity. Charmaz (2006, p.29) advocates an interview guide with 'well planned open-ended questions.' As the interview is an attempt to understand how the participants experience and understand events, it is important to be flexible enough to allow them to develop their ideas and not close off avenues prematurely. This is especially relevant when using a grounded theory approach which advocates not starting out with fixed ideas or preconceptions. The focus was on what is significant to the interviewee and I used my interview guide with that in mind.

McCracken (1988, p.24) suggests the use of a questionnaire which, he says, 'does not pre-empt the open-ended nature of the qualitative interview' but allows the interviewer to concentrate on the task in hand and pursue any opportunities that may arise. He also recognises the importance of the relationship between the interviewer and the participant but cautions against allowing 'intimacy to obscure or complicate the task at hand' (McCracken, 1988, p.26). Many authors (Bryman, 2008; Charmaz, 2006; McCracken, 1998; Mason, 2006; Denscombe, 1998,) offer useful practical advice on interviewing and compelling reasons for the interviewer to be aware of the many factors that affect the dynamic of the process.

For my initial sampling I drew up a short list of potential interview candidates based on those who had indicated on the questionnaire that they were willing to participate. Later, theoretical sampling was used to help explicate the emerging categories (Charmaz, 2006, p.100). The interviews lasted between fifty and ninety minutes. I recorded the interviews and personally transcribed them and although this was very time consuming I was glad I did it. It allowed me to develop a close familiarity with the data and preserved nuances that would have been missed or forgotten using the transcripts or recordings alone and proved to be a valuable part of the process of analysis. Some researchers suggest that taking notes is preferable to recording interviews. I think recording has several advantages. I found it extremely difficult to take notes and give the interview and the participant adequate attention. Note-taking is distracting for both and interferes with the natural conversational style of the encounter. The relevance and importance of data might not be apparent at the time. Points that later turned out to be relevant were often ignored in earlier readings and might not have been considered worth recording at the time if I had been relying on notes. I also wrote brief memos immediately after the interviews outlining my impressions of the encounter and some possible themes emerging. I found that, as the interviews went on, my technique improved and I was more willing to ask for clarification or examples or to suggest an interpretation to the participant. In most cases I had the opportunity for a second interview which was extremely valuable in clarifying or substantiating themes and impressions and getting the participants' opinions on what I had taken from their interviews. I contacted the participants again within two to four weeks of the first interview. I gave them a copy of the transcript at or before the second interview. I checked up on any points which were not clear to me

in the recording and discussed the process of the interview with the candidate and their reflections on it. All had enjoyed it and were glad of the opportunity to discuss and explore the issues. I asked if there was anything they would like to add, change or clarify and asked if they had any questions. I spoke about what I had found to be of particular interest in the interview and asked if they agreed with what I was taking from the discussion. I followed up on some significant themes that were emerging in other interviews and frequently asked 'Can you tell me more about...?' The second interview and subsequent conversations were extremely useful and the participants and I felt we knew each other and the process better and a relationship of trust had been established which made the conversations more satisfactory for both of us. I also had informal chats with the participants as I met them in the college which provided more data and contributed to a more complete understanding.

The interview is a unique situation and it can offer the participant as well as the researcher great benefits (McCracken, 1988; Richardson, 1998; Charmaz, 2006). It offers an opportunity for the participants to reflect on their experience in a way they probably would not normally do. It allows the interviewer to probe the interviewees' thoughts and feelings and to get beneath surface descriptions and generally to go further and deeper than normal everyday conversational would allow. It is therefore, a situation that requires great care and awareness of the potential difficulties that it may cause for the participant. This leads to a consideration of ethical issues that may impinge on the work.

Ethical issues

I would consider it important to work with the general guideline of 'a research ethic of care' especially in a constructivist and emergent process where it is impossible to predict where the process will lead. Interviewing in a qualitative paradigm can put the interviewer and interviewee in potentially vulnerable positions. The nature of the process necessitates a focus on the participant and possibly an exploration of issues that may be sensitive or upsetting. The wellbeing of the participant is paramount.

I had asked students to indicate on the questioinnaire their willingness to take part in an interview. Although it was not intended that the people chosen for interview would necessarily be representative of the mature student population, I wanted to have both men and women and older and younger mature students to see what, if any, differences would be revealed. On this basis I selected a number of prospective candidates from the completed questionnaires and contacted them by phone or in person and asked if they were still willing to take part. One was chosen as he was recommended to me by a teaching colleague and he had also indicated his willingness to take part on the questionnaire. Two of those I contacted felt unable to commit to an interview due to college, work or family demands and one, having agreed, had to cancel due to a family emergency. In these cases I contact the next person on my list of those who were willing to participate. I discussed again the nature of the research and what would be involved in an interview and some of the issues I wanted to discuss with them (Appendix 4). I guaranteed anonymity and their freedom to change their mind or withdraw at any stage, even after the interview. I suggested a meeting to discuss it further. At this meeting we arranged (where possible) a suitable time for the interview and dealt with any other issues.

Before each interview I thanked the students for taking part and again outlined the nature and purpose of the research and their role in it. I told them about myself, my role in the college and the academic work I was engaged in. I stressed that their consent to participate could be withdrawn at any time during or after the interview, or re-negotiated. I was aware that I might need to re-establish consent during the process as the conversation developed. Having discussed the research and clarifying any outstanding issues I asked them to sign a consent form (Appendix 4) and to agree to the interview being recorded. None had a problem with this. Participants were assured that they need not answer any question they felt uncomfortable with and their privacy, anonymity and confidentiality would be protected by pseudonyms, altering details that could make them identifiable and by careful storage of the data. I explained too, how I intended to use the data and reassured them that the data and the participants would be treated with honesty, respect and integrity. I re-established consent before subsequent interviews.

As a member of the teaching staff in the college and the CDVEC, there were relationships with students and colleagues to consider. I was aware of my role as a researcher and a teacher in the college and the implications of this for the interview process. With this in mind I selected for interview only students I did not know, I was

not teaching or involved with in any other way. All the interviews took place in my office in the college although I offered the participants a range of possible venues or the option of deciding on one themselves. The conversational style of the interviews helped in developing rapport and making the interviews a positive and pleasant experience for the participants and for me.

Insider research

As I was carrying out the research in the college in which I work I was aware that, in ways, I was an 'insider researcher.' Rooney (2005, p.6) defined insider research as being 'where the researcher has a direct involvement or connection with the research setting.' Traditional positivist research was based on the notion of an objective researcher producing objective knowledge. In this paradigm the researcher was, of necessity, an independent 'outsider.' Insider research would be problematic for positivists as the insider would be considered to lack objectivity. More recently, there have been 'numerous challenges to the orthodox institutionalised forms of going about research' (Loxley, 2008, p.15) and the notion of an objective researcher was challenged by new ontological and epistemological models that maintain that truth or meaning does not exist independently waiting to be uncovered, but is created by individuals through engaging with the world in specific social and cultural contexts. Objectivity, in the traditional sense, is impossible as the researcher must also, inevitably, draw from his/her cultural and personal context. Thus a conscious awareness of the role of personal beliefs and values on the research process is of central importance and declarations of positionality are now generally included by researchers (Sikes and Potts, 2008).

The increase in insider research in recent years has been attributed to these changes and the acceptance of the relevance and value of the insider experience and the corresponding increase in the number of part-time academic programmes which made insider research a pragmatic and worthwhile endeavour (Mercer, 2007a; Coghlan, 2007; Sikes and Potts, 2008).

Although there have been some attempts to define insider research, most commentators (Styles, 1979; Brewer, 1986; Hockey, 1993; Griffith, 1998; Hellawell, 2006; Loxley & Seery, 2008) reject the idea of an insider/outsider dichotomy and

agree with Mercer (2007, p.5) that the concepts of 'insider' and 'outsider' are not straightforward and are not necessarily 'two mutually exclusive frames of reference' but are better seen as a continuum. Hellawell, (2006, p.488) suggests that the continuum has 'subtly varying shades of 'insiderism' and 'outsiderism' and that there may be 'not one continuum but ...a multiple series of parallel ones'. Even binary concepts such as male and female intersect with other social categories such as class, age and race.

Merton (1972, p.22) argues that 'individuals have not a single status, but a status set: a compliment of various interrelated statuses which interact to affect both behaviour and perspectives.' One may be an insider in some aspects of the research culture but not in others and one's position as insider or outsider may shift with different aspects of the research and it is possible 'simultaneously be to some extent and insider, and to some extent an outsider' Hellawell (2006, p.490).

I was aware of moving along this continuum as I was neither a complete outsider nor a complete insider. I was an insider as I was doing research in my own college where I am intimately familiar with the culture, its workings, its 'lived experience' (Brannick & Coughlan, 2007. p.69) and many of its personnel and as a mature student myself I share many of the demands felt by my interviewees. I was an outsider in that I am not a fellow student in the college and my position as a teacher in the college and a researcher sets me apart from the group. I was not researching the organisation or the practice of colleagues and did not experience 'role conflict' or 'role duality' which Coughlan (2007) describes as likely hazards of executives researching their own organisations.

As an insider I was operating from a position of privilege. I had the practical and psychological advantage of easy access and I did not have to familiarise myself with a different college and negotiate a new environment. I knew what the students were referring to when they spoke of the structure of the courses, assessment systems and other issues to do with being in the college. This 'preunderstanding' which Gummesson, (2000.p57) defined as 'people's knowledge, insights and experience before they engage in a research programme.' also contributes to initial sensitising

concepts. Being on site meant I also had valuable opportunities for casual and opportunistic access and follow up.

However, I chose as interviewees only students I had no involvement with in any way. Although some of the students may have seen me in the college, I was not known to them until I conducted the questionnaire and so, in many ways, I was an outsider from their point of view. Not being known to the interview participants prior to the research meant that we had no preconceptions, 'baggage' or expectations based on familiarity (Hockey, 1993, p.206). However, my familiarity with the college made me aware of the necessity of creating some 'strangeness' and trying to be conscious of my own assumptions and immersion in the culture and discourse of the college and FE. As a mature student myself, I was, to some degree, an 'insider' sharing an understanding with the students of the demands on adults in education and a commonality of age. They would frequently say things like 'you know yourself' to imply this common understanding. In such cases I was careful not to rely on unspoken assumptions and tried to tease them out from the interviewees' perspective. I was aware too that unsought disclosures or references to colleagues might result in ethical dilemmas, in practice this did not happen.

Overall, I consider that my position in the college worked to my advantage and the balance of 'insider/outsider' greatly facilitated contact with the participants and constructing with them an understanding of their experience. It contributed also to the development of a valuable rapport, a reflexivity that would 'explicitly include lived experiences' (Brannick & Coughlan, 2007, p.69) and a heightened awareness of ethical issues.

Data analysis

It is important to point out that the data collection and data analysis took place at the same time although they are discussed sequentially here.

Initial coding

The process of data analysis was conducted in accordance with the guidelines proposed by Charmaz (2006) outlined above. The interviews were analysed as soon as they were completed to help understanding and justify further data gathering. The

initial coding process is 'the first step in moving beyond concrete statements in the data to making analytic interpretations' Charmaz (2006, p.43). These codes stick closely to the data and allocate a label to each section that helps to separate and sort data and represents an 'initial opening up of the text'. Charmaz recommends coding using gerunds and initially coding with speed and spontaneity using short, precise, simple codes and being open to what the text may suggest. I found it to be extremely important to remaining open to the text as initial hunches developed. It was not so much a matter of ignoring the hunches, but of remaining close to the data and not forcing it into emerging categories.

Initial codes	Data from interviews				
Having a	my cousin was here, a few years ago, she was saying that it was				
personal	really, really good.				
recommendation	rearry, rearry good.				
Being aware of	I don't think I'll fit in, I'm 35 and I'm likeis it too late for me				
age.	and all this?really, really nervous.				
Being anxious,	and an unistearry, rearry nervous.				
not fitting in,					
Worried it's too					
late					
Always wanting	I always wanted to do it but genuinely never had the chance				
to go back	because of the job I was in.				
Not having					
opportunity					
Becoming	I was made redundant two years ago.				
unemployed					
Having a	I see myself differentlyI see myself more confident as well.				
different self					
concept.					
Being more					
confident					

Table 4.1: Example of initial coding process

The first interview offered some tentative themes that I pursued in the following interview, each interview being analysed and comparisons made for similarities and differences within and between interviews. After the first three interviews I paused to evaluate emerging themes and see what ideas needed to be followed up in the next phase. Revisiting interviews after a pause and in the light of new data lead me to new discoveries and emerging patterns became visible. Subsequent theoretical sampling

focused on the tentative ideas and emerging categories. For example in follow-up interviews I asked what 'confidence,' which came up strongly in the first session, meant for the individual. This led to the concept of 'revised self.'

I was surprised initially at what did not feature in the interview data. For example the participants paid very little attention to the subject matter on the curriculum and extra support mature students might need. The strong antipathy to younger students and the belief that 'it's their world' voiced in the focus group was not supported by any of the interviewees.

Focused coding

The initial coding process generated a large number of codes. Focused coding means 'using the most significant and/or frequent earlier codes to sift through large amounts of data' (Charmaz, 2006, p.57). Subsequent interviews confirmed some of the earlier trends and a more focused coding allowed the grouping of conceptually linked themes into sub-categories. The table below shows how this process of integration emerged.

Initial Coding	Focused coding	Category
	Sub-categories	
Leaving school early.		
Missing out		
Always wanted to	Unfulfilled potential	
Feeling unhappy in work	Change	Latent self
Children grown	The right time.	
Became unemployed		

Initial Coding	Focused coding Sub-categories	Category
Feeling under pressure. Not having enough time. Concerns about family.	Juggling	
Not fitting in Not being able for it	Anxiety	Emerging Self
Enjoying it		
Feeling supported	Enjoyment	
Academic success	Support	
Growing confidence	Achievement	

Table 4.2: Example of focused coding

This phase of coding was more conceptual and involved developing 'a feel' or sensitivity for what was emerging. It was not a straightforward linear process and required constant re-reading and listening to the interviews, 'constant comparison' of data and sometimes going back to the participants for clarification. Frequently occurring themes such as 'leaving school early' or 'always wanted to' and 'missing out,' while not loosing their individual resonance, were subsumed into the subcategory of 'unfulfilled potential' which seemed to capture the essence of what was going on in this data.

Theoretical coding

In this phase, sub-categories were integrated into the three major or core categories of Latent Self, Emergent Self and Revised Self which I sub-titled 'Finding a Voice' as many expressed their new sense of self in terms of being able to speak and having things to say. I thought that these categories captured what was happening in the data and showed the relationship between them.

	unemployed. Always	AgeAssignments	Revised self-concept
	wanted to.	Fitting in	Revised seif-concept
	Missed out. Liked school.	Fear of failureJugglingcommitments	
		Support	
Core	Latent Self	Emerging Self	Revised Self
concepts		3 3	'Finding a voice'

Table 4.3: Example of theoretical coding

Memos

From the beginning memo writing turned out to be a vital part of the process of coding and analysis. Writing memos, however roughly, helped stimulate emerging thoughts on what was going on in the data. Early memos often took the form of rough diagrams, notes and maps linking ideas, concepts and emerging categories. (Appendix 6). The freedom and provisional nature of memos allows unexpected connections and surprising ideas to emerge. Later memos formed the basis of the analysis. 'Memo writing forces you to stop other activities; engage a category, let your mind rove freely in, around, under and from the category; and write whatever comes to you' (Charmaz, 2006, p. 81). Not all memos proved to be useful but even these were part of the process of interacting with the data and ideas and sometimes indicated the need to revisit data or participants. Mills et al. (2006) suggest that 'Memoing is essentially a reflective process that also provides that researcher with an opportunity to remember, question, analyse and make meaning about the time spent with participants and the data that were generated together'. The often spontaneous and informal nature of capturing ideas in memoing meant that they were not sequential and many disparate ideas were often recorded simultaneously. Shuffling these memos was helpful in sorting ideas and relationships and facilitated what Glaser (1978) called the 'preconscious processing' of data and promoting reflexivity.

Reflexivity

A constructivist paradigm acknowledges the researcher's role in interpreting or constructing the subject's experience with them. 'Our subjectivity and wider cultural understanding, shape our response and interpretation' (Merrill and West, 2009, p.115). This demands a reflexivity and awareness of the researcher's own experience and background and how they might impact on the research process: "...the researcher filters the data through a personal lens that is situated in a specific socio-political and historical moment' (Cresswell, 1998, p.182). Having worked in the City of Dublin VEC (CDVEC) for many years and as a mature student myself, I did not come to this inquiry as a disinterested observer. I felt it important to examine my own motivations and see how my position, experience, views and biases might interact with the work of research to create interpretations of experiences. Part of the research task is to be able to distance myself sufficiently in order to gain an overview and remain open to new ideas. Reflexivity contributes to this by ensuring a focus on the assumptions inherent in my own approach and subjectivity embedded in the language and discursive practices of further education. Mills et al. (2006) suggest that the process of writing memos in the grounded theory approach is essentially reflective and makes clearer the 'researcher's impact on the reconstruction of meaning into theory'.

However, personal history and vocational experience as well as being inseparable from the researcher, can be helpful as it contributes to the development of sensitivity to the ideas emerging from the data. Charmaz (2006, p.16) speaks of Blumer's (1969) 'sensitising concepts' as giving the researcher initial ideas to pursue in the early stages of the research. She sees these concepts as 'points of departure' while emphasising the need to remain open to the concepts emerging from the data. More generally, Strauss and Corbin (2008, p.32) see sensitivity as standing in contrast to objectivity. 'Sensitivity means having insight, being tuned in to, being able to pick up on relevant issues, events and happenings in the data'. This sensitivity to the nuances and complexity of the data comes through an immersion in the data and the interplay between the data and the researcher's 'prepared mind.' However, it is essential to remember the primacy of the participant's perspective and the emerging data.

In a more general sense, all research decisions are filtered through our theoretical sensitivity, a 'unique combination of personal and professional experience, and your

reading of relevant literature' (Ryan et al., 2006), so the researcher is obliged to be reflexive at all stages of the research and analysis. Reflexivity can be seen as an ethical requirement in qualitative research which acknowledges the subjectivity of the researcher and his/her role in the construction of knowledge. The researcher's obligation to make clear where they are 'coming from' is a fundamental aspect of the transparency and integrity of the work.

Although conscious of remaining open to the data, I embarked on this research with sensitising concepts formed from experience and reading. The story I outlined earlier about the mature student who was 'upset' by a discussion in class had prompted me to think about how learning, new information and new ways of thinking could be troubling. I had found Bowlby's ideas on attachment and the 'strange situation' as offering a useful way of thinking about this. Linked to this and the notion of the 'teachable moment,' Mezirow's 'disorienting dilemma' and the transformation of frames of reference and the concept of 'troublesome knowledge' (Meyer and Land, 2003) contributed to my sensitivity.

My view of further education as greatly influenced in its practices and policies by neoliberal thinking also informed my thinking. Also, perhaps somewhat paradoxically, my belief that further education has an important role in addressing disadvantage in welcoming non-traditional students, and providing an alternative route to higher education. I found the concepts of class, habitus and capital to be useful tools in thinking on these issues and how social, cultural, economic and personal factors interact in decision making and educational choice.

These ideas initially were helpful in considering the study and the early data. As the research developed, some of these 'tentative tools' proved quite useful and other, different tools were called on to help in understanding new, and often surprising, emerging ideas.

Being a researcher in an area with which I am familiar also poses the danger of over-familiarity and emphasises the need to interrogate the 'taken for grantedness' of its practices. McCracken (1988, p.22) speaks of the necessity of 'manufacturing distance' so that we can observe with critical awareness situations with which we are

very familiar. He suggests that a researcher's 'deep and long lived familiarity with the culture under study', as well as having the capacity to dull the investigator's powers of observation and analysis, is also potentially a huge asset. What is needed, he says, is a 'cultural review' – a conscious reflexivity which promotes familiarization and defamiliarization which allows the best use of the 'self as an instrument of enquiry'. Eisner, (2001) speaks of the role of the arts in creating 'perceptual freshness ...also critically important features of good qualitative research'. A grounded theory approach is particularly useful, I think, in helping to avoid focusing on preconceived ideas based on familiarity.

Engaging with the Literature

In defining the components of grounded theory analysis, Charmaz (2006, p.6) includes 'Conducting the literature review after developing an independent analysis'. The place of the literature review in grounded theory is a matter of some contention. The argument against conducting a literature review at the outset is based on the idea that it may influence data collection and analysis (Glaser and Strauss, 1967) and force data into pre-existing categories. Charmaz (2006, p.165) suggests that the reason for delaying the literature review is to 'avoid importing preconceived ideas and imposing them on your work. Delaying the review encourages you to articulate your ideas'. Charmaz (2006, p.16) also acknowledges that researchers' assumptions and disciplinary perspectives alert them to look for certain possibilities. These 'sensitising concepts' are useful in providing initial ideas to pursue. However, not all researchers agree and suggest reasons for early engagement with the literature. Morse (2001, p.9) suggests that the literature should not be ignored but 'bracketed' and used for comparison with emerging categories. Coffey and Atkinson (1996, p.157) point out that the researcher who is not aware of the relevant literature is in danger of rediscovering the wheel. Charmaz (2006, p.167) finds the term 'theoretical agnosticism' apt and agrees with Glaser (1978, 1992) that the time to engage with the theoretical literature is when the categories and relationship between them has emerged.

While it is naïve to assume that the researcher is a 'tabula rasa', the current study is exploratory and prior reading was very general with the intention of providing a context and overall familiarity with the area. 'Researchers may not know which

literature is relevant until analysis is well advanced' (Locke 2001, p.124). Engagement with the literature at this stage was directed by the emerging concepts and ideas following Charmaz's advice (2006, p.164) to 'Draft your literature review in relation to your grounded theory'. In this study the engagement with the theoretical concepts that relate specifically to the research findings takes place in Chapter 7 following the presentation of the findings.

Conclusion

The decision to use a questionnaire with qualitative interviews and a focus group was influenced by some of the reports reviewed in Chapter 3. I thought the combination worked well to give an overview of the area as a context. Using a grounded theory approach to data generation and analysis proved to be very time consuming with implications for the management of the whole project. However, it proved to be extremely useful and yielded rich data which will be discussed in the following chapters.

CHAPTER 5

RESEARCH FINDINGS

Introduction

This chapter outlines the findings from the questionnaire, discusses the focus group and presents profiles of the participants, each of which contributes to providing an overall picture of the experience of mature students and introduces in a personal context many of the idea discussed thematically in the next chapter. In keeping with constructivist approaches, I use direct quotation from the participants in this chapter and the next to enable their accounts to remain visible in the final text so 'that the reader can make a connection between analytical findings and the data from which they were derived' (Mills et al., 2006, p.11).

The questionnaire

As mentioned in chapter 4, Methodology, the main focus of the study is qualitative, using interviews, however, I also decided to carry out a questionnaire to get an overall picture of the mature student population in the college and see what issues arose. The questionnaire, which one hundred mature students completed, yielded the following information:

Age

Only seven of the one hundred respondents were aged over fifty while eighty were under 41 years of age. Fifty-seven were between 21 and 31 year of age and twenty-three were between 31-41 years. This is similar to the findings of Fleming and Murphy (1997) and Watson et al. (2006). Most mature students are young and single. However, the numbers of older mature students has been increasing and with the current recession set to continue this increase is likely to accelerate. This research focuses on mature students who have been out of full-time education for some years.

Gender

Fifty-nine respondents were female and forty one were male. Nationally, the participation rates in FE are seventy percent female, thirty percent male (Watson et al.,

2006). I suspect that this ratio has changed with the economic downturn with fewer jobs available to males, especially in the construction industries. There is also some regional bias as not all areas have the same FE provision and some colleges offer courses that traditionally attract one gender almost exclusively, for example child care. In this college fifty-two percent of the total student body is male.

Marital status

Sixty-five respondents were single, sixteen were married and fifteen were co-habiting and thirty-one had children. I suspect the number of single students is somewhat misleading as many in long-term relationships would tick 'single' as they are unmarried, whether co-habiting or not.

School leaving age

Twenty-four left school at 16 or under, the rest at 17 or 18. This supports the idea that many mature students are 'delayed entrants' having put off the decision to progress immediately after school, lacked the necessary leaving certificate results or those who became unemployed having started working immediately after school. FE is an important provider of 'second chance' education and alternative progression routes to those who left school early or who were not qualified to apply to Universities or Institutes of Technology.

Secondary School Programmes Completed

Nine of respondents left secondary school with no qualifications, eight had Junior or Intermediate Certificate only and seventy-eight had Leaving Certificate, Leaving Certificate Vocational Programme or Leaving Certificate Applied. This reflects the age of the cohort surveyed. Older students are more likely to have left school with no qualifications (Watson et al., 2006).

Course in FE

Eighty-five of the respondents were doing a two-year course, forty-seven were in year one, fifty-three in year two. Sixty-seven had previously done a Level 5 course. Seven already had a Level 8 qualification. Eighty-two of the one hundred intended to progress to another course.

Parents' educational qualifications

Twenty-two of the respondents' mothers had left school at primary level and twentythree of fathers.

Socio-Economic Group		Father	Mother
		Percentages	Percentages
Employers and managers	A	3	1
Higher professional	В	6	1
Lower professional	С	1	8
Non-manual	D	22	29
Manual skilled	Е	21	1
Semi-skilled	F	5	1
Unskilled	G	8	7
Own account workers	Н	15	8
Farmers	I	2	1
Agricultural workers	J	0	0
All others gainfully occupied and unknown	Z	0	0
Left blank/not applicable/none		17	18
Housewife / homemaker		0	25
Total		100	100

Figure 5.1 Father's occupation and mother's occupation [by percentage in the various socio-economic groups]

It could be argued that parents' occupations are less relevant when discussing mature students, especially older mature students. However, this questionnaire was part of the initial exploratory research and was done to get an overview of the entire mature student population and to see what issues arose.

I used the categories A-Z outlined in the Central Statistics Office's (CSO) 'Census 2006. Appendix 8. Occupations' and added the category of housewife/homemaker and a category for those who did not answer this question or said 'none' or n/a.

The single largest group is non-manual, which as is pointed out in 'Hidden Disadvantage?' (McCoy et al., 2010) is diverse, composed of a large range of occupations from relatively high status positions to lower level service workers. The

'other non-manual' section of the group share many similarities with lower manual groups across a range of educational characteristics.

These findings, although not broken down into intermediate and other non-manual, also support the contention in McCoy et al. (2010) that 'clear socio-economic disparities exist with the high levels of entry into this form of post-school education evident among young people from other non-manual backgrounds and skilled manual backgrounds' (2010, p.137). It is interesting that this applies to mature students also, although many of this sample are younger mature students who would share most of the characteristics of students a few years their junior. It would be interesting, but beyond the scope of this work, to look at the distribution of various social groups among different courses in FE. There is great demand for some courses and entry is quite competitive. Others have less exacting entry requirements, especially when students are needed to fill classes. As well as my own intuition and anecdotal evidence, there is some research which suggests that 'those from lower socioeconomic groups are concentrated in courses which are less prestigious and consequently those which offer lower occupational status prospects' (Lennon, 1998).

Financial support

Twenty of those surveyed said they work part-time, eight mentioned parental support, ten were supported by a spouse and twenty-nine were on VTOS or Social Welfare, including the Back to Education Allowance (BTEA). Many of the younger mature students live with their parents. I suspect that some do not consider this support. In general, there are fewer students working part-time now than during the boom years. One student claimed to be working full-time while at college. VTOS and BTEA are essential supports and their availability is often a crucial factor in decisions to return to education.

Reasons for coming back to education

Twenty cited interest as their reason for taking their course. Seventy said they came back to get a qualification.

Although a qualification was the stated motivation of seventy of the respondents, this is not the whole story as many other motivations and circumstances impact on the decision to return. This is further examined in the interviews.

Reasons for coming to a Further Education College

Nineteen of the respondents said they wanted to do a specific course only available in FE. Six said they wanted to try out education having been out of it for a while and sixty-eight gave the level of the course and the structure of FE as their reason.

FE offers many courses not available in other sectors or courses with a specific vocational orientation. Colleges are often known for certain vocational areas such as animation or theatre and attract students because of their reputation. Fees in FE are considerably lower than HE or private sector providers. Many occupations now require a qualification where formerly they did not. For example workers in crèches now need a level 5 award.

The vocational approach of FE courses suits many students and employers and in many cases offers progression to HE if this is wanted. Some FETAC one-year courses, although vocational in structure and intent, are not, in many areas, sufficient preparation for direct entry to the workplace. Many students, especially mature students, do a one year course because they lack the requirements for a two year course or because they want to see how they like the subject area and see how they get on.

Included in the sixty-eight students who cited the level and structure of the course as their reason for coming to FE would be a considerable number who did not have the qualifications to go to HE directly and those who saw FE as an intermediate step in progressing to HE. These students wanted to see how they got on in FE and felt the experience would give them more confidence and they would be better equipped to cope with HE when they got there. Many school leavers who are qualified to go to HE come to FE for the same reason. This supports the findings in 'Hidden Disadvantage' (2010, p. 138) which also mentions that there are serious issues over the opportunities to progress from FE to HE. Many see FE as more 'user-friendly' and less intimidating than HE. It represents less of a time commitment and smaller financial investment. FE

is perceived as being more personal and supportive than HE. There is often less cultural distance as the colleges are often familiar to students from the area and they would have known people who attended and in some cases would have done their Junior and Leaving Certificate there themselves (see question below). Those who attended a vocational school are two and a half times more likely to participate in FE (McCoy et al. 2010, p.122). Put another way, the habitus of the colleges are not as foreign to them as the institutional cultures of HE.

Reasons for coming to this particular college

Sixty-one of the one hundred students surveyed said the reputation of the college, twenty-five said a friend recommended it. Seventeen mentioned convenience as a factor. The interviews confirmed the importance of personal contacts, being aware of the college in the neighbourhood and its convenience. Only twelve mentioned career guidance which I found surprising given the large number of younger mature students and effort put into promoting CDVEC further education colleges among guidance counselors in secondary schools and elsewhere.

Experience of FE – most enjoyable aspects

Sixty-five of the respondents said the course was the most enjoyable aspect of being a student in FE, forty-five said the social aspect. Thirty-one said the pleasant atmosphere and thirty-four said the friendly and approachable staff. Many students from the other non-manual group are more likely to have disengaged from secondary school and left unsure of what they wanted to do. When work was available during the boom years, this was an attractive option. Becoming unemployed is crucial in the decision to return to education. The climate of FE is significantly different to secondary school and many younger and older mature students find this a welcome surprise. First names are used for students and staff and the atmosphere is friendly, supportive and encouraging. This is mentioned repeatedly in the questionnaire and interviews as being very important in the decision to come to FE and stay.

Experience of FE – least enjoyable aspects

Thirty-six of the respondents said lack of money was the least enjoyable aspect, twenty-eight said juggling other commitments, twenty-one said assignments and ten said younger students.

Some students had left jobs to come back to education, most came because they had become unemployed. It is likely that they are comparing their financial situation now with when they were working. Students on VTOS and BTEA keep their benefits and

receive other allowances. They are not worse off because they are in education.

Summarise your experience of FE

The respondents were overwhelmingly positive in their comments on their experience

of FE and their decision to return. Only one was disappointed.

Conclusion

The questionnaire was very useful in giving a broad overview of the mature student

body in the college and the issues affecting them. It is, inevitably, interpreted in the

light of my own experience, knowledge and sensitivity. The stories of individual

students that unfolded in the focus group and interviews provide a more personal

account of the experience of being a mature student.

The Focus Group

As mentioned in the methodology chapter, I conducted a focus group made up of

students I knew or who were recommended to me. This was part of the initial broad

exploration of the area 'to see what came up.' I had some general issues from my

reading and experience to prompt discussion if necessary and I hoped the interaction

within the group would provide issues that could be followed up later in individual

interviews. I started by explaining that I was conducting some research on the subject

of mature students and I was hoping to hear what they considered important issues and

something of their own experience. In this section I want to give a brief account of the

conversation and the main issues that arose. There is a more detailed discussion of

these topics in Chapter 6

The participants were: (Pseudonyms are used)

Lisa

A 29 year-old woman who had just finished a two year diploma in

Social Studies, a single parent with two children.

91

Brenda A 48 year-old woman, married with grown up children who had just

finished a two year diploma in Radio having previously done

a one year Return to Learning course. About to progress into year 3

of a degree course.

John A 42 year-old man who had just finished a one year FETAC Art

course. He is not married and has no children. He intends to progress

to another course next year, if he continues to get the Back to

Education Allowance.

Only one of the three had come back to education with a particular vocational goal in mind, and that had changed since. They all had a long-held desire to come back and all said they loved the course they were doing. For Brenda, the poor economic situation and her children being independent gave her the opportunity and incentive to come back. She did the Return to Learning course as she had been out of education for many years and wanted a gentle introduction. She had not, at that stage, envisaged continuing. John had been feeling unhappy in his job and had been saving for some time with a view to returning to education. Like Brenda, he did not have a particular vocational focus but decided to take a course that he thought would interest him. Lisa wanted a specific course as she knew the general area she wanted but changed the specific focus during her time on the course. Two spoke of the financial hardship, having to work part-time, and take care of children and all talked of the necessity to manage their time well. Lisa worked part-time 16-24 hours a week. Responsibilities and financial issues were two of the major issues they saw as separating them from younger students.

It's one of the drawbacks of waking up a third of the way through your life and saying 'I think I'll go back to school'. What'll the bank manager think? (Lisa)

Lisa, who has children in school, said that balancing domestic and college responsibilities was a big issue and that it was impossible to separate from concerns about children and this often impacted on college life.

It's the responsibility as well. You could be sitting here and much as you shouldn't have your phone on, you have to keep it on. You might have to leave. (Lisa)

She also felt guilty about depriving her children of her time and attention and decided to postpone her place in HE.

I owe my children two more years. I was offered a place in Maynooth and I actually turned it down because my daughter is ten and she would be going into first year when I'd be doing my final exams. It's her turn now. (Lisa)

She was particularly critical of younger students, and had the support of the others.

I found the younger people in my class drove me insane. They drove me absolutely crazy to be honest and I'd say I speak for all the mature students. The young people in my class, they drove me nuts. (Lisa)

There were quite a few people there just to waste the time, basically. (John)

They found that many young students lacked the focus they had and many were badly behaved.

I have to be honest, I felt like a freak...when you're around young people... totally different attitudes, different outlooks...(Brenda)

They agreed that their experience and knowledge of the world and the workplace set them apart from younger students. When prompted they agreed that an age balance was probably a good thing and there were aspects of younger students they enjoyed and ways they learned from them.

I found there were a lot of younger people in my class so you feel like you're taking care of them ...but I ended up learning a lot from them. Because I wasn't as up to date on computers as they were...they were whizz kids, like computers were second nature to them. I did, I learned a lot from them. (Brenda)

Another topic they all agreed on was that this (college) was their world that they were invading.

But isn't it their world though? We're invading it ...this is the natural progression for people of their age. We're coming back...they're straight out of school going to college...this is their world. (Lisa)

Absolutely...because. like I mean...when I was...when I got the course I was...at the back of my mind I was thinking ..God, you know I've probably taken a child's place. (Brenda)

I was surprised that his had arisen and that there was such unanimity and strong feelings. I pursued this later in the individual interviews and none of the participants shared this view. Neither did I come across such strong feelings about younger students more generally. It was acknowledged that while they could be annoying, it was just their age and you couldn't expect them to have the same outlook and focus as older people. Similarly, none of the interviewees agreed with the focus group's suggestion that it would be a good idea for mature students to have a special room set aside for them. They saw this as unnecessary and potentially divisive.

If the mature students had their own room like you could go up...not all the time, just sometimes. (Brenda)

Yeah, and if there was someone there you could talk to. We'd know why that room was there like...(Brenda)

They all agreed too, that they had found the college very supportive. They enjoyed college enormously and benefited from it.

It was better than I thought it was going to be. I'm really happy that I did come back. (Lisa)

I suppose it's just being confident about myself ...it's made me believe in myself I can accomplish ...I'm not so stupid like (laughs) (Brenda)

I learned so much about who I was. (John)

yes, it has changed me, I'll never go back to the person I was before I started college. (Brenda)

The focus group was useful in that it raised issues that I explored later in the interviews. The idea of 'unfinished business' was prominent and a long held desire to revisit formal education as well as the vocational motivation, however vague and unformed in some cases. Certainly all the participants felt they had enjoyed being in further education and had changed as a result. This was expressed largely in terms of confidence, self image and 'seeing things differently.' This tended to be a general feeling not connected to any specific new skill or competency.

Time management and finance were issues especially for those with children and there was agreement that mature students were more focused. The strength of the feeling of

annoyance with younger students surprised me as did the notion that they were 'invading their world'.

The focus group format did not allow for in-depth exploration of topics as I was conscious of not asking about personal issues or probing deeply in this context. The participants said they were glad of the opportunity to discuss these issues and found the exercise beneficial.

Profiles of the participants

In this section I want to give a brief biography of three of the mature students I interviewed, two others, Peter and Caroline, are discussed in greater detail. I believe their stories contribute to understanding some of the diversity of mature students in further education and some of the themes they share. The first three students, James, Vera and Daniel, were selected at random and Peter and Caroline as, by chance, I had more opportunities for discussions with them.

Where possible I have tried to use the students' own words to tell their stories. I was anxious that the wholeness and integrity of the stories would demonstrate the complexity of their histories and experiences and the many personal and structural factors that contributed to shaping them. It is not intended that these profiles contribute to generalisable conclusions but that they offer personal accounts and insights and make real the sometimes abstract discussion of 'mature students' and the factors that shape them and contribute to the overall aim of exploring their experience. I was extremely fortunate to have had the opportunity of a second interview with most and other conversations to clarify points and work with the participants on interpretations of the material. I have been careful that the participants would not be misrepresented, however, as West (1996, p. 32) points out, 'an understanding of these learners and such experiences requires more of an imaginative empathy than a cool detachment: an ability to immerse oneself sympathetically in another's world in all its messy and contradictory confusion.'

At this stage I tried to avoid overly interpreting the accounts or engaging with the literature as the intention is to remain close to the personal accounts of the participants. Their stories introduce in a personal context many of the themes discussed later.

Name	Age	M/F	1or 2 yr course	In year 1or 2	Course	Previous courses	Trigger
Caroline	35	F	2	1	Social Studies	None	Became unemployed
James	56	M	2	2	Business	RTL FETAC	Became unemployed
Vera	49	F	1	1	Return to Learning	None	Became unemployed
Peter	55	M	-	Now in HE	Ex-Social Care	FETAC BTEC	Unemployed and opportunity
Maeve	50	F	1	1	Art	Various	Children independent
Abbey	35	F	2	2	Social Studies	FETAC	Time was
Aoife	32	F	1	1	Return to Learning	None	Became unemployed
Daniel	53	M	1	1	Business	FETAC	Became unemployed
Jason	61	M	1	1	Business	FETAC	Became unemployed

Figure 5.2 Interview participants

James

James is a fifty-six year old local man who had worked in the same business since he had left school at sixteen until a few years ago when the firm closed down. He is single and lives locally. He liked school and was told he was good at it but 'didn't get a chance to finish.' He says he came back to education because he got the opportunity. An advisor in Obair, who he knew through football, recommended he come to the college and this was the first time he had considered it and thought 'it might be interesting.' He had often passed the college but had never been in it. He found the college welcoming and enrolled on the Return to Learning course. He saw this as 'a chance to finish what you didn't get a chance to do at the time'. He enjoyed the course very much and was particularly pleased to develop a competence with computers. At the end of this course he was keen to progress and was accepted onto a FETAC

business course and from there went onto a two-year BTEC business course. He finds the course demanding but likes it very much. Part of the difficulty was overcoming the barrier of the terminology.

... getting used to all that terminology, I think that's what it is, it's getting used to all the terminology..if it was all written in plain English I could understand it better.

He is extremely positive about his experience of FE which has exceeded his expectations, and he spoke particularly about the social element, the importance of being known in the college and the supportive staff. He feels that his horizons have broadened considerably since he came. He felt confident enough to get involved with the local community radio station and he now has a regular sports programme of which he is very proud. Overall he feels his social circle and competencies have broadened and his social and cultural capital have increased in ways he had not envisaged.

Confidence, yeah. And doing those programmes on the radio has helped and the course, meeting all the people in the college, all of that has helped, I suppose, to bring out the real me.

When I asked about 'the real me' he spoke of being at a 'bit of a low ebb' before coming back. Now he says he's 'back to what you could call me, and a little bit further.'

He is tolerant of younger people although he feels some are immature, the mixture can be very good.

James is thinking of progressing to an IT when he finishes, which was not part of his thinking when he was coming into the college. He frequently mentioned 'taking one step at a time' but now he is aware of possibilities that never occurred to him before.

Vera

Vera came to do the Return to Learning course when the local business she had worked in for many years closed. She is very focused on getting work, 'more of a career rather than a job' and sees this as a step towards that. She came to the college

when it was recommended by an acquaintance and had not been aware of such courses anywhere.

Vera grew up in Dublin and left school at sixteen to do a secretarial course with a view to getting a job. This was the norm for her at that time. 'You weren't really geared towards a career, especially with girls.' Of six children, only one did the Leaving Certificate, the others did apprenticeships. Vera did reasonably well in school but was keen to get a job.

I never would think of going on to college, at that time, I didn't think I'd want to be a teacher or a nurse or anything like that. It just never entered my head that that's what I'd want to do or think about doing.

She always wanted to go back to education especially when she saw how her children were being taught in school compared with her experience, 'It was just so different and enjoyable'. She also felt uninformed and when the business closed she wanted to have 'something on my CV'. She had hoped to come back the previous year but took the year out to recover from illness. She was nervous about coming back and conscious of her age and being 'older than the teacher' and feels that her classmates are asking the same question she is, 'what's she going to do with this?' Not to 'do something' with it would be a waste for her and she was hoping to progress to another course next year and was taking it 'one step at a time'.

She's enjoying the course and once she got over the hurdle of the first assignment felt a sense of achievement and confidence.

I know what I'm doing, this is brilliant, I actually know what I'm doing, the answers are flowing out, and the words are flowing out...

Her family are friends are encouraging and she thinks her 'dinner table conversation has improved'. She is not irritated by younger students, 'I've five of my own, it's like home from home', but she does consider herself more focused and diligent than the younger mature students in her class and thinks they might see her as 'a bit of a swot.' She sees the course as 'a bit of a luxury' but at the same time, a necessity, if a better job is to come at the end. Being without income is a sacrifice but she is enjoying the

course more than she expected. "You kind of feel a little bit guilty. You know, I say 'I'm loving this' at the same time."

Vera sees becoming a grandmother and coming back to education as having changed her outlook. She is now conscious of her legacy, 'I want to leave something behind' and feels her education allows her to do this differently. 'It wouldn't be there without the education...it would be just the usual passing on motherly and grandmotherly things.' It seems as though she does not value these 'grandmotherly' ways of knowing very highly. She feels more confident that that knowledge is now validated. 'It's nice to know that sometimes your instincts were right'. She also feels she can relate to her sons differently.

I even know now, having a conversation with my sons, the older ones especially, that I'm more in tune with them and I can talk about things that I wouldn't have talked about before. They would have been just mother things, you know, now it's more.

Vera's stated motives for coming back were entirely vocational and that is still her main focus, 'I genuinely have to get a job. That's the priority' she was surprised at how much she gained from the experience.

I didn't think I'd have the appetite I have for the education. I thought it was just a means to an end, that's the way I was looking at it in the beginning.

Daniel

Daniel left school at sixteen. He is now fifty-three. He was made redundant from his job and did a two-year night course in an IT and then a one year FETAC course. He was hoping to get a job in a similar area as he had a lot of experience but realised he would need 'some papers as well.' He had a bad experience of school with 'the brothers' and for secondary school he went to the 'Tec' which was a considerable improvement. He left after the Group certificate.

He came back 'to up-skill' but found the social side of college hugely important. One of the main disadvantages of being unemployed for him was the lack of contact with others, 'it's just like a silence...and I just hate that.' He enjoys college greatly although he finds some of the subjects difficult and frustrating and thinks that extra subject support for those in difficulty would be of special benefit to mature students.

He feels his ability is improving and his academic ambitions are growing, 'I haven't

got a distinction yet, I'd love a distinction.' He prefers to get work done in college and

stays late to avoid bringing assignment home. 'Like a nine-to-five job: this is my

work'. He thinks a lot of the younger students are 'there for the laugh' but is very

tolerant; 'you understand that's their age, we were all like that at one time.'

Although he wants to get a job, he would not consider his time in education wasted if

this were not to happen 'education is never a waste of time.' He finds he knows

'what's going on' and can understand current affairs very much better and has more

interest in them.

When I hear people talking about why they're doing things...like the elections

there...you might be sitting in front of the TV and you'd understand what's

going on, that's important. You know what's going on.

Another example of his new confidence, he said, was his willingness to question and

challenge things he just would have accepted previously. Daniel is not considering

progressing after this course as his priority is to get a job. He feels he is much better

equipped to do so now.

Profiles: Caroline and Peter

Caroline

When I spoke with her to arrange the interview, Caroline struck me as a confident,

pleasant, direct woman. The interview took place in my office in the college. Caroline

is 35 years of age and had been made redundant from her job with a government

agency. She felt she had been quite good at school and had liked it. She left early

because she was offered a full-time job in a supermarket and the family needed the

income. She worked there for eight years before going into the care area. She wanted

to continue to work in the care area and realised she needed a qualification to do this.

She is very committed to care work and had done voluntary as well as paid work with

various organisations catering for homeless men, the elderly and people with

disabilities.

I've always worked with people, you know? I don't know why the care area, I

think it's because there are so many vulnerable people

100

She was familiar with the college as she lived in the area and had started studying for the leaving certificate there before she left to work and felt she had a good experience in the college. She has two children of 12 and 14. Her family, especially her father, valued education and praised her achievements at school.

We were always very encouraged with education, from me father, like you know, he was devastated that I'd left school. His emphasis was always on education.

Her father had been prevented taking up a scholarship to secondary school because of poverty, but had always maintained an interest in education. He was an avid reader and had a range of friends that Caroline thought was unusual for a man of his class. He clearly had a big influence on Caroline.

My father was very very intelligent, you know...I mean he was fluent in Irish and he still, he read books right up to the time he died. He was very aware socially...

I asked Caroline if getting a qualification was the only reason she had come back to education.

It was always something I said I was going to do. To be honest with you, I feel that if we had been pushed more in schools back then, to stay in school, and we had've been made more aware of what the benefits of staying in school were, you, ...You know, maybe it would have...

This was a significant and common theme in the interviews. Some felt they had 'missed out' by not having the opportunity to continue in education at the time or said that they had not been encouraged by schools that had limited expectations for children from their area. Some had bad experiences in school. Common also was the coming together of external factors and internal motivations creating 'the right time'. Caroline had the internal desire to come back to education and this coincided with being made redundant and her daughters being at an age when they were both in secondary school giving Caroline the opportunity to avail of full-time education. '

Despite feeling she had missed out, Caroline was adamant that she benefited more from education by being older and appreciated it more. Unlike younger students who, she thought, see assignments as a pain, she saw them as a 'learning opportunity.' She felt that some of the younger ones did not appreciate the opportunity they had. In general she thought the various age groups mixed well and was against any special facilities being provided for mature students as it might lead to separateness. She felt too, that having experience in the care area she had an advantage and the academic discussions of policy meant a lot more to her.

... the likes of policy and procedure...like some of the care areas I did work in, we wouldn't even be told, do you know? I think it's really opened me eyes, definitely.

Later, when speaking about her work experience and college studies about addiction:

Well, seeing things differently, definitely, em ... it has definitely changed my attitude, it's definitely been a big, big eye-opener.

She also felt she had developed greater empathy and more social awareness. She feels her daughters have benefited too.

I think another point is that my children being the age they are, I can share my knowledge with them and it has greatly benefited them also, it has made them more aware and more empathic as well.

Although Caroline found the college hours fitted in well with her children's school times, there were occasions, particularly to do with work experience, when there were problems. She developed the strategy of involving her daughters with her work and prepared them for when her busy times or assignments are coming up. Like many mature students with young children, she waits until they are in bed before starting her college work.

I wait till the kids are in bed before I start, because I don't want to be sitting there on a computer all day...the only disadvantage is that you can get carried away and sit up till half four, not realising I was up in three hours...

This has obvious disadvantages and creates its own stresses and despite the good timetable in college, juggling was still a problem. In a follow up interview Caroline was more direct about the difficulties.

I feel I'm failing as a parent because. ...I'm trying to cram in assignments too. I'm not putting as much dedication into the kids as I normally would.

All students with young children admit to a degree of balancing. Many have to drop children to school and are late for college. It is common for many students, of all ages, to miss days or have to deal with issues to do with children. The Return to Learning course, which caters specifically for mature students, starts classes later in the morning to facilitate the school run. Students in other courses who have children have to make their own informal arrangements with individual teachers or course tutors. Many put off coming back or progressing until children are older. Caroline felt that mature students should have an input into the organisation of work experience as this could prove especially difficult to balance. In general, though Caroline felt that there was great support from the tutors in the college.

Well, I feel there's great support here, to be honest with you. I find that if you've any problems at all, like if you go to M (course co-ordinator) or N (tutor) like, they're great; I think there is great support.

She sees the support and the general atmosphere as a key component in learning. She contrasts her experience of FE with her own strict schooling and with friends who went to UCD where she thinks she would have been 'in way over my head.' Other mature students speak highly of the support and encouragement in the college. They speak of the overall atmosphere and the college as a place they enjoy coming to. They frequently mention being able to talk to tutors and felt that being know and addressed by name was a major factor in facilitating their learning and greatly enhanced the overall experience. They are aware that the same level of support and recognition may not be available in HE. Care and support are significant themes that will be examined later.

Academically, Caroline is now confident and like many mature students saw the first assignment as a major hurdle and landmark in her career in college. Although she had plenty to say, it was not knowing exactly what was expected and structuring of the work that caused her great anxiety.

...although I'd be really good at writing words, you know like, it was putting it together, the structure of it, quite challenging, in the beginning because I was so unsure of myself as well, constantly asking myself 'is this going to work?'

The first assignment in college is very significant for mature students. An assignment is graded and carries with it all the connotations of being assessed and judged in what

for most is a strange or new area demanding different skills they feel they have not got. It is generally a written piece produced on a computer, which can in itself be a big challenge, and many feel that they are exposing what they consider their weaknesses in this area. It can cause great anxiety and can be important in determining whether a student, of any age, will continue. This is so even in the climate of encouragement and support that was mentioned. It can also be, and generally is, a major confidence booster and most students, like Caroline, are pleasantly surprised by how well they do.

and when I'd be getting them back I was getting all distinctions, I think I was kinda shocked to be honest with you, em ... really pleased, but shocked, like I'd really underestimated my own ability.

This success marks an acceptance into the student role and a feeling of entitlement to be there. The new confidence in academic prowess carries over in other areas of students' lives and is the most often cited benefit of returning to education. Caroline said she came to the college because she wanted a qualification to work in the care area. She feels much better equipped for this but acknowledges that the benefits go beyond the immediate vocational motivation.

... even if I didn't work in the social care system, it's not something I'd regret. It's definitely something that I feel has kinda opened my eyes to people in general, it does make you a better person.

Peter

I interviewed Peter in my office in the college. He is a pleasant, good humoured thoughtful man of fifty-five and is currently doing a course in a HEI having spent three years in the college. He did a one-year FETAC course followed by a two-year course in Social Studies. He was the eldest child and grew up in a large town in the south of the country where his parents were well known and respected business people. They ran a family firm which had existed for generations. His father had been studying law but gave it up to run the business when his father died. Peter said the business demanded much of his parents' attention and as a result his upbringing was 'difficult'. After a very unhappy initial education with the Christian Brothers he was sent to a well known boarding school in Dublin. He had a good relationship with his father, 'a thoroughly decent nice man' but not with his mother, 'a matriarch'. Peter was dyslexic, which was then unrecognized, and he had a very bad experience in school where he was physically and psychologically abused and written off as a

'dunce'. His family, especially his mother, had a strong work ethic and were equally unsympathetic. 'Because my results were poor it was my fault. I was getting it from two sides' He was taken out of school at sixteen and went to Kevin Street (Dublin Institute of Technology) where he did a diploma to acquire skills that would be useful in the practical aspects of the family business. He was surprised he got through as he was 'completely unable to read or write.' He described his school and college career as 'a nightmare'. He knew he was not fitting into the expectation of his family and class where it was presumed that he would go to college. At home, his illiteracy prevented him from doing anything other than manual work and he was treated as 'the lad at home, the boy.' His position was made worse when contrasted with the academic and business success of his younger siblings.

...people who qualified were treated better because education was looked on as something; it was valued, and the fact that I couldn't accomplish it...I was undervalued as a result of it.

Peter worked very hard, 'often sixteen hours a day, six days a week' in this capacity in the business for nearly thirty years. In the first interview he glossed over this in a sentence and moved on to the next stage of his career. In a follow up interview I asked him about this and he explained how his life had been affected by his illiteracy, and family dynamics and expectations. 'I glossed over it because I wanted to leave it behind me.' He felt his choices were curtailed by his lack of education.

Education gave people the freedom to move, to make choices. I didn't have that. I was stuck.

Peter was very aware of his lack of education and the cultural and social capital that went with it. His sense of being stigmatised was made worse by thinking that he had had opportunities but did not or could not avail of them. When the business closed Peter felt he lost out in the division of assets 'because I was seen as not deserving of it'. He felt his status as a single man further added to his marginalisation. He went to work part-time with a relative in his business as a temporary measure where again his illiteracy limited his role.

Although this subject matter was painful and difficult, Peter was willing to talk about it. He had acquired some perspective on his situation and mentioned some insights he

had acquired, through his study of psychology and sociology. He was aware that his account was partial and that his siblings could have a different account of what went on in the family. He recognised too, that despite the unsatisfactory aspects of his circumstances, when the business closed, 'it was like a bereavement.' His status, relationships and daily contacts had gone. As with many other mature students the trauma of being made redundant started a process of evaluation and was a catalyst for further change.

Peter was diagnosed with dyslexia when he was twenty-five but continued to disguise the condition. 'I was always ashamed I couldn't read or write...you tend to disguise it.' Five year ago he heard a radio programme on the subject of dyslexia and applied for a course they mentioned. Altogether he spent fourteen months on the course which was run by FAS and the dyslexia association and made considerable progress. Learning to type and to use the spell check was a breakthrough, 'it opened a new world to me.' He was encouraged to continue in education and as ideally he would like to be either a nurse or a teacher, applied for a one year course in the college. He was not fully aware of the nature of the course and was shocked when he realised how heavy the workload was. This caused him considerable anxiety. However, he did extremely well and finished with the highest grades in all subjects and progressed to the two year diploma course in Social Studies where he did equally well. He was pleasantly surprised and delighted with his success and considers it a major achievement. 'I got all distinctions. That never happened for me in school. It was brilliant.' He found it a major challenge coming to FE from the more relaxed course he had done 'learning how to spell and write and the computer.' He frequently thought of giving up. However he persevered with the help of what he considered great support.

The classes were small, which was great, you know the lecturers on an individual basis, they know your name and you can talk to them...there was an understanding of the problem.

Although he did not focus on it to a great extent, it was clear that being a mature student in a college with mainly younger people, was an extra source of stress. Although he made an effort to engage with everyone, he recognised, as most mature

students do, that an eighteen year-old naturally has different interests to someone of his age.

when you get older your interests gravitate to something else. Then you stick out like a sore thumb in a classroom and you're aware of that too, well aware of that.

Peter also had a scribe who was also a reader but found learning difficult.

My modus of learning is very primitive, it's basically regurgitating stuff, I know I understand it, but it's memory, There was a period of time I was just going to throw it in, but I stuck with it and came away with all distinctions which was very very good.

Despite this success and support he finds college something of an ordeal.

'I found it a struggle...I wish I didn't have to do it...I always go to college in a state of anxiety. College always fills me with fear, even at my stage now and the idea of, you know, what happens to you when you're young has a bearing on...you know, Freud... into your latter years...you can see it with dyslexia anyway...'

He is strongly motivated by a desire to work in nursing, teaching or the care area and sees college as a necessary step on the way.

I wish I didn't have to do it...I wish it was the other way round that I had the qualification. To get work I need a qualification, badly, so that's what I'm aiming for. That piece of paper is important.

His preferred career is in disability nursing or, if that does not work out, social care working with people with disabilities. He altruistically wants 'to give' and works part-time, voluntarily caring for people with severe disabilities.

I'd have to be a social care worker with people with disability, I would gravitate towards them, I can empathise more...I suppose that could be a subliminal reason...From my point of view, it would be a waste if I couldn't help someone with my skills and if I get paid for it well and good. That's the idea of it. I know it's a bit late in life, but anyway...

The idea of *having to use* his education came up a number of times in our discussion. I asked what if none of these career plans worked out.

It would have been a waste. And if I couldn't get a job and I couldn't give, I'd feel it was an enormous waste of effort and time and ...I would, I have to say that.

He recognises that at his age he may well not succeed in getting paid work in the area. Despite these comments and feeling that his time in college was a daily struggle, he clearly has quite mixed feelings, which he acknowledged in subsequent conversations, about his more recent encounter with the education system. He considers it 'time well spent' and in common with many other mature students feels 'it's a pity it didn't happen earlier.' When I asked him what he got from his time in FE, Peter said

'I enjoyed my time here, that's what I have to tell you, I got a lot of self-esteem.' My self esteem has gone up enormously and my idea of who I am as a person ... I value myself more, do you understand?

The struggle clearly had rewards and self-esteem was a huge benefit to Peter. He also felt he gained recognition from others too and saw his qualifications as a manifestation of this.

By doing the course here, by getting the pre-nursing, by getting the social studies diploma it meant that I had a qualification, to work, with high value, you know what I mean? And that's what this college gave me, if that explains it to you.

I'm able to put expression to the knowledge I've acquired which is important to me. I can talk to you about what I've learned and I can discourse with other people on psychology or sociology in a way I couldn't before.

In discussing the difficulties and anxiety involved in doing the course, Peter later said,

What's for you if you don't do it? You're probably sitting at home drawing the dole. That's worse. The alternative is worse, I'd have remained the same.

Peter here recognizes with Marcia (2010, p.27) that 'Change or die is true both evolutionarily and psychologically.'

Peter always liked learning but felt cut off from education by not being able to read and write. All his information was acquired aurally. Learning to use the computer, an important milestone for many mature students, was a major breakthrough: 'I could type...it was marvelous...the words and the spelling correction... Without that, I think education would have been closed to me.' He frequently said he like the subject matter and learning and used the telling phrase 'it's a pity I couldn't enjoy it.' indicating the cloud cast by the daily grind of working with dyslexia and the burden

he carried. He also, interestingly, remarked 'education can give you the tools to deal with the past.'

It was always in the back of my mind I failed somehow - I failed their expectations (his parents) and my expectations.

He acknowledges the element of chance in hearing the radio programme that started his journey.

I'm getting my education at the end of my time...that's how the opportunity presented itself, the programme on the radio, otherwise I wouldn't have gone back to education, I'd probably be stacking shelves in Dunne's Stores or something like that.

It was fortunate that Peter happened to hear the programme. However, this had to coincide with his receptiveness. Clearly he was at something of a crossroads, his part-time job was temporary and unsatisfactory and he was open to change. 'The time was right' for Peter. He saw his academic career so far as a considerable achievement.

Of all the achievements that I've done... all that I've done in life...this would top it.

During my two interviews and many casual conversations with Peter two other aspect of importance in his life emerged. The first was that during the decades of working unhappily in the business he had developed a hobby which gave him great satisfaction and a social outlet. He bred and trained dogs on a very small scale and had some significant successes. This 'passion' which he shared with his father and contributed to the bond between them, he saw, in retrospect, as developing the nurturing side of his personality as he had an 'affinity with animals' and allowing him a focus away from his other concerns.

He also said he now 'has someone' who supports and encourages him, *my rock* without whom he said he could not do what he's doing. He feels he is now getting the kind of support that was not available to him as a younger person.

As well as the personal side to the story, Peter was aware of himself as 'a middle class person' enmeshed with middle class values, expectations of education and status

which he recognises are 'part of the mix.' Class expectations appear, not always as explicitly, in the accounts of other mature students too.

For Peter, the three years he spent at a further education college was part of a progression that began with the course targeting his dyslexia. He has since progressed to a course in a third level institution now where he is now doing very well having been presented with a new set of challenges. Larger classes, lack of supports and accommodations and the remoteness of teaching staff 'you have to email them, for someone like me, that's a barrier' contrast with his time in FE.

The support he received along the way has been critical for Peter and although his motivation for returning to education was stated as vocational, there was clearly much more to it than this.

Conclusion

Some common themes and tentative findings emerged from initial data collection and participants' stories which were explored in further interviews. The idea of the appropriate time to return to education arose. In many cases this was characterised by an opportunity presented by the end of one role, such as a job or being a full-time carer, and a need to retrain or to act on a long-held desire to make up for 'missing out.' The influence of class and gender expectations was also evident in these accounts, as were the challenges that the mature students had to deal with. There was agreement on the importance of support both within the college and from significant others. It was clear too, that 'confidence' was a major benefit of participation.

Johnston and Merrill (2004, p2) suggests, 'particular life stories demonstrate how learning experiences are shaped by the dialectics of agency and structure.' The profiles of the participants were presented to give a sense of the reality and authenticity of the experience of these students and to illustrate the complexity of their lives and identities and what education means to them. These stories are discussed more thematically in the next chapter.

CHAPTER 6

THEMES FROM THE RESEARCH

Introduction

In this chapter I will outline the themes that have emerged from the data as a result of the analysis and make up the three categories of Latent Self, Emerging Self and Revised Self. The methodology chapter outlined how these themes emerged.

I draw here on Baxter and Britton's research (1999) and their notion of the 'latent self,' a term they use to describe feelings of unfulfilled potential, feeling capable of more or feeling unhappy in their career expressed by the students in their study. I use it here as an umbrella term to describe the student on entering further education and the circumstances that lead to the decision to come. It includes a number of extrinsic and intrinsic factors that came together and impacted on the decision. It describes the feeling that many of the students expressed in the interviews that an aspect of their personality was neglected. This concept is explored further in the following chapter which takes a more theoretical perspective.

In keeping with a grounded theory analysis I have grouped the key initial codes into concepts. Groups of similar concepts have been organised into categories.

Data	Initial Coding	Focused coding. Themes	Concept
I left school at 14.	Leaving school early.	Missing out.	
I always wanted to go back.	Missed out	Right time. Needing a qualification	Latent self
My kids were older.	Children grown.		
I lost my job.	Became unemployed	Change /opportunity	
Needing a qualification	Needing a qualification		

Data	Initial Coding	Focused coding Themes Sub-categories	Concept
It's hectic I want to get a Distinction It's given me confidence	Difficulty managing everything Wanting good grades Confidence	Anxieties:	Emerging self

Figure 6.1 Initial and focused coding

Latent self

In this section I will deal with the three subcategories of this concept:

- The Right Time
- Always wanted to go back/Missing out
- Needed a qualification

These emerged from the coding and they come together to give a fuller understanding of what I mean by 'the latent self' as a key finding.

The right time

Many of the participants said things that are best brought together to make a theme indicating that returning to school happened at the right time. It was primarily a sense that one's time had come as a result of a coalition or confluence of events. There was a sense of an opportune moment or an opportunity in the life story, usually filled with work, rearing children and other activities, that now this is 'my time.'

Many said that becoming unemployed was the immediate trigger that made them consider returning to education. This was usually combined with a number of other factors that together resulted in the decision to enroll on a course. In a number of cases it was a chance encounter or recommendation that planted the idea of becoming a mature student. Others had previous experience of the college or knew of it from living in the area or knew someone who had attended. Only one of the participants

had decided on the course and the college by searching for what she wanted on the internet.

Many used the phrase 'it was the right time' to express this coming together of motive, means, opportunity and desire. There may be some common elements in what makes the time right, but it varies greatly from individual to individual and it is not a feature of exterior events alone. The 'right time' for Aoife was a culmination of simmering thinking over a long time and reassessment of what she wanted in her life personally and vocationally.

... it was something that I always wanted to do... I'd been thinking about it and thinking about it I decided yes, I will do it now...and this seemed like the right time. Aoife

Some (mostly women) considered it as an opportunity for some 'me time' having devoted many years to child rearing or paid work or both and saw coming back as something of an indulgence.

I was busy nurturing him over the years...it was time for myself. Maeve

My daughters were grown up and that so I had the time, you know. With the recession as well too, you know, there aren't that many jobs out there so it was a great chance for me to go back to college. It was just actually the right time for me to go back. Brenda

Always wanted to/Missing out

Many said that they had 'always wanted to go back' to education, most had liked school, or at least some aspects of it and felt that they had 'missed out' by having to leave school early. There was a feeling of 'unfinished business' a vague sense that things might have been different if they had continued or that they would have seen themselves differently if they had done the Leaving Certificate or followed a different career path. Clearly it was an important issue for the individuals who felt that an aspect of their identity had been neglected.

Circumstances at the time meant I didn't get the opportunity to finish. Maybe I could have gone on. Maybe I could have been working at something different.. (James)

Well, to be honest with you I think I always kind of felt like I'd missed out kind of by not going back to school, and only as I got older that I realised, you know? (Caroline)

This feeling of missing out, often combined with the opportunity presented by becoming unemployed and a recommendation or discussion with a contact made returning to education a real possibility rather than a vague aspiration. For others, children growing up and becoming independent provided the opportunity, some having postponed returning while their children were young. Some had always wanted to work in a different area (Caroline, Maeve, Abbey) and decided that the time was right for them to do it now. James felt he missed out but was happy enough in his job and it was only when he became unemployed, and it was suggested to him, that he came back.

Needing a qualification

The need for a qualification or 'something for the CV' was frequently mentioned. Some felt that although they had acquired knowledge and experience in previous work, they needed certification, 'the piece of paper' to seek another job or get employment in the area they wanted. A career change was seen as offering greater levels of satisfaction and some had a specific career and a specific course in mind sometimes involving progression to HE before they come to the college, others had a more general desire for a qualification that would make them more marketable. Some wanted to 'try it out' or thought 'it could be interesting'. Further education is often recommended by Obair and Social Welfare and the VTOS and BTEA schemes ensure that there is no loss of benefits and other incidental expenses are met. Whatever the stated reasons or incentives to come back, all of the interviewees wanted change of some kind which prompted them to take action.

Many of the participants said they came to the college because they needed a qualification to enable them to get another or different job. Or they realised that, having worked for many years, they now needed to upgrade or change their skills or to gain accreditation for the skills they had. This instrumental motivation is clear and reasonable and was often the first thing that the interviewees said to explain how they came to be in the college. Some had clear career paths in mind and were attending the course that would equip them for work in their chosen area or lead to specific further

courses. For most, however, it was not so straightforward. The decision to return had been less direct. They had become unemployed but going back to education had not been uppermost in their minds and they were acting on a suggestion, idea or a recommendation that had been smoldering within them for some time. They had a general and somewhat vague aim of getting a qualification or being seen as someone who was keen to up-skill and go to college rather than do nothing. They hoped that during the course of the year or two, their options would become clearer, which often turned out to be the case. Others had a stronger, more active desire to revisit education but with no specific course or career in mind.

In some cases the qualification was needed for more than vocational reasons. It represented something important to the person and gave a degree of recognition they felt they lacked. The vocational motivation came easily to mind when they were asked, but clearly there were many other motivations associated with it for the individual. This is discussed more fully in the next chapter.

Although many mature students hoped to be able to get a better or different job after college they were adamant that if this were not to happen they would still consider their time well spent. Some, however, believed that if they could not use their new knowledge and skills that it would be 'a waste'. 'Using' it meant using it to get work or in work

You get a lot from it, you feel that you're stretching yourself a lot more, you feel you're gaining information and knowledge, but at the end of the day if you can't apply it in work it's been kind of wasted. (Peter)

Peter made this point several times during our conversation and acknowledged that his views were somewhat contradictory. Although he considers that he benefited enormously from returning to education, he says he found it an extremely difficult process and he saw it as a means to an end. He wanted to get a qualification that would allow him to work in the social care area or to progress to nursing. Unlike the others, he said he felt he could not enjoy the process because of his dyslexia and history: 'The pity about the whole thing is that I couldn't enjoy it'. Later in the same interview he was more positive:

And, em...I enjoyed my time here, that's what I have to tell you. I look on it as time well spent. It's a pity it didn't happen earlier. It's been a metamorphosis. (Peter)

All the participants were keen to find employment some though were pessimistic about finding work because of their age or the recession.

I know I'm at a stage that even if I get qualified it may not be much good to me really, I kind of know that. Jason

In saying 'it may not be much good to me' Jason is using an instrumentalist discourse although he is aware of the 'good' it has done him in other areas. He is keen to progress to HE if he can afford it.' Finance is the only problem. 'When I'm finished this course, if I can afford to go further, I definitely would'. It was not uncommon to have a number of different discourses going on within the same interview showing, perhaps, the complex, sometimes contradictory motivations within the one individual. A conversational style interview is not necessarily going to produce logical, consistent reasoning. The respondents are often working out their thinking during the course of the discussion and the interpretation is constructed during the course of the dialogue. It is not a case of 'earlier you said something totally different' but recognising that someone may hold contradictory positions simultaneously.

Four (Abbey, Caroline, Patrick, Jason) said they would work in a voluntary capacity if they could not find paid work in their area and some (Jason, Patrick, James, Caroline) are already engaged in work in their communities. The focus was not entirely on traditional notions of work and their motivations more complex than 'I needed a qualification' on the questionnaire would indicate. This supports the finding in the report 'Where Next?' (McCoy et al., 2010, p.117) where mature graduates saw their degree primarily as a personal achievement.

Change

All of the students involved in this study had undergone change. Through becoming unemployed or their children growing up and becoming independent or through a decision to change career. Some had experienced other changes as well such as illness or bereavement. They had experienced loss; loss of aspects of their previous identity and loss of the way of life and connections that went with it. Their roles and sense of

self had already altered and created a need or openness or desire to return to education where the process of change would continue.

Emerging self

By the concept of the 'emerging self' I mean the processes that contribute the changes experienced by the individual during their time as a student to bring about a revised sense of self. I discuss here a number of issues which arose from the data and can be seen as challenges, developmental tasks or crises which, with adequate support, are successfully negotiated leading to the changes described by the students.

There are a number of subcategories to this concept:

- Becoming a mature student
- Juggling commitments
- Fear of failure Am I up to it?
- First Assignment
- Support, enjoyment

Becoming a mature student

For mature students coming to FE is a major change and is usually accompanied by some feelings of apprehension and anxiety as well as more positive emotions. For some, their previous experience of education was not happy and their knowledge of FE and the structure of the education system generally is limited. Some passed by the college every day but had never been inside. There are also concerns about the subject matter and their ability to cope in what is a strange situation and new challenges. Many worry 'is it for me?' and that they will not be 'able for it' and fear showing themselves up. It is an area where some consider themselves somewhat alien as they see it as the preserve of young people who they see as more competent and familiar with the demands of education. Becoming a mature student is a non-normative transition which is an added stress and the support of peers and family may be lacking or muted. These can be major obstacles to overcome with considerable implications for the student's experience and the colleges.

At the beginning you do get a sense of overwhelming...will I be able to achieve this? Vera

When I saw the young people going up and down I said 'is this for me? am I doing the right thing?' Because I was going back to college, even though I felt it was a good idea....I just felt I didn't belong there. Daniel

Such feelings were not expressed by all and some did not experience any anxiety and were just keen to get in and start learning. All agreed, however, that what initial fears they had quickly vanished when they started their course. Anxiety about assignments and exams lasted longer for some, however.

Despite some degree of anxiety about coming to an FE college it did not take any of the participants long to settle in. All spoke of the supportive staff, pleasant atmosphere and how much they were enjoying the experience. Some spoke of the difficulties they were having with the workload or the subject matter but this was not presented as a major issue. Some questioned their abilities again when progressing to another course at a higher level within the college or elsewhere, but were generally quite confident at that stage. In considering teaching and learning, none of those interviewed thought mature students needed any special attention or different practices. Two suggested that extra tuition would be of help to those struggling in some subjects but stressed that this was something that would be of benefit to all, not just mature students.

Juggling

It is difficult to imagine any of the students I interviewed not juggling. This refers to the complex set of responsibilities and activities that go to make up the commitments and things that must be done as part of an adult life. They include taking care of a child or children, earning a living and negotiating relationships with partners and friends.

For many mature students, especially those with young children, the main difficulty lies in combining their role as a student with other roles as parent, carer or partner. This is not just a matter of having the supports in place for child minding and collecting from school, as there is a feeling of being constantly on call especially if a child is unwell or arrangements do not work out. For many, such concerns are 'running in the background' while in college and it can be hard to switch off and give all the attention to college matters. There are also, inevitably, times when college is

missed because other responsibilities take priority. It can mean 'rushing in and rushing out' with little time for conversation or cups of coffee with other students which are a huge source of support and satisfaction for mature students. For students with other demands, domestic or work, there was a necessity to use their time well and to try to make sure that children did not lose out. Some think that this can make them more focused and less tolerant of time wasters. They often have to do assignments after children went to bed and work late. Some felt guilty about not being as available to their children as they were before becoming a student. However, all the participants found their families very supportive and commented especially on the benefits for their children in terms of greater organisational ability and the knowledge and enthusiasm for learning they had passed on. All found the college and tutors understanding and accommodating.

Only one of my interview participants had young children and one member of the focus group. Most were either childless or had grown-up children. As mentioned above this for some was part of the time being right to come back. Thirty one of the one hundred who responded to the questionnaire had children. However, many students, not just those over 21, have children and these issues are very much part of FE.

Fear of failure

Most mature students are unsure of their ability to perform adequately in college. Many have not been in education for many years and are not familiar with requirements or standards and worry that they may not 'be up to it.' Academic failure could mean that the whole idea of coming back to education was a mistake, 'a waste of time' with consequences for the student's confidence. This is part of the risk involved in coming to college.

The first assignment

The first assignment is a major landmark for most mature students. An assignment is graded and carries with it all the connotations of being assessed and judged in what for most is a strange or new area demanding different skills that many feel they are lacking. It is generally a written piece produced on a computer, which can in itself be a big challenge, and many feel that they are exposing what they consider their

weaknesses in this area. Some spoke about knowing the material but being worried about expressing it properly. The 'language' of assignments and academic work generally took some getting used to. Students on the Return to Learning course have a more gentle introduction to assignments although the first one is still seen as very significant. Students on other courses sometimes feel dropped in the deep end where aspects of language, presentation and even formatting are taken for granted. The first assignment can be traumatic and can be important in determining whether a student, of any age, will continue. This is so even in a climate of encouragement and support.

It can also be, and generally is, hugely important in boosting the confidence of mature students who are generally very pleasantly surprised at how well they do.

In the beginning I was so unsure of myself, constantly asking myself 'is this going to work?' ...and when I'd be getting them back I was getting all distinctions, I think I was kinda shocked to be honest with you. Caroline

When I did so well so well in that (first assignment) it made me excel at the rest of them. I felt confident, more confident ... When I was doing it I was so unsure, am I doing it right? Maeve

The exam terrified me. To sit in a room with just a single desk in front of you I thought was unbelievable ...Jesus, I was terrified. I'd never done an exam before. Jason

Success in assignment and exams validates their academic abilities and is seen as an acceptance into the student role and student identity and bestows a feeling of entitlement to be in college. It also develops and affects other aspects of students' lives.

Keen to succeed

As well as being conscientious, many mature students were surprised at how they became ambitious to do well in college work, especially having done well in the first assignment.

I was getting merits on several subjects...merits, this is alright! I haven't got a distinction yet, I'd love a distinction. Daniel

I don't want to just pass. I didn't think I had that in me, I thought it would be just 'right we'll get through this'. From the first assignment on I wanted to

do it as expertly as I can. The feedback I'm getting has been fantastic, very encouraging and it just encourages me to do it more than ever. Vera

This ambition to achieve, and the focus and dedication of mature students, contributes to the discourse of 'the ideal student' found in some of the literature. Most teachers would agree that mature students are usually more willing to contribute in class. They are conscious of this and would often apologise saying they don't want to take over, but are aware of the silence of their classmates. Some, especially those with young children, have less time and are conscious of being focused and having to use their time well and are less tolerant of time wasters. Those with progression or a definite career in mind are motivated by achieving this goal. Many spoke of the importance of finishing and getting through and were determined to complete.

Computer competence

Some mentioned the importance of developing computer competence and, apart from those like Peter, for whom computer competence had a particular significance, this was seen as allowing participation, for writing up assignments and researching and represented competence in an area where younger people are generally seen to have an advantage and familiarity from having grown up with them. The advantages of computers were enthusiastically embraced once initial difficulties were overcome.

I felt more confident when I got the hang of using the computer...and when I got the hang of doing spell checks and when my second assignments came up and they were fonted right and justified right, I seen the difference and as I say...even the achievement... it was a success...I felt I didn't do too badly. Jason

I'm achieving a lot more because I'm simply pressing buttons. The computer, they (younger people) would have used a computer, known about a computer, known what it could do. I'm only learning the benefits of that now. Vera

Digital literacy is now seen as an essential requirement for social as well as economic participation and access to a wide range of everyday services and information. Those with low levels of education as less likely to have computer skills and educational disadvantage is exacerbated (Casey, 2009). It is unsurprising then, that for students who lacked digital literacy, acquiring these skills had important symbolic as well as practical importance.

Benefits of being older

Despite regrets expressed about not completing their education earlier, all the participants that they benefited more or differently from it now that they were older.

Some students are coming out of necessity or need, I'm coming because I like it, that's the other thing. Daniel

I think mature students...there's something in the back of their heads that asked them to come back and they're eager to learn something that they never learned. They're coming out of...maybe not them all, but I know from myself, they're coming out of the love of it more so than anything else, you know? I don't go home depressed from it every day, you know? Jason

All reported thinking that being older and having more experience of the world and in some cases specific areas, business or social care, gives an advantage by having a context to frame new information and another layer of understanding. All thought too that mature students are more focused and dedicated than their younger classmates.

'I couldn't get over how lax some of the younger students were.' Jason

Some found them ill-mannered and loud but none echoed the strong antipathy to young students expressed in the focus group. In general the lack of dedication displayed by younger students and their behaviour was viewed tolerantly as being 'their age.'

I just see them for what they are and that's immature, they just haven't grown up like. Caroline

Some were aware of being seen as more conscientious and anxious to meet deadlines than their younger classmates. This was true even in the Return to Learning class dedicated to mature students where the younger 'mature students' were seen as different, in this regard, to the older mature students. Older students and younger socialised together within the college but, except for special events, none of the mature students socialised with younger colleagues outside of college hours. They just thought it was natural that they would prefer the company of people of their own age and none, even the younger mature students, were fond of the traditional students' nights out involving drinking and late nights.

Support

Another key factor in the students' experience mentioned by all the participants was the support they received in the college from course co-ordinators and teachers. They found the relationship very positive and based on mutual respect. They commented on the accessibility, friendliness and encouragement of staff and many said they were aware that this may not be the case if they progressed to Higher Education. It was important to them that that they were recognised by staff and others in the college and known by name. They felt part of the college and valued. The flexibility and understanding afforded them with deadlines and assignments when they were having difficulties in college or home was greatly valued and it many cases is vital in their decision to stay.

Well, I feel there's great support here, to be honest with you... Vera Enjoying it

Many mentioned the atmosphere of the college and that they enjoyed coming in. The social side of college life was very important. This was mentioned especially by the men, but important to all.

The social side, meeting all the people from the other classes...you'd meet them and they'd stop and have a chat. James:

The idea of college being treated as work came up with some. It told of a certain business like approach and the idea that it was an occupation that offered some of the other aspects of work such as social interaction and sense of purpose. The enjoyment of learning, finding out new things and relating new knowledge to experience was a feature.

I'm not working now so this is something I'm concentrating on, this is my work, it's like a nine-to-five-job. Daniel

It's my job now. Jason

I love each and every subject as much as I can. I'm enjoying each and every subject and getting something out of it. Vera

Although some mature students have a definite career or progression route in mind when they came to FE many do not. Even with those who do, it is not unusual for their plans to change as they become aware of other possibilities or discover other interests.

Many, especially those who in to the Return to Learning class or one year FETAC courses, wanted to see how it would go and had no plans to progress at that stage and were taking it 'one step at a time.' James is in his fourth year in the college now and is considering going to an IT next year where he can get into the second year of a degree course. His plans evolved as his success in the Return to Learning course encouraged him to go on to the FETAC and from there he went on to a BTEC two year business course. James is typical of many students who surprise themselves by progressing much further in education than they would have thought possible when they came in. Further education thus offers an important alternative route to higher education for mature students and others who lack traditional Leaving Certificate admission requirements.

I didn't know what I wanted to do, so that course, RTL (Return to Learning) was there so I did that. Aoife

No, that didn't enter my head (progressing to an IT). James

My participants included students who had been in the college for only four months and some were in their second, third or fourth year. All of interviewees spoke about how much they enjoyed being on the course. Even a few months into the course Aoife, Vera and Maeve were enthusiastic.

It's good, I'm really enjoying it. I don't want to leave. Aoife:

So far so good. I am enjoying it. I'm enjoying what I'm learning, definitely. Vera

I love every aspect of it. It's a great college, the tutors are so helpful. I'm just soaking it up like a sponge. Maeve

Most admitted that they had not expected it to be so enjoyable and were surprised that they were enjoying it so much.

I suppose I'm surprised at the way it's gone so well. I'm enjoying it so much at times I forget I'm actually an older person. James

When pressed about what was enjoyable, it seemed to be 'everything' the academic side and the social. It was clear too that there was a sense of achievement and pride

I think just having ...maybe....you're doing something for yourself, you have a goal, you're getting up in the morning, and I really enjoy doing all the assignments. Aoife

I'd say I got great benefits out of it, I can't really put my finger on it, other than, I'm not bored. I'm happy to get up in the morning, happy to come in, happy to go home, happy to come back in the next day Jason

Of all the achievements that I've done... all that I've done in life...this would top it. Peter

Revised self – finding a voice

The 'revised self' is a core concept that emerged from how the participants described the change they had experienced as a result of their time in the college.

When asked to summarise what they had got from coming back to education in further education by far the most common response was 'confidence.' Confidence came up first when discussing academic work and assignments but it was soon clear that it extended beyond college work. It was apparent quite early that confidence was going to be a significant theme and in subsequent interviews I asked the participants what confidence meant to them and how it manifested itself in their lives. As so many responses explained their confidence in terms of speaking, saying things and having things to say, I developed the core category 'Finding a Voice' as a 'conceptual label' (Strauss & Corbin, 1994, p.121) to describe this revised self.

Confidence, a hundred percent. Not afraid to speak up or ask questions. Jason

I can voice a viewpoint that I wouldn't have voiced before. Vera

Some mentioned have more knowledge in a particular area, for the most part, there was a feeling of general competence and seeing themselves as people who not only have knowledge but the ability to acquire it as well.

I'd have no problem tackling something, I'd have a go at it and not be afraid to make a mistake on it. It would give you confidence, at least now I know if I had to do something where I have to go and read up on it or whatever, I wouldn't be afraid to tackle it or open a

book to look at it. Where before I never dreamt of opening a book just to read up on something. Jason

Confidence means me being myself and doing what I want to do. I think I have the confidence to move on, even though there is the possibility that I might not be able to follow through on it. Caroline

The kind of confidence mentioned in these quotes again imply a greater sense of control and being able to impact on conditions not just be subject to them. They display a self-assurance to cope with making mistakes as opposed to being afraid to take a risk. It is reminiscent of a comment Peter made about education: 'Education gave people the freedom to move, to make choices.' For many, their participation in FE brought about a change in their self-concept, self-respect and their sense of identity.

I see myself differently...I see myself more confident as well. Daniel

Well, seeing things differently, definitely, em ... it has definitely changed my attitude, it's definitely been a big, big eye-opener. Caroline

My self esteem has gone up enormously and my idea of who I am as a person ...I value myself more, do you understand? Peter

Confidence seems to have come from the culmination of dealing successfully with a new situation, academic success, being recognised, supported and valued by tutors and peers and others leading to an improved self-concept and greater readiness to take risks without fearing the consequences of failure for a delicate self image.

All recognised that they had changed since starting their course and most say 'it's a pity it didn't happen earlier' but recognise that they are different now to earlier and that for many unemployment was the unwelcome start of the process of change and that without it, the opportunity may never have arisen. They were surprised at the change in themselves having come to the college expecting to acquire more or different academic knowledge or skills to enhance their CV or prepare them for employment. These things came as well, but the personal changes were more significant and were manifested in the willingness shown by the students to progress to another course or to HE which they never would have thought themselves capable of before, or to made other significant career changes or changes in their personal lives.

Conclusion

This chapter outlined how the students came to be in FE and the process of change experienced by students before coming to college and while there. It suggests that by far the most significant result for them was the sense of greater confidence, awareness and agency. This was not a straightforward or linear process and was the result of encountering a number of milestones or challenges. These concepts are discussed more theoretically in the next chapter.

Chapter 7

Discussion

Introduction

The previous chapters described the development of the categories of 'Latent Self' and 'Emerging Self' and the core concept of 'Finding a Voice.' These chapters stayed close to the voices of the participants. This chapter uses some perspectives from the literature to help illuminate these stories and develop a fuller understanding of the issues that emerged and place them in the context of existing knowledge.

The chapter is divided into three sections according to the themes that emerged in the previous chapter.

- The first section, deals with the concept of the 'latent self' (Baxter and Britton, 1999). This concept came from a number of categories describing the various circumstances and the decision making involved in the students' return to education. It also looks some perspectives that offer ways of understanding motivations for participating in FE.
- 2. The second section discusses the challenges facing mature students and their role in the 'emerging self.'
- 3. The final section deals with revised self and the concept of 'finding a voice'

Section 1

The Latent Self

The previous chapter discussed how the concept of 'latent self' emerged to describe the students, and the circumstances that lead them to enroll in a full-time course in the college. The students I interviewed seemed initially to fall into roughly into three categories.

- 1. The first group consisted of those who wanted to move from unfulfilling jobs. They knew what they wanted to do and sought out a particular course that would lead to realising this ambition. The time was right for them to make the change even though it sometimes involved financial sacrifice, working part time, availing of savings or the support of a partner. This group tended to be confident of their abilities and had recent positive experience of education and did not share many of the anxieties about returning to college or assignment expressed by the others.
- 2. The second group was less definite about their ambitions. Education had not been prominent in their minds. Most were hoping to enhance their employment prospects by up-skilling and were acting on a suggestion or recommendation by 'trying out' education. This second group came back to college through 'a happy accident' the components of which were becoming unemployed, a desire to do something (develop skills, social interaction) a suggestion that they should consider returning to education, and being aware of the specific college or knowing someone who was a student in the college or who recommended it.
- 3. The third group had a long held and definite desire to return to education but with no clear career path in mind. They wanted to try it and see where it would lead. Common to all three groups was an affinity to education, sometimes strongly expressed, or at least not an antipathy to it. The interviews revealed however that their motivations to return were more complex than the first explanation and that education had multiple meanings for the participants.

Missing out

A feeling of 'missing out' was a common feature of the 'Latent self' (Baxter and Britton, 1999). The affinity for education and a belief in the value of education seem to have developed quite early. All spoke of their parents' encouragement to go to school and being encouraged to do well even though circumstances and culture often meant that employment was a priority and often a preference. 'I was always quite good at school. My father was devastated when I left' (Caroline). Many felt that they had not been 'pushed' enough in school and that staying on in school or going on to

higher education was not seen as an option for people from their background. The report 'Hidden Disadvantage?' (McCoy et al.2010) reviewed earlier, pointed out that retention in secondary school and progression to HE are highly patterned by socioeconomic background and teachers are often seen as having low expectations for disadvantaged groups. Most of the participants liked learning, although they may have disliked school, and felt they had some academic potential which was undeveloped and contributed to the feeling of 'missing out' which was mentioned by most of my participants, men and women, and indicated that they believed that things could have been different for them if they had remained in education and that they had ability that had remained untapped. Many said they 'always wanted to go back' indicating a long held feeling that they felt somewhat incomplete. James spoke of wanting to 'finish, complete' suggesting an 'incompleteness.' Work or a focus on family took over and the need to concentrate on these issues pushed education or a different career into the background. 'I always wanted to do it but genuinely never had the chance because of the job I was in.' (Vera). Peter said he had 'always wanted to be a nurse,' Maeve had 'always wanted' to do art and Caroline and Abbey 'always' wanted to work in the care area. Important aspects of the self were being denied or not fully used and the opportunity for reassessment and change came through unemployment or the time being right. This is supported by Baxter and Britton (1999, p.185) who refer to the 'latent or submerged self' wanting to be reclaimed and suggest that here education is not seen instrumentally, but as a means of 'realizing the self or becoming the self one always was.' Mercer (2007, p26) speaks of 'reclaiming a part of the self that had not been developed earlier in the lifespan.' Similarly, Dawson and Boulton (2000, p.168) speak of such motivations as 'imbalances' which they describe as 'situational' or 'dispositional.' This includes unhappiness in a job or dissatisfaction with their image of themselves and a desire for another 'possible self.' This could include the self as the artist instead of the salesperson or seeing themselves as academically competent and recognised as such. The desire to return to education they see as a desire to restore balance. They conclude that if people are happy with their lives and work, they have no motivation to return to education.

The importance of the affinity for education is supported by Courtney (1992) who cites Love's early study (1953) which sees participation in education not as a single decision or action but a series of actions, each stage of which is accompanied by

reflection. It suggests that before deciding to enroll in education an adult must have an awareness of education as a positive value in the solution of problems and must equate education with success and happiness. To participate a person must have (a) a current problem for which he or she seeks a solution and that problem must be acute and well defined; (b) be aware of a specific course or field of study; (c) be directed towards specific schools or programmes; (d) actually enroll. He thus connects the underlying orientation to learning and actually coming to the college to sign on for a course. In this way the personal psychological factors which produce an inclination to learning interact with the social factors that convert the inclination into action. The personal factors could be the 'transitions' mentioned above such as the divorce, the death of a loved one, or more commonly in this study, becoming unemployed. It also draws attention the some of the barriers mentioned later and the need to consider how the provision of information on courses, the availability of supports such as child care and factors such as personal contacts and friends who participated, the proximity or colleges and local knowledge impact on the decision to take the step of enrolling.

However, the first condition in Courtney's study above – 'that the problem must be acute and well defined' would not apply to all my participants. Some had only a vague notion of 'the problem' and in some cases came to education because 'it might be interesting' or because they had nothing else going on. They would not all meet the second condition either, as they were not aware 'of a specific course or field of study'. Some certainly wanted a specific course, others just wanted to come back to education and were guided in their choice by the college. They were trying it out and wanted to see what they liked.

An important insight from Courtney's work of particular relevance to this study, is that for adults to return to education, there must be an underlying orientation to learning which portrays learning in a positive light and predisposes the individual to participate. Education is a positive force and equated with happiness. While my participants certainly had this affinity for education, this relationship to education was not as straightforward or necessarily obvious to the individual as Courtney's account might suggest.

Houle's Typology

Another perspective on these students is offered by the theory of motivational orientations which came largely from Houle's (1961) classic typology of learners. This was based on in-depth interviews with 22 learners and was elaborated by others (Boshier, 1971, 1976; Morstain and Smart, 1974). The three classifications of 'goaloriented, 'activity-oriented' and 'learning-oriented' try to account for the origin of the need to learn by looking at the personality traits. Goal orientated learners use education as a means to an end and want specific things, the activity oriented who take part for the sake of the activity itself and the social interaction and learning oriented are seeking knowledge for its own sake. Houle's approach is highly generalized and assumes that learners can be easily categorised and that their reason for participating does not change. It also ignores the impact of the adults' circumstances and how these impact on decision making. It assumes too, that the respondents are able to identify and articulate their reasons. Morstain and Smart (1974) extended Houle's three categories to six, adding External Expectations (complying with authority), Social Welfare, (altruistic orientation) and Escape/Stimulation (alleviating boredom ore escaping home or work routine). They also allowed for learners to have multiple reasons for participation. Of my participants two, Maeve and Abbey, would fit the category of 'goal oriented' as they sought out a specific course to meet their needs. Another, Daniel, fits the 'activity oriented' category as the social aspect of college was very important to him. However, these and all the others, have elements that would qualify them for many other categories as well as having other more complex motivations and circumstances.

Cross (1981) acknowledged the usefulness of Houle's three-way typology and maintained that most adults were 'goal oriented'. 'Learning that will improve one's position in life is a major motivation. Just what will 'improve life' varies with age, sex, occupation and life stage in rather predictable ways' (p.6). She also acknowledges, however, that question about why adults participate in education are not going to be answered by applying any simple formula. Addressing the imbalances referred to earlier could be seen as improving one's life although such motivations might not be easily uncovered by surveying students.

Blair et al. (1995) suggest that conceptualisations of motivation based on quantitative studies of reasons, motivational typologies, life transition models and studies focusing on the institutional and social context offer incomplete explanations of participation in education. They point out that the predictive power of these approaches is weak and that adults in similar circumstances often make very different choices. Their 'new conceptualization' attempts to integrate motivational typologies, life transition models and ideas on barriers to participation. Their model considers the interaction of goals and conditions which determines participation. Adults must have goals that they feel can be realized through returning to education and the conditions (their circumstances) must be such that a return to education is possible. Goals may be vocational or personal, proactive or reactive. Conditions include the availability of information and suitable institutions, child care, previous experience, family situation and a supportive culture. Participation, they maintain, results when education is both desirable and possible.

They do recognize, however, that participation in education is not always the result of logical decision making and that strength of an individual's goals may overcome unfavourable conditions and similarly, conditions and circumstances may draw people to education even if they seem to have no educational goals. Such conditions might be joining in with a group of colleagues who decided to do a course or acting on a friend's suggestion to do something to get out of the house. This fits with some of the people in my study.

Barriers to participation

Motivational typologies and some life cycle theories see adults' decisions to return to education in terms of their individual psychology and with little reference to the social and structural factors that can affect these decisions. In trying to understand why some adults do not return to education many writers discuss 'barriers to participation' (Johnstone and Rivera, 1965; Cross, 1981; McGivney, 1991). Blair et al. (1995) outline some studies which consider the institutional and social contexts which they consider as barriers to participation.

Firstly, there are situational barriers which arise from individual's circumstances and may include issues such as lack of money, family circumstances and work situation. There are times when attending a full-time course is not possible because of work or other commitments that would preclude full-time attendance. 'The right time' referred to earlier usually included being free of these other constraints. However, not everyone can devote themselves exclusively to education. Most mature students have other concerns and demands. Managing or 'juggling' these roles and the role of student can be a major challenge and is considered the second section of this chapter.

Secondly, they consider dispositional barriers which include negative attitudes to education or classroom learning and lack of self-confidence. As mentioned earlier, all of my interviewees had some affinity to education which they were keen to pursue in circumstances that were better than their first encounter when they were in school.

Thirdly, institutional barriers (Cross, 1981) which are the alienating features of the education system and include things like difficulty in accessing information, lack of facilities and the elitist image projected by some institutions. Finally, sociocultural barriers, the most important of which they identify as social disapproval where there is an environment where education is not considered important or useful. This was not an issue for my participants.

Other versions of barriers to participation have specifically included lack of confidence, lack of support systems, lack of interest and what Hall and Donaldson (cited in Merriam and Caffarella, 1999) term 'lack of voice.'

The way a woman feels about herself, her self-esteem and self confidence, and the way she can express herself are significant elements in her decision about whether to participate in adult education.

Merriam and Caffarella (1999) point out that while looking at barriers to participation can be useful, it focuses on the individual and so only tells part of the story. They point to other research that sees education as part of general social participation which, they say, has implications for marketing and recruitment. They cite Benn (1997) who suggests that adult education providers should advertise through social groups and organisations.

Other perspectives emphasise class and gender issues. Adult education and especially higher education is seen as a middle class activity where working class adult students bring different social and cultural capital to that of younger students and the institution Merrill (2007). Students who were successful in school are more likely to participate in education as adults and success is largely determined by class, age, gender and educational background. From this perspective, improving participation is dependant on more fundamental social change (Merriman and Caffarella, 1999, p. 60).

Many FE colleges are well established and familiar features of local areas and do not carry connotations of elitism that can be associated with universities and other higher education institutions. The RANLHE report (2010) discussed earlier, referred to nontraditional students being like 'fish out of water' in HE. FE colleges are local and the level of the courses is seen as less formidable for those who have been out of education for some time. Liaising with the local community is considered important by FE colleges and special efforts are often made to accommodate people from the local area. For most of my participants a personal contact or previous experience of the college was the reason they came. They were often dealing with a person they knew or had been put in touch with and not a 'faceless institution'. They were all ready to come, even though some were very apprehensive and lacking confidence in their academic abilities. For many, the initial contact and interview are crucial in reassuring them and advising them on which course to take. However, there are those for whom returning to education, even in FE is a very forbidding prospect. Previous associations with education, lack of skills, whether actual or perceived, or lack of confidence prevents them making the initial contact. These potential students need to be addressed differently.

Lifestage Theories

Accounts of adult development have attained a certain status among adult education practitioners and in the broader society (Tennant, 1988). The terms 'midlife crisis' and 'identity crisis' are often used in popular parlance and other stereotypical views of adulthood have taken hold as a look at birthday cards for 30, 40 and fifty year olds and so on will testify. Such outmoded views still carry a certain currency and general acceptance even in an age of rapidly transforming ageing patterns and social change.

Traditionally developmental psychology focused on childhood and the importance of these formative years. More recently adulthood has been considered as a time of development also with various 'stages' of fairly predictable challenges from adolescence to old age (Coon & Mitterer, 2007, p. 138). Central to these changes is the idea that the sense of self can and does change, that identity is not fixed but dynamic.

There is no universal agreement among lifestage or adult development theorists on the number of stages or whether they are determined by age or developmental tasks. They assume that adulthood evolves through a series of phases of crises and resolution or stability and change. Each stage of the life cycle has specific tasks. The developmental changes in the life of the individual create needs, especially during periods of transition. These needs are translated into a need to learn new knowledge and skills to adjust to these changes.

Aslanian and Brickell (1980, p.34) drawing on lifespan theorists, maintain that during periods of transition adults can experience disequilibrium requiring new learning, 'new knowledge, new skills, and/or new attitudes or values'. Some events are identifiable and trigger a desire to learn at that time. Events such as becoming unemployed, changing job or getting divorced are such triggers. In their quantitative national study of American adults who retuned to education, Aslanian and Brickell found that eighty-three percent of their sample of 2,000 said their learning was associated with a change in their lives. Fifty-six percent of these changes were related to career, and only seventeen percent said they were motivated by learning for its own sake. Cross (1981. p. 95) is somewhat critical of this study and says it 'demonstrates one of the problems of field research designed to test hypotheses: that investigators are likely to find what they are looking for'. The terms 'turning point' (Crossan et al., 2003), 'fateful moment' (Giddens, 1991) and 'epiphany' (Denzin, 1989; Merrill, 2007) are useful in revealing the importance of critical incidents but carry connotations of a sudden singular event, whereas deciding to return to education is more likely to be a much longer process involving the coming together of many internal and external factors. This coming together of factors is seen in my participants' stories and captured by the notion of 'the right time.'

Adult development theories suggest that adults go through periods of stability and periods of transition. Each period or stage has specific tasks (Havighurst, 1948) that distinguish it and affect subsequent development. Most recognise the influence of external cultural and social forces and how they interact with the individual. In any society people may experiences a number of significant life changes at roughly the same age, but development is not necessarily chronological or age related. There will also be considerable differences in how changes are experienced and the individual responses to them. The social and cultural influences also vary greatly and change with time. It has been suggested (Neugarten and Neugarten, 1987; Sheehy, 1996; Arnett, 2004; Sugarman, 2001) that patterns of work, education and family have significantly changed even within a generation altering the shape of the life cycle.

In the space of one short generation the whole shape of the life cycle had been fundamentally altered. People today are leaving childhood sooner, but are taking longer to grow up and much longer to grow old. (Sheehy, 1996)

Tennant (1991) cautions against subscribing to rigid views about the 'proper' course of development and acknowledges that people chart their life course in diverse ways which may not fit in with any pattern or theory. Perceptions of age have changed as people live longer and are enjoying better health for longer. Marriage may be postponed, co-habitation is common and re-marriage and second families are not unusual. People often change careers and participate in education after 'retirement'. To some extent these theories are a product of their time and appear somewhat normative. Individual experiences vary greatly and age is just one factor that influences development. Neugarten and Neugarten (1987) stress that personal and sociohistorical circumstances must be taken into account and age is becoming less relevant to significant events in adults' lives.

Tennant and Pogson (1995) also point out that the life course is socially constructed and is, therefore, a social as well as personal phenomenon. Different social and historical contexts impose their own forms of life structuring through available discourses and norms. Importantly too, socially prescribed age categories and behaviours become embedded in the psychology of individuals who deals with adapting to or struggling with these established norms. Mature students have the

added challenge of dealing with what is often seen, by them and others, as a non-normative transition. My interviewees were aware that they were coming to education at an unconventional time 'doing it the wrong way round' (James) and in what is seen as a young person's environment. Neugarten (1976) suggests that such non-normative events cause potential crises, not events that occur as expected or at the 'right time.'

Gilligan (1977) points out that development theories that stress the development of individual identity, independence and separateness, reflect a gender bias common in the literature. Males are taken as the norm and women are considered 'as either deviant or deficient in their development' (p.482). Women develop differently and have qualities that are undervalued in the literature (she cites especially Kohlberg's stage theory of moral development). Erikson has been criticized for his reliance on Freud and the idea of the biological distinctiveness of men and women. Gilligan found that women develop differently and define their development mostly in the context of connections rather than separation. Women do not deal with identity issues during adolescence in the same way as men (Josselson, 1987) and may postpone dealing with concerns of personal identity while prioritizing the achievement of intimacy (Chodorow, 1978). Lowenthal et al. (1975) found that women were more likely to make more changes in mid-life than men. This coincides with children being reared and the possibility of other roles and opportunities to address deferred identity issues. Many middle-aged female mature students in my study referred to returning to education as 'me time' or 'it's my turn now' implying that they had given themselves over to other concerns, childrearing and homemaking, until now. They spoke too of the cultural norms of the time and their class when they were leaving school where the emphasis was on getting a job rather than progressing in education 'especially with girls' (Vera). Many who felt they were good at school or who felt they had 'missed out' on education wanted a chance to fulfill their potential and maybe develop a different career or role. These participants' stories illustrate the influence of gender and class and how they can intersect with development issues to influence decisionmaking.

Lifespan theories include the idea of crisis or conflict as a necessary element in bringing about development. As we have already seen, the mature students in my study encountered change or a number of changes and experienced 'disequilibrium' (Aslanian and Brickell, 1980, p.34). This presented them with an opportunity or a necessity to consider other options and which resulted in them, sometimes indirectly, returning to education. However, for most, these changes and developments were not age related. The significant event of becoming unemployed was a trigger for most.

Bearing in mind their limitations, lifespan theories provide a way of seeing adult development and the position of the individual within it and how this might affect decisions to return to education. They also highlight the importance of early experiences for the construction of the framework within which we operate as adults, often unconsciously, and have implications for how we engage in the development tasks of adulthood.

The idea of a 'moratorium', suggested by Erikson, Marcia and Arnett has particular relevance for mature students.

Moratorium

Erikson's psychosocial account is heavily influenced by Freudian developmental thinking and suggests that the adult goes through a number of developmental stages, each characterised by particular concerns and requiring a balance between opposing elements. The adult is seen in the context of the social system with which he/she interacts. The psychosocial stages are produced by the interaction between the individual and the social context. In Erickson's theory eight stages of development unfold as we go through life. Each stage presents a unique developmental task or crisis that must be resolved. Development occurs through resolution of these crises. This crisis is 'not a catastrophe, but a turning point marked by both increased vulnerability and enhanced potential' (Santrock, 2009, p.23). After adolescence and the development of identity, the main task of early adulthood is becoming intimate with another person (intimacy v isolation) which usually happens, it is suggested, in the twenties. The next stage is dominated by the tension between generativity and self absorption (stagnation). This involves moving the focus from the self and becoming concerned with others, society and the world and one's legacy. The final stage, (integrity v despair) involves the task of looking back over ones' life and either accepting and being content with it, or despairing.

For Erikson development is age related and sequential and the outlined stages are cumulative. The outcome at each stage will influence the next so the capacity to form intimate relationships depends to a large extent on outcome of the previous stage, the identity crisis of adolescence. Although identity is generally formed and integrated in adolescence, it is renegotiated through adulthood as events such as divorce and bereavement force a readjustment. Erikson suggested that to assist in the process of identity formation, adolescents needed a psychological moratorium, time out from the world and their other activities, to explore different identities.

Marcia (2010, p.27) suggests that Erikson's stages are not straightforward and the suggestion that if a stage resolution is missed the individual's life will be flawed developmentally, is not necessarily the case. Marcia points out that every stage occurs at every other stage and allows for the remediation of uncompleted stages and 'the precocious resolution of normally unexpected ones.'

Marcia, drawing on Erikson, suggested that for adolescents to establish an identity there must be an exploration and a commitment to an identity, values and roles. If there has been no exploration or crisis and no commitments made this results in diffusion. Adopting the values of family or others without any exploration results in a state of foreclosure. Identity moratorium is actively exploring and questioning but not yet making any commitments. This is the stage of typical adolescent experimentation and uncertainty and is necessary for the resolution of this stage of development.

Fleming (2009) points out that in our society; some privileged young people get a state sponsored opportunity of a few years in third level education which offers the security and time out to address these issues. This moratorium offers an opportunity to grow and is denied to other less privileged people. Early school leavers and others who take up adult roles early do not have this period of development or the foundation needed for subsequent stages of development. Poverty and exclusion can therefore be seen to be damaging for development with consequences for the individual, their relationships and careers. Fleming cites Rubin (1976) who maintains that poverty is more likely to lead to a negative resolution of crises at each of Erikson's developmental stages. The moratorium, Rubin says, is 'a luxury of the affluent middle

class.' Most of my cohort would not fit into the category of 'the affluent middle class' and did not have the opportunity for a moratorium as adolescents or young adults. West (1996, p.9) suggests that 'Damage, even of a basic primitive kind can be repaired through good enough relationships in later life. Some people turn to therapy but reparation might also be possible in educational settings through contact with people who respect, care and empathise with us.' This is not to suggest that my participants did not achieve a sense of identity in adolescence or that they suffered damage of any kind. Identity needs to be worked on throughout life and most of these students had experienced major change and were in the process of discovering new roles professionally and personally.

Arnett (2004) also speaks of a moratorium. He observed that the road to adulthood has become longer in recent decades. Traditional markers of adulthood such as getting married, embarking on a career and having children are now occurring later. He uses the term 'emerging adulthood' to describe the period of roughly age eighteen to the late twenties which he sees as a distinct stage of life, not just a prolonged adolescence, characterised by identity exploration, instability, self-focus and a 'sense of possibilities.' Arnett does not see this as a universal phenomenon and points to cultural conditions which facilitated its emergence in recent times in Western countries. They include longer periods being spent in college, fewer entry level jobs, less need or desire to marry because of social acceptance of alternative arrangements, easy availability of contraception and having children later. These young people are sometimes referred to as 'Kippers' (Kids in Parents' pockets Eroding Retirement Savings), in Italy they are called 'Mammone' (won't give up on mother's cooking) and in Germany they are 'Nesthocker' (nest squatters) (Coon and Mitterer, 2007, p.135). Arnett admits that in the developing world and in less affluent parts of other countries people have to grow up fast and do not have this opportunity for development. He acknowledges too that this moratorium, like Marcia's, is something of a luxury as these 'emerging adults' often need financial support which is not available to the less privileged who often have to assume adult roles without the chance to experiment.

Some⁴ see 'emerging adulthood' simply as a by product of unemployment, high house prices and more time spent in education. It could also be seen as a manifestation of the changing timetable for adulthood and postmodern fragmentation, impermanence and the dilemmas posed by a bewildering range of choices.

This idea of a moratorium is relevant when considering adults in further education. Identity development is not just a task of adolescence and adults continually reconstruct their identities over time and in response to different circumstances, demands and 'normal expectable disequilibrating events associated with each of the succeeding adult life cycle stages' (Marcia, 2010, p.27). Many adults return to education having experienced a change in their lives, their roles and their sense of identity. Unemployment and grown children leaving home were cited by many of the adults in this study. Time in the college offered them many of the characteristics of a moratorium. West (1996) too suggests that education can offer space for adults to experiment with their identities. It is seen as an enjoyable, if sometimes fraught, opportunity for time apart, 'me time' which allows for questioning, learning, discussion with other adults and discovering or reclaiming aspects of the personality that had been neglected or submerged. This moratorium provides a liminal space for the latent self to develop and emerge. This is not a straightforward process but involves several stages or 'crises' the successful resolution of which mark progress to the 'revised self'. This takes place in a supportive environment which affords respect and safety.

Post-modern uncertainty

Importantly for our current post Celtic Tiger Ireland, West (1996, p.8) locates personal situations in the wider context of late modernity which 'constantly precipitates crises of, as well as opportunities for, self.' With the major changes in traditional certainties and biographical trajectories, 'individuals have to make more of their own choices and construct more of their own meanings and biographies' (West, 1996, pp.8-9). The lost of faith in social institutions, the financial crisis and seeming loss of control over our destiny have created a society characterised by anxiety with none of the traditional metanarratives to offer support (Beck, 1992). West suggests there are two possible responses: a fluid and generative creativity, or a pathological

_

⁴ The majority of bloggers responding to an article on Arnett's ideas in the New York Times.

defensiveness against change. The former implies a 'reflexive project of the self' (Giddens, 1991) sustaining a coherent yet continuously revised biographical narrative. For West, the strength of the self determines if one is secure enough to cope with the opportunities for experiment with identity in this fragmented and uncertain context.

Ahl on motivation

Cross (1981, p.106) comments on the difficulty of accurately estimating the importance of dispositional barriers to participation. It is more acceptable, she maintains, to say you are too busy to take part in education that to say that you are not interested. In the neo-liberal context discussed earlier (chapter 2) this could be seen as a manifestation of the hegemony of dominant discourses of lifelong learning emphasising the responsibility of the individual to up-skill and be flexible and able to accommodate the demands of globalisation and changing work practices. Lifelong learning is an individual imperative for citizenship and social inclusion (Harvey, 1989).

Ahl (2008) suggests that from a Foucauldian perspective, knowledge is any discourse that has been deemed 'truth' and can be used to support or justify actions and ignore or marginalise other versions. Truth is historically contingent and may be an unconscious process and embedded in the taken for granted and unquestioned assumptions that constitute our frame of reference. Motivation is a social construct and, Ahl maintains, is seen from a Western and male perspective.

Early theories on motivation (Mayo, 1933; Herzberg, 1966), Ahl suggests, are based in industrial psychology and prompted by an interest in what motivates workers. These theories see motivation as always latent, temporarily hampered by dispositional, situational or structural factors. Remove barriers and motivation will resurface. This assumes that an entity called 'motivation' exists, that motivation causes behaviour, that it is possible to affect motivation and that motivation resides with the individual. In this way the focus is on the individual who is seen as deficient and held responsible for social problems. The role of government and policy is made invisible. The solution is seen to lie in the individual, not in social change or government action. The unwilling learner is the reason for, and the solution to, social problems. Ahl sees motivation as a relational concept. Motivation is not seen as residing in the individual

but in the relationship between those who are not motivated and those who think they should be. People who value things other than education draw on other discourses that produce other subject identities than the discourse of lifelong learning. Sargant, (1990) cited in Jarvis (1995) raises the moral question of how much non-participants should be persuaded to learn.

Conclusion

Although a number of these studies are somewhat dated and very much of their time, they highlight some of the methodological and design difficulties in getting a complete picture of motivation (Cross, 1981, p.107). However, they provide useful ways of looking at the reasons adults give for their decisions to participate in education and offer a lens through which to view the individual stories of my participants. A limitation of many studies on motivation, especially quantitative studies, is that they do not take sufficient account of the subjective emotional factors and look at the individual and individual change without adequately considering how socio-cultural positioning interacts with individual biographies in determining motivation to return to education.

Adults generally have not one, but many reasons, often inter-related, for returning to education. The decision is influenced by a range of psychological, social, vocational and emotional aspects that may be masked in the constraints of replies to a survey. West (1996) looks at motivation in the context of the students' biographies and their socio-cultural circumstances. Using biographical methods, his study interviewed thirty students in the UK over a period of three years as they progressed from Access or Foundation courses to higher education institutions. Biographical approaches (West, 1996; Merrill. 2007) offer ways of exploring personal accounts, attitudes and feelings. West (1996, p.1) suggests that 'understanding of student motivation is limited precisely because learners themselves have rarely been asked or encouraged to reflect in unstructured and longitudinal ways on their reasons for learning in relation to personal histories.' Cross (1981, p.97) recognized the inadequacy of simplistic explanations for participation. 'Motives differ for different groups of learners, at different stages of life, and most individuals have not one but multiple reasons for learning.' Although instrumental reasons are often cited for returning to education, this does not always tally with the choice of course and does not take account of the

meaning this has for the individual. Available cultures and discourses influence how learning choices are explained. Vocational motivations are often the most immediately accessible and acceptable reply to a direct question about motivation for the respondent. Vocational motivations do not preclude other motivations. Even if people know why they act as they do they may be unable to explain it. Unpacking such motivations is time consuming and demands effort and commitment from participant and interviewer and can not be done within the confines of a survey response. McGivney (1991, p.17) points out that adults 'may find it simpler to express a simple, concrete and easily understood purpose for their involvement than to disentangle the mesh of inter-related and perhaps not fully recognised motives that have led them to this step.' Motives for coming to education and reasons to stay may be different. Even if the initial motivation was vocational, other interests, goals and motivations can develop as a result of participation these can include a sense of achievement, personal satisfaction, enjoyment confidence, an enhanced self concept, and can be a major factor in the decision to continue.

Section 2

The Emerging Self

This section looks at the challenges facing mature students and their role in the 'emerging self.'

Many of the students in my study experienced a change in their sense of who they were by becoming unemployed. Others wanted to address a felt lack in elements of their identities by returning to education or changing careers. Other aspects of their biographies, personal circumstances and the broader social climate contributed to their decision and their ability to act on it. It is suggested here that the experience of being a mature student involves a continuing process of change for the individual. This transformation is not immediate or straightforward. It is characterized by encountering and dealing with a number of developmental tasks. Successfully dealing with these crises marks stages in the development of the transformed identity described as 'the revised self' or 'finding a voice'. This section looks at some of these anxieties and their role in the process of identity transformation.

Becoming a mature student

Mercer and Saunders (2004) and West (1996) suggest that the process of change in mature students in higher education is not straightforward or linear and involves losses as well as gains and periods of conflict and imbalance.

A fragmentary, even fragmenting process in which the self becomes divided..... between private lives (as parent or partner) and public spheres (as student in higher education); and between experiential ways of knowing (personal, subjective, emotional and academic knowledge (objective and abstract) (West, 1996, p.ix).

The students in my study also spoke of difficulties they encountered on their journey to a revised sense of self. Only two of my participants, Maeve and Abbey, (both goal oriented) said that they did not feel any anxiety returning to college. Both had recent and positive experience of education and were keen to get on with what they saw as a means to the goal of working in their preferred area. Transitions to full-time education are major life changes and are stressful for most students of any age (Palmer et al., 2009). Transitional periods are, of their nature, stressful as they involve change and greater reflexivity. Stressful events, however, 'are not necessarily negative [they are] inevitable ... without them there is no growth' (Marcia, 2010, p.33). Mature students are embarking on a major change of role and identity. They are conscious of the nonnormative nature of their move and most see themselves as being like 'fish out of water' where their age and cultural and social capital set them apart. Most are not familiar with the system and presume that younger students are faster learners than they are and that they know how to study and negotiate the demands of college. These adults find themselves in unfamiliar territory occupying roles associated with 'youthful dependency' (Crossan et al., 2003). Illeris (2006) observed that adults often slip into the traditional role of pupil and surrender control to the teacher although, paradoxically, they have a hard time accepting the lack of responsibility that this role entails. The first few days in college can be a major hurdle.

I suddenly just stopped and said, 'what am I doing here?' I was ready to walk back out. Daniel

When I saw the young people going up and down I said 'is this for me? am I doing the right thing?' ... I just felt I didn't belong there.' Jason

Although most quickly overcome these initial difficulties, the first few days can be a 'make or break' ordeal with implications for induction programmes and retention. In core VETOS or Return to Learning courses more attention is paid to the specific needs of older students. With other courses, mature students are not afforded any special treatment and many feel 'thrown in the deep end'. Successfully negotiating this transition into further education and the student role is an identity shift and a major landmark on the road to a changed self concept.

Home culture and college culture

Reports of non-traditional students in higher education often refer to the difficulties caused by the disparity between the culture of the home and that of college (Reay et al., 2002; Mercer and Saunders, 2004; Tett, 2004; Brine and Waller, 2004). While my participants were aware of how their experience of education had been shaped by their culture and class backgrounds, they did not see this as a problem. All spoke of the support and encouragement they received from family, friends and significant others in their lives and they were proud of their status of student. While the nature of further education is often not completely understood, it is generally recognized as part of the VEC and looked on as something familiar, accessible and not separate from the local community. Doing a course in 'the college' is not uncommon in the area. University, however, has well established, if changing, connotations of greater elitism and cultural difference. Even so, for adults who have been out of school for many years and for whom education often carries mixed feelings, coming to the college is a major step. Tett (2004) suggests that 'There is considerable evidence that working-class students 'choose' HE institutions where they are more likely to be with people that are 'like me' in terms of class, 'race, gender and so on'. While this is not always true and some students from disadvantaged backgrounds do go to 'elite' colleges, this could be seen as a reason why these students choose FE rather than university. While it would appear reasonable to seek the security of the familiar in such circumstances, many other factors impinge on the choice. The location of the college, the courses on offer and the level were important factors for the students in the survey. Most did not see themselves as university material and some wanted to 'try it out' with a one year course. The perception of FE as accessible is one of sector's great strengths. While most did not see themselves progressing past FE when they enrolled, most now see themselves as having the educational and cultural capital to do so.

Baxter and Britton (2001) suggest that to be 'educated' is to stake a claim to a new identity which has implications for relationships with family and friends. The new identity can be threatening to one's own sense of self and to others. These risks they see as especially relating to gender roles in the family and risks accompanying the move away from working class habitus which they see as 'an inevitable consequence of being in higher education' (p.87). They see mature students in the context of postmodern fragmentation (Beck, 1992) as leaving behind the certainty of their old identities in an attempt to forge new ones. These new identities have to be negotiated in the classed and gendered lives of the individuals. Although they were conscious of the changes in themselves and aware of their class positioning, none of the students in my cohort spoke of any difficulties with partners, friends, colleagues or class identity. None spoke of conflicts of identities or of having to compartmentalise their lives or others not understanding what they were doing and why.

Juggling roles

Older students are likely to have more established roles, obligations and demands on their time and attention than younger students. Many have children, part-time work or elderly parents to consider. Lack of time is frequently mentioned, especially by those with young children, and study and assignments have to be organised at the times least disruptive to family life, often late at night. There is also the awareness that illness of a child or a breakdown in arrangements will result in having to take time off college. A number of periods of unavoidable absence and consequently falling behind are, in my experience, often cited by adults as the reason for dropping out or deferring (Where Next? 2010, p.115). Some spoke of 'not being able to switch off completely' and having to have their phone on, 'just in case.' Feelings of guilt for not being available to children are common. Also common, however, is the recognition that the children benefit from their mother's or father's participation, and the role model it provided. Mothers are commonly cited in the literature as feeling the demands of childcare differently to fathers (Reay et al., 2002; Edwards, 1993). Some expressed guilt for being in college with the long-term goal of a better career and enjoying it so much, when they could have had a 'basic job' (Vera). Managing multiple roles is another stressful challenge that is part of being a mature student. Adult relationships can be strained by such re-negotiation of roles and the change and development in the

individual brought about by education has implications for the student, partners and relationships.

While mature students with children complained of not having time for everything and their social life being curtailed, others without such responsibilities said they had no problems keeping up with friends. Some spoke of treating college 'like a nine-to-five job' (Daniel). Daniel often stayed late in college to do course work and this, for him, enabled the separation of college and home and minimised disruption to long established domestic routines. Most were not able to make such clear distinctions. Many men spoke of college giving a structure to their day and providing valuable social contact which they greatly missed since becoming unemployed. The social aspect of college was important to all the interviewees and most enjoyed mixing with younger students although some found them to be 'messers' and not serious students, they were tolerant and understanding of their perceived failings and immaturity. 'It's just their age.' The antipathy of my focus group to younger students, echoed by Keane (2009) was not borne out in the interviews. Although they got well and were friendly with everyone, mature students tend not to socialise with younger ones outside of college. This was partly lack of time but also not sharing the same idea of what constituted a good night out. Bean and Metzner (1985) note that the campus environment and socialising is not as important for mature students as it is for younger ones. They have already established networks and relationships. My participants felt it was natural to form closer alliances with others with whom they shared the commonality of age and circumstances. This was true especially of the dedicated 'Return to Learning' class. Mature students see themselves as more focused, being there because they want to be, and being able to appreciate it better. This fits with Warmington (2002) who speaks of studenthood as a surrogate occupation among mature students. The group he studied, which was very similar to my students, constructed themselves into a group that had many of the characteristics of 'a selfsustaining closed occupational community' (p.584). A feeling of solidarity was engendered by the shared experience of being on a programme whose demands and trials were not understood by others. This had an instrumental value in providing useful support and helped create a shared work ethic in striving towards the ultimate goal of qualifying and eventually a more satisfying or rewarding occupation. It also had a discursive value in that it contributed to the construction of the celebratory

discourse of 'maturity, motivation and commitment' (Avis, 1997; Keane, 2009) that contributed to the identity of the group and provided support. This did not appear to contribute to 'othering' or an 'us and them' attitude among mature students or the younger ones.

Am I up to it?

Most mature students coming back to education wonder if they are 'up to it' whether they will be able to cope with the academic demands of the course. Many of these fears centre on the first graded assignment which is another crisis or anxiety and a major landmark on their journey. People who have been out of education for many years, often with unhappy associations, are worried about their ability to perform in this environment. Assessment is a risk and carries connotations of being judged with consequences for self-esteem and confidence in what is still a new environment. Knowles (1998, p.88)) said 'Nothing makes an adult feel more childlike than being judged by another adult'. Young (2000) suggests that response to feedback on assignments is related to the student's self esteem. Students with high levels of self esteem receive feedback of any kind positively, whereas those with low self esteem can see even positive comments as being negative and a judgment of their ability rather than suggestions for change. Assessment also contributes to students' self esteem and self confidence. By being recognised by the tutor and given official confirmation of their abilities it marks an important transition in their recognition of themselves as entitled to be in the college pursuing this activity and able to meet the demands put on them. It is part of the reconstruction of the self.

At the beginning you do get a sense of overwhelming...will I be able to achieve this? and that risk is there... Was it a wasted year? Before, up to now. I mean I know myself I can do this now. Vera

Similarly, Palmer et al. (2009) speak of the first feedback as a key turning point in first year students feeling full members of university life. These students are in a 'liminal' space not yet fully integrated in their new student life. A turning point is defined as 'an event or an experience...that stands out, and which triggers and results in the student developing (or not) a sense of belonging to university life'. Students speak of coming to grips with the language of assignments and of working out what is expected of them. This represents the acquisition of new forms of cultural and

educational capital which is carries with it a revised sense of themselves as capable learners and being able to operate in this environment. Similar anxieties arise when a student moves to a higher level course, although they report that the very first assignment is the most significant for them.

Risk

The concept of risk is useful in looking at mature students. Going to college for them is a non-normative transition which carries extra burdens by being out of step with their peers. Further education and its practices for most are unknown territory and there is the possibility that it will not work out or they may find themselves 'not up to it' – the risk of failure with consequences for self esteem and identity. There are also risks to relationships and family and not having time for other things in life. New learning and questioning can be threatening to established views and frames of reference and carry risks of disturbance. There is the risk that it may not bring the desired result in terms of employment, progression or otherwise.

Davies and Williams (2001) see the risks to mature students in higher education in five broad categories which encompass most of those above: Future rewards – what if no job results and if I accrue enormous debt? Personal achievement – what if I fail? Finance – what if I cannot afford to live and /or support my family? Time-what if I cannot maintain existing obligations or my family relationships suffer? Resources and services-what if the institution won't accept me or if there is insufficient support?

Davies and Williams (2001) suggest that mature students see the decision to go to university as an investment with risks attached. They wanted a 'better' job, but what this meant varied considerably. Some of my group came with the idea of working in a specific area (Caroline, Peter, Maeve, Abbey) some were less specific in their goals although they hoped to find interesting work (Joe, Vera, Aoife, Daniel, Jason). Even those most focused on a career when they started would not consider that their time was wasted if they did not find a job. They knew their age and the current economic downturn made finding employment difficult. However, they felt they were more employable and had experienced great personal benefits; their risk paid off even if not in ways they had anticipated. Most had started in FE on one-year courses. VTOS and BTEA meant there were no fees or loss of income for those on social welfare. Some

needed to find a better paying job and were anxious for a more satisfying career (Vera, Daniel) even if this meant more time in college. Five were willing to do voluntary work in the community if paid work was not available in their chosen area. This indicates that the focus was not entirely on individual financial improvement but on expressing aspects of the self.

Davies and Williams (2001) point out that in terms of time commitment, even part-time courses often mean that students complain 'I've had to put the rest of my life on hold'. This and other issues to do with time have greater impact on those with children. This is true of my cohort too. Those without young children generally found they could manage to balance college and other parts of their lives. This was much more difficult for those with children. Davies and Williams also point out that these issues inevitably are linked to the willingness of others, usually family, to bear at least some of the costs. Individual circumstances, reactions and resources of intellectual, social and emotional capital vary greatly. Students' biographies impact on decisions and responses to risk in ways that may not always be available for conscious consideration. As West and Holliday (2010, p.13) point out, "Present anxiety – 'are we good enough?' -May feed on deeply embodied past anxieties."

Attachment Theory

Sheehy (1996, p.513) observed that growth has potential for pain and 'growth demands a temporary loss of security'. The development of the new sense of self is brought about by encountering risk and successfully negotiating these crises. Bowlby's attachment theory offers a useful way of looking at adults' readiness to surrender security, undertake these challenges and engage with risky situations.

Like Erikson and other developmental theorists, Bowlby demonstrates the importance of early experience for later development. His theory of attachment (1979) proposes that the early bonds formed by children with their caregivers impact on relationships throughout life. Children who are confident of their caregiver's availability establish a sense of security and have a secure base from which to explore the world. Such security is necessary for 'normal' cognitive, social and emotional development. Attachment has a biological and evolutionary component also. Strong bonds between parents and child are necessary for the child's survival and lack of interaction and

stimulation prevents normal development of the brain. Securely attached children see themselves as loved and happy to explore and grow towards autonomy. The response of the mother or other caregiver leads to the establishment of patterns of attachment and internal working models which affect later relationships. An internal working model is a set of expectations about relationships based on early experiences with the primary care giver. The child's experience of attachment strongly influences subsequent reactions to stress, to relationships, self-esteem, sense of security and identity (Fleming, 2008b).

...the quality of the child's experience with his/her attachment figures plays a central role in personality and social development, and in the intergenerational transmission of caregiving quality (Byrne et al., 2005, p. 118).

The implications of attachment theory for adult learners are significant if we accept that early patterns of attachment can influence the adult student and impact on how they cope with new situations, risk and new information. It may affect their ability and willingness to expose themselves to such situations in the first place (Fleming, 2008b).

Involvement in new situations with new people can be somewhat stressful for most people. Using Bowlby, we can understand that an adult who is insecurely attached may be unwilling to put themselves into positions which will cause anxiety or feelings of inadequacy. Such individuals will find these situations stressful and threatening and they will not be optimistic about their ability to cope. For some students even coming into the college or the class may have required overcoming negative feelings and their active and cheerful participation in the learning situation can not invariably be expected.

Adult students are not homogenous and they may not be able and willing to absorb the teacher's input and behave reasonably according to expected social norms, to ask questions when perplexed and participate with others in an open co-operative manner. A student who fails to do so can often be considered awkward or difficult. The material being taught is usually seen as neutral and that the students can absorb it rationally and impersonally. Attachment theory, however, makes us aware that an adult learner may well have strong emotional attachments to previous knowledge or

ways of seeing the world. Lisa, the student mentioned in the introduction who was upset by a discussion on romance which challenged her psychological and cultural assumptions. She took a risk that resulted in growth. Any such challenge to a world view, or exposure to discussion of it, can be threatening and even if accepted, can involve a genuine sense of loss and a period of adjustment to the new reality. Their reaction to this 'strange situation' and the ability to cope with it is influenced by attachment styles (Fleming, 2008b). 'The meaning that learners attach to their experiences may be subjected to critical scrutiny. The teacher may consciously try to disrupt the learner's worldview and stimulate uncertainty, ambiguity, and doubt in learners about previously taken-for-granted interpretations of experience' (Tennant, 1991).

For Fleming (2008), attachment styles and internal working models are examples of psychological filters that affect ways of feeling and acting in adulthood. The process of transforming a frame of reference begins with a 'disorienting dilemma' – similar to the 'strange situation' in which what was taken for granted does not hold any more. Questioning assumptions or long held ideas can cause considerable discomfort (Cranton, 1994; Andreotti, 2010). For a student who has a secure attachment style, this causes less anxiety and less likelihood of avoiding the situation. By engaging with the strange situation or disorienting dilemma, a transformed, better frame of reference results and a more secure attachment style, making it more likely that further strange situations (learning, relationships, social) will be met more confidently and so on (Fleming, 2008b).

Similarly, Atherton (2010), in considering failure to learn, looks at the 'cost' of learning and suggests that learning can be traumatic especially what he refers to a 'problematic supplantive learning' and 'troublesome knowledge.' Such learning involves the replacement of existing knowledge, not just adding to it, and can be particularly difficult when it touches on sensitive areas such as politics or religion where there might be a considerable emotional attachment to the established learning and ways of thinking. This learning is as a process of personal change and involves a degree of loss, (the cost) with implications for the student's self-concept. This involves risk for the student and a potentially difficult transition in assimilating the

new knowledge. A student may not be willing to pay the cost of such learning resulting in a 'failure' to learn.

Being a mature student involves risk, and one's attachment style affects how willing one is to encounter risk and how the associated stresses are dealt with or avoided. The tutor's ability to provide a secure base for learning, exploration and adaptation is of key importance. Tutors' own attachment style also impacts on the situation and their ability to form supportive learning relationships. West (1996, p.x) suggests that education can offer 'a supportive space during periods of profound change and uncertainty' where adults can explore their identities and possibilities of new lives.'

Finding ways of encouraging mature students, especially those whose attachment style make them risk avoidant, to take the first step and ensuring that the first step is successful, is clearly vital for those engaged in the education of adults. Also recognising how attachment styles may impact on classroom relationships and participation as well as encountering new ideas and possibilities, offer ways of understanding the mature student. Learning occurs in ideas encountered in the classroom and in meeting the challenges of being a mature student. My participants reported the unsettling effects, disorientation and questioning that participation in education can (and should) produce. Bowlby's ideas help focus on the importance of a supportive environment and people who can offer a 'secure base' for learning and change.

Attachment theory clearly has much to contribute to discussions on how adults encounter new experiences and risks including those of being involved in an educational setting (Fleming, 2008b). Ideas on children and childrening are also part of a wider socio-cultural context and as parents' behaviour is influenced by their own experience of being parented, the process can be seen as an example of cultural reproduction. This can intersect with thinking about class, habitus and cultural capital which reinforce how individual psychological processes are impacted by social and cultural practices.

West (1996) maintains that the cultural and social impact on individual psychology and stresses the need for satisfactory relationships and a supportive environments for the development of the self. Without adequate support and relationships, a false self may emerge, experienced by adults as meaningless, inauthentic and empty. He suggests that early damage to the emergent self can be repaired through 'good enough' (Winnicott, 1971) relationships in later life and that education might offer such an opportunity for reparation through contact with supportive, caring people. Education, he suggests, is 'potentially a space in which to manage and transcend feelings of marginalization, meaninglessness and inauthenticity in interaction with others; in which it is possible, given their support and encouragement, to compose a new life, a different story and a more cohesive self' (West 1996, p.10). This aspect of education helps to illuminate the emotional and relational needs of adult learners in unstable times and the processes of change in the emerging self and resonates with the experience of my group of mature students. The college offered a safe place and a moratorium that facilitated risk taking, learning and development.

Recognition

Honneth (1995) tries to overcome the separation between psychological and sociological approaches to the development of identity. His theory of recognition emphasises the importance of intersubjective processes in personal development and offers a way of seeing some of the students' biographies. Honneth suggests three types of relation to self: self-confidence, self-respect and self-esteem. In the first, an individual's self-confidence is established in relationships of love and friendship. The child's growing trust in the continuity of the mother's care eventually allows it to see him/herself as separate from and independent of the mother and develop 'the capacity to be alone.' This is reminiscent of Bowlby's Attachment Theory (Fleming, 2008b) and the idea of a 'secure base.' Recognition, in the form of love and care, is necessary for the formation of self-confidence and individual identity and is a prerequisite for the development of self-respect and self-esteem.

Self-respect involves recognizing a person as capable of acting on reason. To have self-respect, then, is to have a sense of oneself as a person, that is, as a 'morally responsible' agent or, more precisely, as someone capable of participating in the sort of public deliberation that Habermas terms 'discursive will formation' (Anderson, 1996, p.xv). Fundamental to this is the idea of self-respect is the notion that this

recognition and the facility to exercise it, is enshrined in legal rights. Respect is shown to others by recognizing them as having legal rights.

Self-esteem is the third form of recognition which is built through the respect one earns through work and involvement with others. The conditions for this kind of recognition are determined by what is a considered worthwhile contribution in that society. Mutual recognition of work creates a feeling of solidarity in the community.

Using Honneth's (1995) idea of 'Recognition,' Peter's story (Chapter 5) can be understood as 'mis-recognition' where his dyslexia and perceived lack of academic ability meant he was not recognized, and did not recognize himself, as someone capable of equal participation. With no recognition of his self worth his personality could not fully develop. Being seen as 'the boy' implies the idea that he was not fully an adult deserving of equal status or afforded equal rights. He saw education as a way of achieving recognition.

People who qualified were treated better because education was looked on as something, it was valued. And the fact that I couldn't accomplish it...I was undervalued as a result of it. Peter

My self esteem has gone up enormously and my idea of who I am as a person ...I value myself more, do you understand? I've achieved academically. Peter

Education is a form of public recognition and respect and is an important part of identity development. This notion expands what is meant by 'needing a qualification' beyond simply meeting the demands of the labour market.

For James, literally being recognized in the corridors of the college was important to him. People knew him (he is in his fourth year in the college) and knew his name and he saw himself as a valued member of the class and the college community. He received very positive feedback for his work in the local community radio station. He saw these interactions and affirmation as vital in bringing out 'the real me.'

Other participants' stories underline the importance of being recognized as capable students and how seemingly causal words of encouragement were vital, especially when confidence had been damaged by previous experience. The adult relationship

with tutors also reflected a relationship of respect and support which created a 'good enough' space for growth (Winnicott, 1971). The concept of recognition is a useful way of seeing both the student's motivation to come back to education, their reasons for staying and the process of transformation in the 'emerging self'.

Walters – 3Rs

Walters (2000) suggests that many mature students' journey into and through higher education follow a similar route. The 3Rs framework she formulated is located in the context of adult development and psycho-social transitions. The 3Rs stand for Redundancy, Recognition and Regeneration.

The first concept, 'redundancy' she sees as a common and major motivating factor in motivation. She sees redundancy as positive and part of the dynamic process of continually restructuring the past and reinterpreting it as perspectives change and the old perspective becomes redundant. Perspective change is brought about by 'life dilemmas' often involving changes which can be upsetting as they precipitate a change in the self concept, questioning and a restructuring of one's frame of reference. Unemployment or children growing up involves such a change involving the redundancy of role and skills and prompting a search for a new identity. Walters suggests that the process of questioning may have already have begun, but whether the process is reactive or proactive, redundancy is still the issue. Redundancy can refer to, socio-economic circumstances, relationships, a frame of reference or a self-concept which one wants to change.

The second concept 'recognition' is essential for progression. Recognition means recognising the reality or desirability of change or the possibility of change. However, recognising the internal process as well as the external event, for example unemployment, is crucial. Walters quotes Mezirow (1978, p. 105):

...this is not simply role taking for it also implies a conscious recognition of the difference between one's old viewpoint and the new one and a decision to appropriate the newer perspective as being of more value.

Recognising appropriate action and revising behaviour if necessary may be assisted by the recognition of a suitable role model or support embodying the new perspective. This can be a teacher, friend, organisation or fellow students.

Regeneration is the third concept, and following the previous two, is the outcome of the developmental process. Again, drawing on Mezirow, having restructured one's frame of reference to one that is more appropriate and satisfying, one participates again in society on new terms defined by the new personal meaning perspective. Education offers the opportunity to be challenged and to rethink present knowledge and values. Walters maintains that the content of education is unimportant; what matters is 'the opportunity to examine one's attitudes and behaviour-to reflect on one's life history' (Walters, 2000, p.7). While it could be suggested that studying psychology or sociology encourage a more direct reflection and a different perspective (Mercer, 2007; Guiffrida, 2009) all students I interviewed reported a revised sense of self, underlining the idea that 'it's not what's done that is important, but how it's done.'

This framework offers a useful way of looking at the students in my cohort. They all experienced redundancy of some kind. For some their role changed or they recognised a disparity between the way they were and other 'possible selves' (Markus & Nurius, 1985). For some, circumstances allowed them to act on a long held desire to go back to education. Although some mentioned how acquiring new knowledge was interesting and relevant to them, the actual new information or subject matter was given very little attention. They saw themselves differently, as people who had more information and who had the skills to acquire it. The real change was in themselves, their sense of identity, confidence, self esteem and control.

Mezirow speaks of perspective transformation being 'epochal' where an individual faces a sudden crisis and is forced to take stock of their lives as previous perspectives are inadequate to deal with the new situation. Becoming unemployed or the death of a spouse can be such changes. For some, especially for students dealing with social or psychological ideas, a change had resulted from an encounter with subject matter that exposed them to new insights. For example, in media analysis, the concept of 'ideology' is a 'threshold concept' that can go beyond the subject and deeply affect the student's way of seeing the world. He also sees the possibility of a more evolutionary change or 'a cumulative progressive sequence of insights resulting in changes in point of view and leading to a transformation in habit of mind.' This

change comes about 'by a series of transitions which permit one to revise specific assumptions about oneself and others until the very structure of assumptions becomes transformed' (Mezirow, 1981). This seems particularly relevant to the students in this study and their progress to a revised sense of self.

Support and care emerges as a critical element in dealing with transitions and the development of a new sense of identity. It is also addressed by the mature students in my group who acknowledged the key role of family, friends and tutors in the college and the general atmosphere in which they felt comfortable, welcome and respected. The literature suggests that a supportive environment is not just a fortuitous pleasant extra, but an essential aspect of negotiating the transition to a revised self. Noddings' observation (1992, p.27) that 'caring is the very bedrock of all successful education' is particularly relevant in this context and implies an obligation to provide 'a safe place' and attend to the emotional as well as intellectual needs of adults.

Section 3

Finding a Voice

The core concept of 'finding a voice' represents the 'revised self,' the outcome of the process of change for the mature students in this study. The quotations from the students in Chapter 6 show how forcefully and positively they expressed their conviction that they had changed as a result of their time in the college. They had a revised sense of self. 'Voice' is used here to include not just the act of speaking but a sense of agency, respect and entitlement to participate.

Greater confidence came about as a result of doing well and receiving positive feedback on assignments, especially the first one. This is an obvious and recognised rite of passage for many students with special associations and anxieties for those who are older. It is also an important form of personal and social recognition. Doing well acted as a motivator and created a desire to do even better. Similarly, developing a sense of being confident with computers marked another significant achievement and a major landmark for some. It is clear, however, that the feeling of confidence extended beyond academic competence into other aspects of the students' lives and was largely responsible for the 'revised self' they reported. Mercer (2007), from her

study of mature students, also suggests that academic and personal growth are not mutually exclusive categories, but inter-related. She points out that researchers consider personal and academic growth as separate and have not considered personal development an integral part of the process. She speaks of a 'dual process between personal and academic growth' which supports what emerges from my study. It is clear, in the comments of the students, that the two are inter-related. New knowledge is not much mentioned, although it is welcomed as offering a new understanding. Of much greater importance is a more general feeling of knowing more, being more aware and 'knowing what's going on.'

You're filling yourself up with something and it does show in everyday things, no matter when you're at home or whatever. Simply just in conversation.

Aoife

When I hear people talking about why they're doing things...like the elections there...you might be sitting in front of the TV and you'd understand what's going on, that's important. You know what's going on. Daniel

Personal as well as academic development that shows 'in everyday things' and the general feeling of being more informed and competent is evident in these quotes. Knowing 'what's going on' also carries with it a sense of involvement, inclusion and acceptance.

Academic achievement is also significant for many mature students as their previous experiences may not have led them to see themselves as competent in this area. Going back to education may have been the fulfillment of a long held wish and realise a part of the self that had not been addressed earlier. Mercer (2007, p.26) calls this 'resolving the past' and she sees this as reclaiming a part of the self that had been missing and having a central role in the renegotiation of a revised identity. Unfinished business was certainly a factor for some of my group. This could be seen as resolving an imbalance between an actual self and a desired 'possible self'. For some, this included a self that was recognised by others as well as themselves.

The successful negotiation of challenges and the accompanying risks in the transition to becoming a mature student and while in the college also offered the potential for

growth and contributed to developing self confidence and a different perception of self.

Confidence is also experienced in relation to others. As mentioned in Chapter 6, when asked to elaborate on what their new confidence meant to them, most of the students referred to speaking, being able to speak and having things to say. I called this concept 'finding a voice' as it has metaphorical connotations of not having had a voice, of speaking up after a long silence and finding that you have something to say. It also implies speaking up for yourself, asserting a claim for recognition and a right to participate. It is not only a personal development but it is also social as it manifested in dealings with others. It describes and symbolises the transformation described in the students' narratives.

Em, well yeah, I'd say confidence. I suppose yes, confidence wise and then you know sometimes especially when you're coming from an area... and you feel you haven't completed your education and somebody tries to have a conversation with you about something academic, you're kind of saying 'I'm not opening my mouth here, in case you make a thick out of yourself' or whatever, but like I don't feel that as much, like I never really did feel it hugely, at the same time I'd probably go 'I wouldn't be too sure about that' so I wouldn't ...I'm not going to argue a point with somebody even though in my mind I'd be saying 'I don't think that's right.' Caroline

Vera spoke about having a wider range topics of conversation and being 'on a par'.

I'm on par with them to a certain extent and I can voice a viewpoint that I wouldn't have voiced before. I wouldn't have had the knowledge, you know? Vera

Feeling 'on a par' implies feeling unequal previously. She can now engage on an equal footing. There were many more comments of this kind. Many relate to greater social confidence and competence and being able to take part in conversations that that they would have felt excluded from or unable for before. Some of the comments relate to questioning. Daniel queries his bills and looks for a better deal, Jason can ask questions, and some speak of having things to say. Having confidence and being able take part in conversations in ways they felt unable to previously indicates a new sense of agency, assertiveness and social participation. Caroline's comment above reflects the influence of class 'coming from an area' and her feelings about leaving school

early which resulted in an unwillingness to be part of a conversation despite her instinctive feelings. Finding a voice enables participating from a position of equality

Many spoke of previously being unable or unwilling to speak in certain situations 'afraid to say something wrong' and being afraid to speak in case she made 'a thick' of herself, not challenging bills and not being an equal conversational partner with her children, or not being able to express themselves. Bourdieu (1992, p.127) described habitus as 'a socialised subjectivity'. It is found, not just at the level of the conscious, but in unconscious, unthinking behaviour. Language is key part of the embodied habitus in its codes and registers and in its use or non use. Mastering the language of assignments and academic language generally is a major step for students. Being able to use language differently, in different contexts marks a change in the habitus. The change experienced is expressed in very strong and emotive terms: 'a metamorphosis'. 'It's like a weight lifted off your shoulders,' 'not afraid to speak up'. None of these people were 'shrinking violets' before they came to college. They were competent and sociable and yet felt intimidated and unable to participate adequately in certain circumstances.

Belenky et al.(1986, p.14) suggest that while visual metaphors imply distance and separateness, whereas voice implies connectivity, proximity and relationship. It acknowledges subjectivity and intersubjectivity.

Voice is a fundamental feature of being human. Talking to others is a way of telling yourself what you think, 'I know what I think when I hear what I say' (in Tarule, 1996). Opinions are often worked out in the act of discussion and this is dependent on an ability and willingness to take part in such discussions. The students are talking about being able to take part in discussion and opinion forming without the feelings of inferiority that they were conditioned to.

For Belenky et al. (1986, p.44) Dialogue or 'real talk' implied 'real listening', exploration and the conditions that mean 'half-baked or emergent ideas can grow.' 'Connected Knowing' is marked by 'really listening'. It involves the 'capacity to attend to another person and to feel related to that person in spite of what may be enormous differences (Belenky et al., 1986, p.143). Likewise, for Belenky et al.

separate knowing is knowing at a distance and involves 'the doubting game' (Elbow, 1973, in Belenky et al., 1986, p.104), it is adversarial and seeks to find fault. Connected knowing, on the other hand, demands intimacy, equity and empathy rather than judgment. Instead of 'the doubting game' it involves 'the believing game' (Elbow, 1973, in Belenky et al., 1986, p. 113) and seeing the other in their own terms. Similarly, Brookfield (1987, p. 238) speaks of the importance of 'learning conversation' for developing critical thinking. Such conversations involve reciprocity and an acceptance of disagreement, diversity and a multiplicity of interpretations.

Dialogue with oneself, 'talking to oneself,' is a reflexive process that involves the interpretation of ourselves to ourselves and the construction and reconciling of our various identities. We define ourselves through our narratives and their interaction with the narratives of others. As McIntyre (1981, p.194), says 'The narrative of anyone's life is part of an interlocking set of narratives'. Voice is therefore fundamental to our construction of identity. It carries with it the idea of being listened to, afforded respect and recognised (Honneth, 1995). Taking part in discussion is fundamental for the individual and for social interaction, involvement and change.

Couldry (2010. p.1) suggests that 'voice is the capacity that humans have that enables them to 'give an account of themselves' Neo-liberalism, he maintains, as a discourse that has come to dominate the world, denies that voice matters. Through 'a reductive view of politics as the implementing of market functioning' (Couldry, 2010, p.2) he suggests that there is a 'crisis of voice' under neoliberalism. He also uses 'voice' as a 'connecting term' that challenges neoliberalism's claims and enables an alternative view based on people's capacity for social cooperation based on voice. Dialogue, for Freire, was necessary to develop critical consciousness and an ability to question and change the status quo.

To exist, humanly, is to *name* the world, to change it. Once named, the world in its turn reappears to the namers as a problem and requires of them a new *naming*. Human beings are not built in silence, but in word, in work, in action-reflection (Freire 1970 b, p.69).

For Freire, the word involved action and reflection and praxis was about balancing both.

Harre and Gillett (1994, p.175) also stress the importance of discursive skills. 'As human beings extend their discursive skills, so they expand their consciousness; that is, they extend the range of matters to which they can attend and of which they are aware'. This 'new awareness' came across in the interviews. 'Seeing things differently' implies an awareness of other possibilities and changed frames of reference. Having a voice allows a questioning of conventional wisdom.

Finding a voice, speaking up, questioning, knowing what's going on are crucial for the fulfillment of the individual and imply a greater ability for meaningful interaction with others and engagement in society. For Belenky et al. (1986, p.134) to be silenced is to be voiceless, selfless and mindless, subject to the authority of others. It is fundamental for reflexive learning (Habermas, 1975) and fostering co-operative discussion, communicative action (Habermas, 1984) and a revived public sphere. It is fundamental too, for an ability to participate in discussions about differences and develop new understandings that are not determined by the experiential context and the ideologically predetermined (Brookfield, 1987, p.357). This implies a respect for difference and a willingness to subject our own views, as well as those of others, to critical scrutiny. Tarule (1996, p.276) sees voice as an 'indispensible aspect of knowing and thinking' and dialogue as 'making knowledge in conversation.' Finding a voice means being able to participate in discussion and 'making knowledge.' It is fundamental too for countering a system of education and world view that promotes a 'culture of silence' Freire (1970b) in which preparation to work within the system minimises critical questioning and awareness.

Conclusion

This chapter looked at the concepts of Latent Self, Emerging Self and Revised Self from a more theoretical perspective. 'Finding a voice' was used to describe and symbolize the 'revised self' and personal development described by the participants which was the most important, if unexpected, part of their experience. It is not just an added bonus of the instrumental learning and training in FE but the main benefit with significant implications for the students themselves, their families and society more generally.

Chapter 8

Conclusion

It was suggested in the introduction that the further education sector is a major provider of adult education. The history of FE in Ireland, its status of part of VEC provision and the climate of neo-liberalism in which it emerged is the important context of this study. The aim was to explore the experience of mature students, their reasons for coming to an FE college and what they consider the benefits of participating. I sought to explore the issues and themes through the voices of students largely neglected in the literature. Although the majority of students in the sector are now over twenty one and classed as mature students most of these are younger mature students and this study focused on those who had been out of education for some time. It is not suggested that mature students have more difficulties that younger students, but that they are different and that they are attending institutions that, despite the growing number of older students, were established for school leavers and usually still seen as the prerogative of younger people.

The findings

The findings indicate that for mature students FE is a site of significant learning that goes beyond vocational preparation. The students in this study felt they had changed significantly as a result of their participation.

Virtually all the students surveyed and all of those interviewed were very glad they had come back to FE and said that they had a very positive experience despite difficulties with lack of money, juggling other commitments and assignments. Most came to FE because of the level and structure of the courses or because they wanted a particular course that was not available elsewhere. Some just wanted to come back to education with no particular course or career in mind. FE is available to those who would not have qualified for HE directly and offers progression to many third level courses. It is therefore important in providing alternatives routes to HE and in providing alternatives to HE and in offering 'second chance' education. It is also seen as convenient, accessible and relevant to those do not see themselves going to HE.

Not having work or being in an unsatisfying job, becoming unemployed and a desire for a qualification were commonly stated reasons for the decision to go back to college. There were also other factors such as children being grown up or starting school that allowed attendance at a course made it 'the right time' for the individual. A personal contact or recommendation often provided the immediate spur and helped decide on the particular college. Significant too was some affinity for education and learning and, in many cases, a long-held desire to go back and a feeling that they had missed out by not pursuing education earlier. The interviews, however, revealed that there were other complex motivations best understood in the context of the individual's life.

For nearly all, a process of change had already started as their roles and sense of self had altered as a result of becoming unemployed or their children growing up. The 'emerging self' is the name given to the process of change in FE characterized by a number of milestones or developmental stages. Often there were other personal changes and adjustments too. Adjusting to being a mature student and developing a sense of academic competence is part of this process The successful negotiation of these stages, which is not straightforward or linear, contributes to the identify shift described by the students. The supportive environment described by the participants is a key factor in the process.

I used the conceptual label of 'Finding a voice' to describe the revised self or changed identity the students claimed. When asked what they had got from coming back to education, all said 'confidence.' I was surprised by how general this was and how strongly expressed it was by most. This went far beyond academic confidence and extended into other personal and social aspects of their lives. It was not just a matter of feeling able to speak up in situations in which they would have been silent before, it combined a sense of having something to say and a more general competence that involved a greater awareness and 'knowing what's going on' and a feeling of entitlement to engage and question. For the students this development was unexpected and they were pleasantly surprised by it. All, whatever their initial reasons for coming, felt they had changed during their time in college, most very significantly. They were surprised too by their enthusiasm for learning and how enjoyable their experience was.

Finance is a major issue. Some students were supported by partners, savings or parttime work. VTOS and BTEA are vital supports without which returning to education would be impossible for many. The participants in this study also had the vital emotional and practical support of their extended families.

Implications

The students in this study found their time in FE enjoyable and beneficial; however, the results give rise to some comments about the system.

The increase in numbers of mature students in FE occurred by default, much like the growth of the system overall, without any stated policies or target quotas. This needs to be addressed in the context of a review of the role and purpose of the sector.

At present, while FE generally is inclusive in its policy and practice, there is no consistency between colleges, even within the one VEC, or within the same college. This in effect leaves it up to the course co-ordinator or those interviewing prospective students to interpret admission policies. In a time of high demand for places and in a system whose main stated purpose is to prepare students for employment, mature students, especially older mature students, may be seen as less 'worthy' of a place than younger applicants. In this context, it is important to ask how we in FE can formalize a commitment to improving access and success and enhancing the experience of non-traditional students. This could involve looking at targets and quotas and liaising with community education facilitators and adult education organizers who are also under the remit of the CDVEC. However, these sectors operate separately with very little attention given to co-operation or encouraging student progression. This needs to be addressed with a view to developing the successful 'Return to Learning' courses already operating and enhancing the routes into FE and progression within FE and beyond.

FE offers progression routes to third level education which is especially important for those who lack traditional Leaving Certificate admission requirements. Many mature students do not come to FE intending to progress but, having attended FE, many feel able to and want to. However, some come with progression in mind. This is a

significant access route for non-traditional students with implications for equality and widening participation in HE, especially among groups who are under-represented in third level education. The report 'Where Next?' (2010) affirms the importance of preliminary courses for mature students in HE. However, by requiring applicants to have achieved all the components of a full FETAC award in one session and by greatly limiting the number of places available to FETAC applicants, many HEIs are putting obstacles in the way of non-traditional applicants. Greater co-operation between the FE sector and the HEIs clearly need to be developed. This could include the expansion of specific access courses or looking at arrangements such as those in Warwick university in the UK whose 2+2 degree offers adults who lack formal qualifications an opportunity to study for a degree by spending two years in an FE college (the equivalent of 1year of a traditional degree) and the final two year in the university.

Although there have been policies in place for many years at government and college level which encourage access to HE by non traditional groups, no such policies exist for FE. Access offices and designated support officers for mature students have been successful in attracting mature students and supporting them in HE. FE has not been included in these developments. This reflects FE's second level status and the unsuitability of these structures for what happens in the sector. The implementation of the changes called for in the McIver Report (2003) would be extremely helpful in addressing these deficits and allow for the expansion of non-teaching roles and supports. Similarly, traditional timetabling and the scheduling of the 'school' day imposed by the DES greatly limit FE in the flexibility it can offer which seriously curtails the provision of part-time courses and the accumulation of modules or credits over a number of semesters or years. This prohibits the participation of adults who can not commit to full time attendance. This needs to be addressed with the DES by the CDVEC and the Irish Vocational Education Association (IVEA), the national representative association of Ireland's Vocational Education Committees. Even without structural changes at government level individual VECs and colleges can introduce and implement policies that foster inclusiveness as an agreed aim.

At the college level supportive and encouraging attitudes of teaching staff and others were considered of key importance by the students and vital to their success. Support

meant being treated with respect, as an adult and feeling that their situation and issues were understood. This was particularly important when things got difficult. Support systems failing, or the illness of a child or parent often involves missing classes and work piling up. At such times the support of the staff and flexibility with deadlines often makes the difference between staying and dropping out. Having easy, informal access to staff who were willing to give their time and listen was very highly valued. The course co-ordinator, who has overall responsibility for the class, was generally seen as a key source of support.

Although it is impossible and undesirable to prescribe for all aspects of inclusion, and most teachers' sensitivities serve them and all their students well, there are aspects of good practice that contribute to inclusion that could be cultivated and disseminated. Very often this good practice applies to all students with some awareness of the particular characteristics of various groups. For example images in publicity material and brochures can create a sense of inclusion. Particular care for the sensitivities of those who have been out of education for some time is especially important at the initial contact and at interview. Likewise an awareness of the difficulties that such students can face in the initial period of adjustment and other 'strange situations' and with the first assignment can be particularly helpful. This highlights the need for appropriate training and recognised qualifications for those working in the sector. Individual colleges can, however, do much to increase awareness of these issues. Staff meetings and local in-service sessions, as well as disseminating good practice, can be used to highlight inclusiveness and make it a stated policy in colleges' mandatory quality documents and putting procedures in place to ensure implementation of these policies. This stated commitment to mature students and other non traditional groups may require a re-examination of admission policies and entry criteria which again raises the question of the role of FE. This question is of particular importance now as legislative change in 2012 will mean major transformations in the sector as a new Further Education and Training Body, Solas, will be established. It is vital that the FE sector, the CDVEC and individual colleges have a clear vision of their purpose and value and a strong voice in these significant developments in Irish education.

Limitations of the study

Not all mature students do well in further education. It is one of the limitations of this thesis that it did not include a focus on students who left. This would be interesting work for a further study. There is no co-ordinated disaggregated information available on non-completion in FE, and while non-completion is not necessarily a bad thing, knowing why students do not complete could enable better preparation and provision to be put in place. Hidden Disadvantage (2010) found considerable variations in drop out rates in HE with those from lower socio-economic groups more likely to drop out. Anecdotally and from my own experience, mature students are less likely to drop out than their younger colleagues. There are a number of factors that contribute to noncompletion by mature students in FE. These include: students feeling that they are doing the wrong course or the course not being what they thought it was, the course being more academically demanding than expected, the first assignment proving too stressful and falling behind with coursework, often as a result of domestic demands requiring absence from college. This usually applies to women with younger children. Most of these could be addressed with more concerted planning, monitoring and advice. Students sometimes drop out because they feel they are not being noticed, they do not feel involve in the college or their concerns are not being taken seriously. This highlights the need for awareness, sensitivity and timely feedback and consultation. Similarly, a follow up study some time after graduation would provide information on the longer term outcomes for the students.

Competing discourses

There are two discourses operating side by side in FE. One is characterised by the neo-liberal discourse of individualism, accountability and servicing the needs of the economy described in Chapter 2, and the other provides the supportive environment and individual care that is fundamental to the changes described by the students. The neo-liberal influences at the official and policy level are being mediated by the staff of the college who deals directly with the students and who exercise a degree of flexibility in interpreting regulations, guidelines and course content. Perhaps this is unsurprising in a system that owes its existence to stretching official boundaries and continues to have an ambivalent relationship with the DES. The practice on the ground, in general, reflects a belief in an ethos of care and a broader vision of education. This vision includes a concern for the development of the person, the encouragement of a questioning critical approach that goes beyond the transmission of

skills and competencies and meeting outcomes. However, much of this depends on individuals and all teachers in FE do not operate in this way. As mentioned in Chapter 2, there is some evidence that this ethos is being eroded with more teachers coming from industry and changing practices. This highlights the issue of the lack of a shared vision of FE and the need to support present good practice and make it explicit in policy and the training of those working in the sector.

The benefits of participation in FE are not just for the individuals and their families. Interestingly, in the context of the dominant neo-liberal climate, the increase in confidence, sense of agency and 'knowing what's going on' mean that these students have a greater ability and willingness to participate in the community and contribute to an enriched public sphere. 'Finding a voice' implies a greater capacity for discussion and the development of a more critical aspect to their reasoning that enhances their functioning as active citizens to the benefit of society more generally. Teaching and learning knowledge and skills for employment are an important part of the work that is done well in FE. It is important to recognize the other significant benefits as well as these economic, employment-related aspects that dominate discussion of FE. The students felt they were more employable, although they were all too aware of the difficulties of getting employment in the current climate and at their age. They also felt more able and inclined to question established practices and systems. This study demonstrates the limitations of the neo-liberal customer oriented model of education as responding to students' needs. Many of my cohort recognised they were unclear of what their precise needs and were surprised by the eventual result. In a system dominated by outcomes, it is a reminded that the process of education is as important as the product. The development of discursive skills allows the type of conservations through which, for Habermas, real needs can be discovered. Our frames of reference are constructed in part by the socio-cultural conditions of neo-liberalism and the reclaiming of reason from this distortion is an adult learning project (Fleming, 2008a). These students are better equipped to engage in this project. Their stories show that learning for living and learning to make a living need not be incompatible.

A final thought

From a personal perspective, as a mature student myself and sharing many of the concerns and experiences of the participants, I found it easy to empathize and identify with the students' stories. I tried to be a participative and collaborative enquirer and to be mindful of the need for reflexivity and an awareness of how the research was impacting on the participants and on me. The interviews asked the students to reflect on their experience and to explore and articulate their thoughts and feelings in a way that they would not normally do. They said they found the process beneficial and interesting and were glad of the opportunities it offered to develop their awareness and personal insights, as I was myself. I felt my skill as an interviewer improved during the course of the research and the use of the participants' voices lend authenticity and credibility to the findings. Overall I think the methodology worked very well to produce rich and insightful personal accounts of experience. These accounts offer a way of understanding FE as facilitating a process of change for mature students.

As a researcher and as someone working in the area, the process offered an opportunity to reflect on my own career, teaching in FE and the role of the sector itself. I was surprised by the findings. I had expected that there would be much talk about subject matter and its relevance to the world of work, classroom and assessment practices and learning as an older person. These barely featured as the other themes emphatically emerged. I was surprised too, by the complexity of the motivation to return to education and what becoming a student meant in the individuals' lives. Most surprising was the unanimity and strength of feeling the students reported about the changes they experienced. Overall, for me, the study offered a chance to see FE from a different vantage point. It succeeded in 'making the familiar strange' and made me aware of aspects of FE and its students that had become unnoticed through familiarity and dealing with the constant immediate demands of daily practice. It renewed and increased my admiration of what the sector can do and the importance of preserving what it does well.

This is a small study with no claims to generalisability. It uses the participants' own stories and grounded theory analysis to illuminate the experience of a group of adult learners and to increase understanding of the factors that contribute to it. It is hoped that this will contribute to an understanding of what goes on in further education and

what it offers mature students and its important role in education in Ireland. It is hoped too, that in the face of policies that promote individual learning for the workplace, that this other important learning might be recognised and valued.

References

- Addreotti, V. (2010). Global Education in the 21st Century: Two different perspectives on the 'post' of postmodern. *International Journal of Development Education and Global Learning* 2 (2) 5-22.
- Ahl, H. (2008). Motivation theory as power in disguise. In A. Fejes & K. Nicoll (Eds.), *Foucault and Lifelong Learning*. Oxford: Routledge.
- Ainley, P. (1993). Class and Skill: Changing divisions of knowledge and labour. London: Cassell.
- Anderson, D., Brown, M. & Rushbrook, P. (2004). Vocational education and training. In G. Foley (Ed.), *Dimensions of adult learning: Adult education and training in a global era*. Maidenhead: OUP.
- Anderson, J. (1996). Translator's introduction. In A. Honneth (Ed.), *The struggle for recognition: The moral grammar of social conflicts*. Cambridge: Polity Press.
- Antonesa, M., Fallon, H., Ryan, A. B., Ryan, A., Walsh, T. & Borys, L. (2006). *Researching and writing your thesis: A guide for post graduate students*. Maynooth: MACE.
- Apple, M. (2005). Are markets in education democratic? Neoliberal globalism, vouchers and the politics of choice. In M. Apple, J. Kenway & M. Singh (Eds.), *Globalizing Education*. New York: Peter Lang.
- Apple M. (2006) Educating the 'right' way: Markets, standards, God and inequality. New York: Routledge.
- Archer, L. (2003). Social class and higher education. In L. Archer, M. Hutchings & A. Ross (Eds.), *Higher education and social class: Issues of exclusion and inclusion*. London: Routledge Falmer.
- Arnett, J. (2004). *Emerging adulthood: The winding road from late teens through the twenties*. Oxford University Press.
- Aslanian C. B. & Brickell H. M. (1980). *Americans in transition: Life changes as reasons for adult learning*. New Jersey: College Entrance Examinations Board.
- Atherton, J. S. (2010). Teaching and learning; Contents (on-line) retrieved 30th June 2011 from
 - http://www.learningandteaching.info/teaching/contents.html
- Avis, J. (1997). 'What's this got to do with what I do?' Contradictory Views: Students in further education. *Journal of Vocational Education and Training*, 49(1), 81-106.

- Baker, J., Lynch, K., Cantillon, S., Walsh, J. (2004). *Equality from theory to action*. New York: Palgrave Macmillan.
- Ball, S. (2003). Class strategies and the education market: The middle classes and social advantage. London: Routledge Falmer.
- Bates, I. (1991). Closely observed training: an exploration of the links between social structures, training and identity. *International Studies in Sociology of Education*, 1, 225-243.
- Baxter, A. & Britton, C. (1999). Becoming a mature student: Gendered narratives of the self. *Gender and Education*, 11 (2), 179-193.
- Baxter, A. & Britton, C. (2001). Risk, identity and change: becoming a mature student. *International Studies in Sociology of Education*, 11 (1), 87-104.
- Bean, J. P. & Metzner, B. S. (1985). A conceptual model of non-traditional undergraduate student attrition. *Review of Educational Research*, *55*, 485-540.
 - Beck, U. (1992). Risk society: towards a new modernity. London: Sage.
- Belenky, M. F., Clinchy, B. M., Goldberger, N. R. and Tarule, J. M. (1986). *Women's ways of knowing: The development of self, voice and mind.*: New York: Basic Books.
- Blair, A., McPake, J. & Munn, P. (1995). A new conceptualisation of adult participation in education. *British Educational Research Journal*, 21(5), 629-645.
- Blair, T. (1998). *The learning age: A renaissance for a new Britain*. London: DfEE.
- Boshier, R. (1971). Motivational orientations of adult education participants: A factor analytic exploration of Houle's typology. *Adult Education*, 21, 3-26.
- Boshier, R. (1976). Motivational orientations revisited: Life-space motivation and the educational participation scale. *Adult Education*, 27, 89-115.
- Bourdieu, P. (1986). The forms of capital. In J. Richardson (Ed.) *Handbook of theory and research for the sociology of education* (New York, Greenwood), 241-258
 - Bourdieu, P. (1996). The state nobility. Cambridge, U.K.: Polity Press.
- Bourdieu, P. (1998). *The essence of neo-liberalism*, Le Monde Diplomatique, Paris. Retrieved 6th June 2010 from www.mondediplo.com
- Bourdieu, P. (1998). The myth of 'globalisation' and the European welfare state. In Nice, R (Ed.), *Acts of Resistance. Against the Tyranny of the Market*. New York: The New Press.

- Bourdieu, P. & Passeorn, J.C. (1977). Reproduction in education, society and culture. London: Sage.
- Bourdieu, P. & Wacquant, L. (1992). *An invitation to reflexive sociology*. Cambridge: Polity Press.
- Bowlby, J. (1979). The making and breaking of affectional bonds. London: Tavistock
- Brannick, T. and Coughlan, D. (2007). In defense of Being "Native": The case for insider academic research.. *Organizational Research Methods*, (10), 59-74.
 - Brewer, R. (1986). Research for Whom and by Whom: A reconsideration of the black social scientist as insider. *The Wisconsin Sociologist*. 23 (1), 19–28.
- Brine, J. & Waller, R. (2004). Working-class women on an access course: risk, opportunity and (re)constructing identities. *Gender and Education*, 16(1), 97-114.
- Brookfield, S. (2005). The power of critical theory, liberating adult learning and teaching. San Francisco: Jossey-Bass.
- Brookfield, S. (1987). Developing critical thinkers: Challenging adults to explore alternative ways of thinking and acting. Milton Keynes: OUP.
- Bruges-Copenhagen Process (2002). *Vocational Education. Retrieved on* 12th October 2010 from <u>www.europeunit.ac.uk/eu-policy-education/vocational-educatiion-and training.cfm</u>
 - Bryman, A. (2008). Social research methods. Oxford: OUP.
- Burke, A. (2007). Crosscurrents in the competencies/standards debate in teaching and teacher education In Dolan, R. and Gleeson, J. (Eds) *The competences approach to teacher professional development: Current practice and future prospects*. Dublin: Marotone Press.
- Byrne, J.G. J., O'Connor, T. G. T., Marvin, R. S. & Whelan, W. F. (2005). Practitioner review: The contribution of attachment theory to child custody assessments. *Journal of Child Psychology and Psychiatry*, 46, 115-127.
- Casey, L. (2009). *Pathways to competence and participation in the digital world*. Unpublished Ph. D. thesis, NUIM.
- CEDEFOP, (2008). (European Centre for the Development of Vocational Training.) (2008). *The shift to learning outcomes: Conceptual, political and practical developments in Europe*. Luxembourg: Office for Official Publications of the European Communities.

- Commission of the European Communitite (CEC) (1995). White Paper on education and training, teaching and learning: Towards the learning society. Luxembourg: Office for Official Publications for the European Communities.
- Charmaz, K. (2006). Constructing grounded theory: A practical guide through qualitative analysis. London: Sage.
- Chodorow, N. (1978). *The reproduction of mothering*. Berkley: University of California Press.
- Coffey, A. & Aktinson, P. (1996). *Making sense of qualitative data:* Complementary research strategies. California: Sage.
- Coffield, F. (1999). Breaking the consensus: Lifelong learning as social control. *British Educational Research Journal*. 25(4), 479-499.
- Colleran, N. (2006). Issues and Tensions in Adult Education in Ireland. In Colleran, N (Ed.), From the Inside Out: Emerging perspectives in adult and further education in Ireland. Nenagh: Tipperary North V.E.C.
- Colley, H., James, D., Tedder, M. & Diment, K. (2003). Learning as becoming in vocational education and training: class, gender and the role of vocational habitus. *Journal of Vocational Education & Training*, 55(4), 471-498.
- Coon, D. & Mitterer, J. (2007). *Introduction to Psychology: gateways to mind and behavior*. Belmont, CA: Thomson.
- Connolly, B. (2007). Beyond the third way: New challenges for critical adult and community education. In B. Connolly, T. Fleming, D. McCormack & A. Ryan (Eds.), *Radical Learning for Liberation*, 2. Dublin: MACE.
- Connolly, B., Fleming, T., McCormack, D. & Ryan, A. (2007). *Radical Learning for Liberation 2*. Dublin: MACE.
- Conway, A. (2009). Widening participation in higher education and the resultant diversity of learners: A review of government policies, the academy and market demands that may influence this trajectory. Paper submitted for term review as part of the EdD at the University of Sheffield. Retrieved 10th January 2010 from www.arrow@DIT.ie/authors.html
- Corbin, J. and Strauss, A. (2008). *Basics of qualitative research: techniques and procedures for developing grounded theory*. Los Angeles: Sage Publications.
- Coughlan, D. (2007) Insider action research doctorates: Generating actionable knowledge. *Higher Education*, 54: 293-306.
- Courtney, S (1992). Why adults learn: Towards a theory of participation in adult education. New York: Routledge.

- Couldry, N. (2010). Why voice matters: Culture and politics after neoliberalism. London: Sage.
- Cranton, P. (1994). *Understanding and promoting transformative learning*. San Francisco: Jossey-Bass.
- Creswell, J. (1998). Qualitative inquiry and research design: Choosing among five traditions. London: Sage.
- Cross, K.P. (1981). Adults as learners: Increasing participation and facilitating learning. San Francisco: Jossey-Bass.
- Crossan, B., Field, J., Gallacher, J. & Merrill, B. (2003). Understanding participation for non-traditional adult learners: Learning careers and the construction of learning identities. *British Journal of Sociology of Education*, 24(1): 55-67.
- Davies, P.& Williams, J. (2001). For me or not for me? Fragility and risk in mature students' decision-making. *Higher Education Quarterly* 55, 185-203.
- Dawson, D. & Boulton, P. (2000). Returning to education: A question of balance? *Journal of Access and Credit Studies* 2(2), 164-177.
- Deakin-Crick, R. & Joldersma, W. (2007). Habermas, lifelong learning and citizenship education. *Studies in the Philosophy of Education* 26, 77-95.
- Deering, R. (1997). *Higher education in the learning society*. London: The National Committee of Inquiry into Higher Education.
 - Denscombe, M. (1998). The good research guide. Buckingham: OUP.
 - Denzin, N. (1989). Interpretive biography. Newbury Park: Sage.
- Denzin, N. & Lincoln, Y. Eds. (1998). *The landscape of qualitative research*. London: Sage.
- Department of Education and Science (1995).: *Charting our education future:* White Paper on Education. Dublin: Government Stationery Office.
- Department of Education and Science. (1999). Report of the commission on the points system.. Dublin: Government Stationery Office.
- Department of Education and Science (2000). *Learning for life: White paper on adult education*. Dublin: Government Stationery Office.
- Department of Education and Science (2004). *Overview of further education and training*. Retrieved on 9th September from: www. education.ie
- Department of Education and Science (2008). *The development and state of the art of adult learning and education (ALE)*. Dublin: Government Stationery Office.

Department of Education and Skills (2010). *Educational statistics*. Retrieve .4th April 2011 from <u>www.education</u> .ie

Department of Education and Science [UK] (1978). Special courses in preparation for entry to higher education. letter to Chief Education Officers, 2 August.

Dey, I. (1999). Grounding grounded theory. San Diego: Academic Press.

Dewey, J. (1916). Democracy and education. New York: Free Press.

Disability Act (2005). Retrieved on 12th December from www.oireachtas.ie/viewdoc.asp?DocID=4338&CatID=87

Education for Persons with Special Needs Act (2004). Retrieved 12th December from http://www.irishstatutebook.ie/2004/en/act/pub/0030/index.html

Edwards, R. (1993). *Mature women students: Separating or connecting family and education*. London: Taylor & Francis.

Expert Group on Future Skills Needs (EGFSN). (2010). *Monitoring Ireland's skills supply: Trends in education and training outputs.*. Dublin: Forfás.

Eisner, E. W. (2001). Concerns and aspirations for qualitative research in the new millennium. *Qualitative Research* 1, 135 - 145.

doi: 10.1177/146879410100100202

Enterprise Strategy Group (2004). Ahead of the curve - Ireland's Place in the Global Economy. Dublin: Forfás.

Equal Status Act (2000). Retrieved 12th December 2010 from http://oireachtas.ie/documents/bills28/acts/2000/a800.pdf

Fairclough, N. (1997). Critical discourse analysis and the marketization of the universities. In T.A. van Dijk (Ed.), *Discourse as social interaction, discourse studies:* A multidisciplinary introduction. London: Sage.

Finnegan, F. (2008). Neo-liberalism, Irish society and adult education. *The Adult Learner: The Journal of Adult Education in Ireland.* (pp. 54-74). Dublin:

Fleming, T. (2007). Picking wildflowers and orchids: Attachment theory and the implications for adult education. In B. Connolly, T. Fleming, D. McCormack & A. Ryan, (Eds.), *Radical Learning for Liberation* 2, (pp. 79-94). Maynooth: *MACE*.

Fleming, T. (2008a). We are condemned to learn: Towards higher education as a learning society. *Level 3*, 6(1) *Retrieved from*

http://level3.dit.ie/html/issue6/fleming/fleming_1.html

Fleming, T. (2008b). A secure base for adult learning: Attachment theory and adult education. *The Adult Learner*, 25, 33-53.

- Fleming, T. (2009). *James Marcia's theory of adolescent identity*. Course notes. National University of Ireland, Maynooth.
- Fleming, T. (2011). Models of lifelong learning: An overview. In M. London (Ed.), *Oxford Handbook of Lifelong Learning*. New York: Oxford University Press.
- Fleming, T. & Murphy, M. (1997). *College knowledge: Power, policy and the mature student experience at university*. Maynooth: MACE.
- Fleming, T., Loxley, A., Kenny, A. & Finnegan, F. (2010) Where next? A study of work and life experiences of mature students (incl. disadvantaged) in three higher education institutions. Dublin: Combat Poverty Agency.Retrieved 2nd February from www.pobal.ie
 - Flick, U. (1998). An Introduction to Qualitative Research. London: Sage.
- Foucault, M. (1980). *Power/knowledge: Selected interviews and other writings* 1972-1977. New York: Pantheon Books.
- Franklin, M. (1997). Making sense: Interviewing and narrative representation. In M. Gergen & S. N. Davis (Eds.), *Towards a new psychology of gender: A reader* (pp.99-116). New York: Routledge.
- Freire, P. (1970a). *Cultural action for freedom*. Harvard Educational Review Monograph Series No.1. Cambridge: Harvard University Press.
 - Freire, P. (1970b). Pedagogy of the oppressed. London: Penguin.
- Geaney, F. (1998). The development of further education in Ireland, *Irish Educational Studies*, 17(1), 51-64.
 - Gergen, K. (1999). An invitation to social construction. London: Sage.
- Giddens, A. (1991). *Modernity and self-identity: Self and society in the late modern age.* Cambridge: Polity Press.
- Gilligan, C. (1977). In a Different Voice: Women's conceptions of self and of morality *Harvard Educational Review*, 47(4), 481-517.
- Glaser, B. & Strauss, A. (1967). *The discovery of grounded theory*. Chicago: Aldine.
 - Glaser, B. G. (1978). *Theoretical sensitivity*. Mill Valley: Sociology Press.
- Glaser, B. G. (1992). *Basics of grounded theory analysis: Emergence v. forcing.* Mill Valley: Sociology Press.

- Giroux, H. (2004). *The terror of neo-liberalism*. Boulder, CO: Paradigm Publishers.
- Government for National Recovery 2011-2016. (2011). Retrieved June 2011 from http://www.labour.ie/policy/listing/129943050328508293.html
- Griffith, A. (1998). Insider / outsider: epistemological privilege and mothering work. *Human studies*, 21, 361-376.
- Grummell, B. (2007). The 'second chance' myth: Equality of opportunity in Irish adult education policies. *British Journal of Educational Studies*, 55(2), 182-201.
- Guiffrida, D. A. (2009). Theories of human development that enhance an understanding of the college transition process. *Teachers College Record* 111(10), 2419-2443.
- Gummesson, E. (2000). *Qualitative methods in management research*. Thousand Oaks: California: Sage.
- Habermas, J. (1969). The university in a democracy: Democratization of the university. In, Habermas: *Towards a Rational Society*, London: Heineman.
 - Habermas, J. (1969). Towards a rational society. London: Heineman.
 - Habermas, J. (1973). Theory and practice. Boston: Beacon Press.
 - Habermas, J. (1975). Legitimation crisis. Boston: Beacon Press.
- Habermas, J. (1984). The theory of communicative action. (1): Reason and the rationalization of Society. Boston: Beacon Press.
- Hanlon, G. (2000). Sacking the new Jerusalem? The new right, social democracy and professional identities. *Sociological Research Online*, 5(1). Retrieved from

http://www.socresonline.org.uk/5/1/hanlon.html

- Harré, R. & Gillett, G. (1994). The discursive mind. London: Sage.
- Harvey, D. (1989). The condition of postmodernity: An enquiry into the origins of cultural change. Oxford: Blackwell.
- Harvey, D. (2005). A brief history of neo-liberalism. Oxford: Oxford University Press.
- Havighurst, R. (1948). Cited in Sugarman, L. (2001). *Life-span development: frameworks, accounts and strategies.* Sussex: Psychology Press.
- Hellawell, D. (2006) inside-out: analysis of the insider-outsider concept as a heuristic device to develop reflexivity in students doing qualitative research. *Teaching in Higher Education*, 11, (4) 483-494.

- Heraty, N., Morley, M. & McCarthy, A. (2000). Vocational education and training in the republic of Ireland: Institutional reform and policy developments since the 1960s. *Journal of Vocational Education and Training*, 52(2), 177-199.
- Herzberg, F. (1966). Work and the nature of Man. New York: World Publishing.
- Hockey, J. (1993). Research methods researching peers and familiar settings. *Research Papers in Education*, 8 (2), 199-225
- Hodge, S. (2010). Trainers and transformation: Facilitating the 'dark side' of vocational training. *International Journal of Training Research*. 8(1), 53-62.
- Hogan, P. (2010). *The new significance of learning: Imagination's heartwork.* New York: Routledge.
- Honneth, A. (1995). The struggle for recognition: The moral grammar of social conflicts. Cambridge: Polity Press.
- Hopper, E. & Osborn, M. (1975). Adult students: Education selection and social control, London: Frances Pinter.
- Houle, C. (1961). *The Inquiring mind*. Madison: University of Wisconsin Press.
- Illeris, K. (2006). What is special about adult learning? In P. Sutherland, & J. Crowther (Eds.), *Lifelong Learning: Concepts and Contexts*. London: Routledge.
- Inglis, T. (1997). Empowerment and emancipation. *Adult Education Quarterly*, 48(1), 3-15.
- Inglis, T. & Murphy, M. (1999). No room for adults? A study of mature students in University College Dublin. Dublin: Social Science Research Centre and the Adult Education Office, UCD.
 - IVEA (2006). Irish vocational education association newsletter. Winter 2006.
- James, N. (1989). Emotional labour: Skill and work in the social regulation of feelings. *Sociological Review*, *37*, 15-42.
- Jarvis, P. (1995). Adult and continuing education: theory and practice. New York: Routledge.
- Jarvis, P. (2007). *Globalisation, lifelong learning and the learning society*. New York: Routledge.
- Jarvis, P. (2009). The EU and lifelong learning policy. In: P. Jarvis (Ed.), *The Routledge International Handbook of Lifelong Learning. London: Routledge*.

- Johnston, R. & Merrill, B. (2004). *Non-traditional students tracking changing learning identities in the inter-face with higher education*. Paper presented at SCUTREA conference 2004. 34th Annual Conference, University of Sheffield, UK. 6-8 July 2004.
- Johnstone, J. & Rivera, R. J. (1965). *Volunteers for learning: A study of the educational pursuits of adults*. Hawthorne, New York: Aldine.
 - Josselson, R. (1987). Finding herself. San Francisco: Jossey-Bass.
- Keane, E. (2009). Frictional relationships...tension in the camp: focusing on the relational in under-represented students' experiences in higher education. *Irish Educational Studies*. 28(2), 85-102.
- Kearney, M. H. (1998). Ready to wear: Discovering grounded formal theory. *Research in Nursing and Health*, 21(2) 179-186.
- Kincheloe, J. L. (2000). Making critical theory critical. In D. Weil & H. K. Anderson (Eds.), *Perspectives in critical thinking: Essays by teachers in theory and practice*. New York: Peter Lang.
- Knowles, M. (1998). Andragogy: An emerging technology for adult learning. In R. Edwards & A. Hanson (Eds.), *Boundaries of adult learning*. London: Routledge.
- Lennon, N. (1998). Post leaving certificate courses an alternative for who? *Irish Educational Studies*, 17(1), 353-362.
- The Lisbon Agenda (2000). Retrieved from www.consilium.europa.eu.uedocs/cms-data/docs/prssdata/en/ec/00100-rl.en0htm
 - Locke, K. (2001). Grounded theory in management research. London: Sage.
- Love, R. (1953). The use of motivation research to determine interest in adult college level training. *Educational Record.* 34, 210-218.
- Lowenthal, M., Thurnher, M. & Chiriboga, D. (1975). *Four Stages of Life*. London: Harper and Row.
- Loxley, A. and Seery, A. (2008) Some philosophical and other related issues of insider research, in Sikes, P. and Potts, A.(eds) *Researching education from the inside: Investigations from within.* Oxford: Routledge.
- Lynch, K. (1997). A profile of mature students in higher education and an analysis of equality issues. In R. Morris (Ed.), Mature students in higher education. Proceedings of conference in Athlone RTC. March 1996. Cork: Higher Education Equality Unit.
- Lynch, K. (2006a). Neo-liberalism and marketisation: The implications for higher education. *European Educational Research Journal*, *5*(1), 1-17.

- Lynch, K. (2006b). Neo-liberalism and Education. In S. Healy, B. Reynolds, & M. L. Collins (Eds.), *Social Policy in Ireland*. Dublin: The Liffey press.
- Marcia, J. (2010). Life transitions and stress in the context of psychosocial development. In T. W. Miller (Ed.), *Handbook of stressful transitions across the lifespan*. New York: Springer.
- Markus, H. & Nurius, P. (1985). Possible Selves. *American Psychologist*, 41(9), 954-969.
 - Mason, J. (2002). Qualitative researching. London: Sage.
- Mayo, E. (1933). The Human problems of an industrial civilization. New York: MacMillan.
- McCoy, S., Byrne, D., O'Connell, P. J., Kelly, E. & Doherty, C. (2010). *Hidden Disadvantage? A study on the low participation in higher education by the non-manual group.* ESRI. Retrieved from <a href="http://www.hea.ie/files/f
- McCoy, S. & Byrne, D. (2011). The sooner the better I could get out of there: Barriers to higher education access in Ireland. *Irish Education Studies*, 30(2), 141-157.
- Mc Cracken, G. (1988). The long interview. *Qualitative Research Methods*, 13. London: Sage.
- McGivney, V. (1991). Education's for other people: Access to education for non-participant adults. Leicester: NIACE.
- McGivney, V. (1998). Excluded men: Men who are missing from education and training. Leicester: MACE.
 - McIntyre, A. (1981). After virtue: London: Duckworth.
- McIver Report. (2003). Report of the Steering Group to the PLC Review established by the Department of Education and Science. Dublin: Government Stationery Office.
- Mercer, J. & Saunders, D. (2004). Accommodating change: the process of growth and development amongst a mature student population. *Research in Post-Compulsory Education*, 9(2), 283-300.
- Mercer, J. (2007a) The challenges of insider research in educational institutions: wielding a double-edged sword and resolving delicate dilemmas. *Oxford Review of Education*, 2007, 33 (1), 1-17.

- Mercer, J. (2007). Re-negotiating the self through educational development: mature students' experiences. *Research in Post-Compulsory Education*, 12(1), 19-32.
- Merriam, S. B. & Caffarella, R. *Learning in Adulthood: a comprehensive guide*. San Francisco: Jossey-Bass.
- Merrill, B. (2007). Recovering class and the collective in the stories of adult learners. In L. West, P. Alheit, S. A. Anderson & B. Merrill (Eds.). *Using biographical and life history approaches in the study of adult and lifelong learning: European perspectives.* Franfurt-am-Main: Peter Lang.
- Merrill, B. & West, L. (2009). *Using biographical methods in social research*. London: Sage.
- Mertens, D. M. (1998). Research methods in education and psychology: Integrating diversity with quantitative and qualitative approaches. California: Sage.
- Merton, R. (1972). Insiders and outsiders; a chapter in the sociology of knowledge. *American Journal of Sociology*, 78 (July), 9-47.
- Meyer, J. H. F & Land, R. (2003). Threshold concepts and troublesome knowledge: linkages to ways of thinking and practicing. In C. Rust (Ed.), *Improving student learning ten years on*. Oxford: SCSLD.
- Mezirow, J. (1978). Perspective transformation. *Adult Education*, 28(2), 100-110.
- Mezirow, J. (1981). A Critical theory of adult learning and education. *Adult Education*, 32(1) 33-24.
- Mezirow, J. (1995) Adult education and empowerment for individual and community Development. In B. Connolly, T. Fleming, D. McCormack & A. Ryan (Eds.), *Radical learning for liberation* 2. Dublin: MACE.
- Mezirow, J. (1998). On critical reflection, *Adult Education Quarterly*, 48 (3) 185-199.
- Mezirow, J. (2003). Transformative learning as discourse. *Journal of Transformative Education*, 1(1), 58-63.
- Mezirow, J. (2007). Adult education and empowerment for individual and community development. In *Radical learning for liberation* 2, 10-17 Maynooth MACE.
- Mills, J., Bonner, A. & Francis, K. (2006). Adopting a constructivist approach to grounded theory: Implications for research design. *International Journal of Nursing Practice*. 12, 8-13.

- Morse, J. M. (2001). Situating grounded theory within qualitative inquiry. In R. Schreiber, & P. Stern (Eds.), *Using grounded theory in nursing*. New York: Springer.
- Morstain, B. R. and Smart, J. C. (1974). Reasons for participation in adult education courses: A multivariate analysis of group differences. *Adult Education*, 24 (2), 83-98.
- Murphy, M. & Fleming, T. (2000). Between common and college knowledge: Exploring the boundaries between adult and higher education. National University of Ireland Maynooth, Ireland. *Studies in Continuing Education*, 22(1), 77-93.
- National Council for Vocational Awards, (NCVA). (1992). *Preparing for the new Europe: A national framework for vocational qualifications.* Dublin: NCVA
- National plan for equity of access to higher education 2008-2013.HEA Dublin: National Office for Equity of Access to Higher Education, HEA.
- The National Qualifications Authority of Ireland (NQAI).(2003a). *Qualifications matter: A brief guide to the national framework of qualifications*. Dublin: NQAI
- The National Qualifications Authority of Ireland (NQAI). (2003b). *National framework of qualifications: determinations for the outline framework of qualifications*. Document number 2003/5. Dublin: NQAI
- Neugarten, B & Neugarten, D. (1987). The changing meanings of age *Psychology*. *Today*, 21, 29-33.
- Neugarten, B. (1976). 'Adaptation and the life cycle' *Counseling Psychologist*. *6*, 16-20.
- Noddings, N. (1992). The challenge to care in schools: an alternative approach to education. New York: Teachers' College Press.
- O'Brien, S. & Brancaleone, D. (2011). Evaluating learning outcomes: in search of lost knowledge. *Irish Educational Studies* 30(1), 5-21.
- O'Reilly, B. (1989). Issues in the development of vocational education. *Administration*, 37(2), 152-170.
- O'Sullivan, D. (2005). *Cultural politics and Irish education since 1950s: Policy paradigms and power.* Dublin: IPA.
 - OECD (2006). Higher education in Ireland. Paris: OECD
- OECD. (2010). Learning for jobs: OECD Reviews of vocational education and training in Ireland. Paris: OECD

- Palmer, M., O'Kane, P. & Owens, M. (2009). Betwixt spaces: student accounts of turning point experiences in the first-year transition. *Studies in Higher Education*, 34(1), 37-54.
- Palmer, P. (1998). The courage to teach: Exploring the inner landscape of a teacher's life. San Francisco: Jossey-Bass
- Pring, R. (1995). Closing the gap: Liberal education and vocational preparation. London: Hodder & Stoughton.
- Pring, R. (1996). Values and education policy. In M. Halstead, & J. Taylor (Eds.), *Values in education and education in values*. London: Routledge.
- RANLHE (2008-2010). 'Access and Retention: the experience of non-traditional learners in Higher Education.' RANLHE.
- Reay, D., Ball, S. & David, M. (2002). 'It's taken me a long time but I'll get there in the end': mature students on access courses and higher education choice. *British Educational research Journal*, 28(1), 5-19.
- Reay, D., & Ball, S. (2005). *Degrees of choice: social class, race and gender in higher education.* Stoke-on-Trent: Trentham.
- Reid, A. (2005). Rethinking the democratic purpose of public schooling. In M. Apple, J. Kenway, M. Singh, & P. Lang (Eds.), *Globalizing education*. New York: Peter Lang.
- Richardson, J. T. E. (Ed.). (1999). *Handbook of qualitative research methods for psychology and the social sciences*. Leicester: British Psychological Society.
 - Robson, C. (2002). *Real world research* (2nd ed.). Oxford: Blackwell.
- Rooney, P. (2005). Researching from the inside does it compromise validity? A discussion. *Level 3. (3)* Retrieved from: www.level3.dit.ie/html/issue3/rooney/rooney1.htm
- Rubin, L. (1976). Worlds of pain: Life in the working-class family. New York: Basic Books.
- Ryan, A. B. (2004). Subjectivity and consumption: concerns for radical adult education. In A. Ryan & T. Walsh (Eds.), *Unsettling the horses: Interrogating adult education perspectives* (pp. 137-160). Maynooth: MACE.
- Ryan, A. B. (2006). Methodology: Collecting data. In M. Antonesa, H. Fallon, A. B. Ryan, A. Ryan, T. Walsh, & L. Borys (Eds.), *Researching and writing your thesis: A guide for post graduate students* (pp. 70-89). Maynooth: MACE.
 - Santrock, J. W. (2009). *Life-span development*. New York: McGraw Hill.

- Sheehy, G. (1996). *New passages: mapping your life across time*. London: Harper Collins.
- Sikes, P. and Potts, A.(eds) *Researching education from the inside: Investigations from within.* Oxford: Routledge.
- Simpson, B. (1996). 'Narration and transition: Experiences of mature students in higher education'. In *Auto/Biography*, 4 (2/3), 1-11.
- Stewart, J. (2008). *The VEC sector: Our track record in further education and training.* National Adult Education Conference, Westport.
- Strauss, A. & Corbin, J. (2008). *Basics of qualitative research: Grounded theory procedures and techniques*. California: Sage.
- Strauss, A. & Corbin, J. (1994). Grounded theory methodology: An overview. In N. K. Denzin, N. & Y. S. Lincoln (Eds.), *Handbook of qualitative research*. California: Sage.
- Student Support Bill, (2008). Retrieved April 2010 from www.oireachtas.ie/documents/bills28/bills/2008/0608/b608d.
- Styles, J. (1979). Outsider / insider: researching gay baths. *Urban Life: A journal of ethnographic research*, (8) 135-153.
- Sugarman, L. (2001). *Lifespan development, frameworks, accounts and strategies*. New York: Psychology Press.
- Tarule, J. (1996). Voices in dialogue: Collaborative ways of knowing. In N. Goldgerger, J. Tarule, B. Clinchy, M. Belenky (Eds.), *Knowledge, difference, and power*. New York: Basic Books.
- Tennant, M. (1991). The psychology of adult teaching and learning. In J. Peters, & P. Jarvis. (Eds.), *Adult education, evolution and achievement in a developing field of study*. San Francisco: Jossey-Bass.
 - Tennant, M. (1988). Psychology and adult learning. London: Routledge.
- Tennant, M & Pogson, P. (1995). Learning and change in the adult years: a developmental perspective. San Francisco: Jossey-Bass.
- Tett, L. (2004). Mature Working Class students in an elite university. *Studies in the education of adults*. Autumn 2004, 36 (2), 252-264.
- Thomas, L., & Quinn, J. (2007). First generation entry into higher education. OUP.
- Tierney, L. & Clarke, M. (2007). The European qualifications framework: Challenges and implications in the Irish further education and training sector. *European Journal of Vocational Training*, 42/43, 129-142

- Tinto, V. (1993) *Leaving college: Rethinking the causes and cures of student attrition* (2nd ed.). Chicago/London: University of Chicago Press.
- UCAS, (2004). Applicants to higher education through UCAS with Access qualifications, 2004 entry. Retrieved on 6th June 2010 from www.ucas.ac.uk/ucc/access/ewnin/questions.html
 - Irish Government. (1997). Universities Act. Retrieved from http://www.irishstatutebook.ie/1997/en/act/pub/0024/index.html.
- van Vught, F. (1994). *International developments in assuring quality in higher education*. A. Craft (Ed.). London: Falmer Press.
- Waller, R. (2006). 'I don't feel like 'a student', I feel like 'me': The over-simplification of mature learners' experience(s). Research in post-compulsory Education 11(1), 115-130.
- Walters, M. (2000). The mature students' three Rs. *British Journal of Guidance and Counselling*, 28(2), 267-278.
- Walters, S., Borg, C., Mayo, P. & Foley, G. (2004). Economics, politics and adult education. In Foley, G (Ed.), *Dimensions of adult learning*. Oxford: OUP.
- Warmington, P. (2002). Studenthood as surrogate occupation: Access to higher education students' discursive production of commitment, maturity and peer support. *Journal of Vocational Education and Training*, 54, 583-600.
- Watson, D., McCoy, S. & Gorby, S. (2006). The post leaving certificate sector in Ireland: A multivariate analysis of educational and employment outcomes. Dublin: ESRI and Department of Education and Science.
- Webb, J., Schirato, T. & Danaher, G. (2002). *Understanding Bourdieu*. London: Sage London.
- Weil, S.W. (1986). Non-traditional learners within traditional higher education institutions: Discovery and disappointment. *Studies in Higher Education*, 2(3), 219-235.
- West L. (1996). *Beyond fragments: Adults, motivation and higher education,* London: Taylor and Francis.
- West, L. & Holliday, M. (2010). Structure and subjectivity, reproduction and transformation: *Understanding non-traditional students in higher education; a trans-European approach*. RANLHE.
- Wheelehan, L. (2009). The limits of competency-based training and the implications for work. In J. Field, J. Gallagher & R. Ingram (Eds.), *Researching transitions in lifelong learning*. London: Routledge.

Whitehead, J. (1984). Motives for higher education: A study of intrinsic and extrinsic motivation in relation to academic achievement. *Cambridge Journal of Education*, 14, 26-34.

Winnicott, D. (1971). Playing and reality. London: Routledge.

Yorke, M. (2004) Retention, persistence and success in on-campus higher education, and their enhancement in open and distance learning *Open Learning*, *Volume 19*(1), 19-32.

doi:10.1080/0268051042000177827

Young, P. (2000). 'I might as well give up': Self-esteem and mature students' feelings about feedback on assignments. *Journal of Further and Higher Education*, 24(3), 409 - 422.

APPENDICES

		Page number
1	Pilot Questionnaire	193
2	Questionnaire	196
3	Interview schedule	199
4	Information for participants	200
5	Consent form	201
6	Sample memos	202

Pilot Questionnaire

This questionnaire is part of a research project on mature students in further Education. Thank you for agreeing to take part.

1.	Age
2.	Male Female (please tick)
3.	Nationality
4.	Marital status (please tick)
	Single Widowed Married Separated Divorced
5.	Do you have children? Yes No
	Please indicate the number of children and their ages:
	Age Under 5 years old Between 5 and 10 years Between 11 and 15 years Between 16 and 20 years 21 years or older Number of children
6.	At what age did you leave secondary school?
7.	Please tick which, if any, of the programmes below you completed. The Leaving Certificate The Leaving certificate Applied Programme The Leaving Certificate Vocational Programme Other final secondary education qualification Please give details None
8.	Are you doing a one or two year course in further education? Please tick
	One year Two year
9.	If you are doing a two year course, please indicate what year you are now ir First year Second year
10.	Please list any other courses you have undertaken since leaving secondary school.

What is your mother's highest educational qualification?	
What is your father's highest educational qualification?	
What is your father's occupation?	
What is your mother's occupation?	
Did any of your siblings attend further or higher education? Please tick	
Yes No Not applicable	
When you finish your present course do you intend to go on to If yes, please say which course. Yes	another course?
Between the time you left secondary school and came to further what was your main occupation? (If you had several different occupations, please write 'various' and indicate the type of work manual, shop work etc.)	
Please indicate how you support yourself financially while at copart-time job, parental support, spousal support, VTOS etc.)	ollege (e.g.
Why did you come back to education?	
Why did you come to further education rather than another type university?	e of course e.g.

	ducation?
	you consider to be the least enjoyable aspects of being a student in ducation?
	how would you summarise your experience of being a mature studen reducation?
project	ou for taking the time to fill in this questionnaire. The research will also focus, through interviews, on the experiences of maturin further education. Would you be willing to take part in a
	_ No
If 'yes' ¡	please give a contact number or email address.
	_

Questionnaire

This questionnaire is part of a research project on mature students in further Education. Thank you for agreeing to take part.

1.	Age			
2.	Male Female (please tick)			
3.	Nationality			
4.	Marital status (please tick)			
	Co-habiting Widowed Married Separated Divorced Single			
5.	Do you have children? Yes No			
	Please indicate the number of children and their ages:			
	Age Number of children Under 5 years old Between 5 and 10 years Between 11 and 15 years Between 16 and 20 years 21 years or older Number of children			
6.	At what age did you leave school?			
12.	Please tick which, if any, of the programmes below you completed. The Leaving Certificate The Leaving certificate Applied Programme Junior Cert The Leaving Certificate Vocational Programme Other (please give details)			
	None			
13.	Are you doing a one or two year course in further education? Please tick One year Two year			
14.	If you are doing a two year course, please indicate what year you are now in First year Second year			

W	hat is your mother's highest educational qualification?
V	What is your father's highest educational qualification?
W	That is your father's occupation?
W	That is your mother's occupation?
	id any of your siblings attend further or higher education? ease tick
Y	es No Not applicable
If	Then you finish your present course do you intend to go on to another course, please say which course.
1	eses.
	es o
No Wi	
No Wi oc mo	etween the time you left secondary school and came to further education, that was your main occupation? (If you had several different ecupations, please write 'various' and indicate the type of work, e.g.
Bo windown ————————————————————————————————————	etween the time you left secondary school and came to further education, hat was your main occupation? (If you had several different ecupations, please write 'various' and indicate the type of work, e.g. anual, shop work etc.) ease indicate how you support yourself financially while at college (e.g.

28. Why did you come to this college specifically?

Its reputation A friend recommended it Another reason (please give detail	Career Guidance advice It's convenient to where I live
Thiomer reason (prease give actual	
What do you consider to be the most engurther education?	joyable aspects of being a student in
The course	
The social aspect	
Pleasant atmosphere	
Approachable staff	
Other – (please give details)	
What do you consider to be the least enj further education?	oyable aspects of being a student in
Assignments	
Facilities	
Lack of money	
Juggling other commitments	
Younger students	
Other – (please give details)	
Overall, how would you summarise you in further education?	ar experience of being a mature student
Thank you for taking the time to fill project will also focus, through interstudents in further education. Would interview?	views, on the experiences of mature
Yes No	
If 'yes' please give a contact number or	email address.
	 Thank you

Interview schedule

How did you come to be a student here in the college?

How did you feel about coming back to education?

How do you feel about it now?

What was your experience of school?

Has it affected you life? In what ways – positive, negative? – Family, friends?

What would be a good day in college for you?

What have been the best / worst aspects of being a student?

What do you like most / least about being a student?

What do you think of the younger students? How do you get on with them?

What could be done to help mature students in the college?

How did you find the academic demands? Studying? Time management? Assignments?

What did you get out of being a student?

What would you be doing if you weren't a student now?

What do you hope to do when you finish here?

What advice would you give to someone thinking of coming back to college?

Is there anything else that you'd like to say about being a mature student?

Is there anything you'd like to ask me?

Thank you very much for talking to me.

INFORMATION FOR PARTICIPANTS

Research Project on Mature Students in Further Education

I want to find out about the experiences of adults who have returned to education

in a Further Education college. The aim of the study is to understand why mature

students come to an FE college, and how they get on in further education. My

report will be part of a thesis for a Doctorate in Education in NUI Maynooth.

Interview

If you agree to take part I will meet you for a one-to-one interview, where we will

discuss your background and your experience of further education. The interview

will be recorded and transcribed. I would like to meet you a second time to give

you a copy of the interview and to get your opinion on the points I have taken

from it. The first interview will take about an hour to an hour and a half, the

second will be shorter. If you decide to take part, you are free to change your

mind and withdraw from the research project at any time, even after the interview.

During the interview you do not have to answer question if you do not want to.

Anonymity

In any reports or publications resulting from this study your identity will remain

anonymous. Your name and anything else that might identify you will be changed.

Your name will not be used or disclosed to anyone.

If you need to speak to somebody during or after the study or if you have any

concerns or questions regarding this study please feel free to contact me at any

stage.

Fergal Hardiman

The College

200

Participant consent form

Please tick your response.

 I have read and understood the Participant Information. Yes No
I have had the opportunity to ask questions and discuss the study Yes No
 I have received satisfactory answers to all my questions. Yes No
 I have received enough information about this study Yes No
I agree to take part in the study.YesNo
6. I understand that I am free to withdraw from the study at any time. Yes No
 I agree to have my interview tape-recorded. Yes No
Participant's Signature:
Researcher's Signature:

Memo

Some thoughts after initial interviews

Redundancy – unemployment- loss of role –mourning? Loss of contacts, friends, daily contact – more of a problem for men it seems – Daniel, James – 'the silence – I hate that.' -Important social aspect of FE, also importance of informal chats, coffee, work groups, - MS stay together? Being known – in the college.

What's this – enjoying what? Social ? achievement? doing well? Computers- more on this** College – James loves coming in! college as a social outlet ? belonging, achievement. All are enjoying it - surprised to be enjoying it so much. –they're respected, part of something, working with people.

Loss of role, job, children? Identity? Change of role? Who am I now? – I'm a student – Daniel, College as a job? – Warmington, - Daniel? Need to change –

Need to retrain, get 'something for the CV' Vera, Daniel, James – why FE? Other options? – knowing someone, recommendation, level and structure of course, FE.

In some cases coming to college was an afterthought – recommended by someone else, but the idea was not unacceptable? Why? – previous experience of ed? - ** most seemed to like school (mostly)

All appreciate the support and this seems to be a key factor – personal contact with teachers, tutors, Friendly atmosphere. Feeling understood- even in general classes. Respect, belonging. – very important.

Feeling terrified – not all - Abbey, Maeve- used to education, - Abbey in education all her life- never left-

A major issue for others? – fish out of water?

Assignments – fear, not knowing what was expected, doing well, confidence.

It seems that although unemployment is often the trigger to come back to education, it is also seen, at least in retrospect, as an opportunity. Everyone seems very happy to have come back and wouldn't have done it if the job hadn't ended. Some (Brenda, Caroline) always wanted to come back. The opportunity came for them and they took it. They felt they missed out by not continuing in education. A vague feeling that their lives could have been different (James) or can be different. So far all are enjoying it.

Confidence coming up as a major thing with all. 'more confident' 'it's changed me.' What's changed them? – the subject matter? Knowing more? Not feeling inadequate because of lack of education? - more on this.

What is confidence? How has it changed them? Is this a question only for mature students? What about younger ones?

Memo after 3 interviews

James	Caroline	Vera
Job finished Local to the college –	Job finished Local	Job finished Always wanted to go back
Obair connection.		Wanted a better job
Always wanted to go back	Left school for a job	Needed a qualification
(Dinishing advection)	Good at school	Good at school, left to get
'Finishing education' 'coming full circle'	Felt she had 'missed out' Always wanted to come	a job No expectations of staying
'didn't get a chance'	back	in school or progressing at
Importance of 'finishing'		the time
	Needed a qualification	Wanted more education –
Good in school	Kids older – 'time was	to be better informed.
Family encouraging	right'	N 1' '. C .1 CN
Confidence	Family ancouraging	Needing it for the CV
Recognition	Family encouraging	Not being 'on a par'
Pride	Better being older	Wanting to 'do something
	0 1 11 1100 1	with it'
'the real me'	Seeing things differently	One stan at a time
Surprised it's gone so well	More empathy More socially aware	One step at a time.
Aware of possibilities now	whole socially aware	Enjoying the course very
– progression – IT	Confidence	much
	Being able to speak	
One step at a time	'opened my eyes'	First assignment stressful
Maating paople	'makes you a better	but pleased at result and
Meeting people Social interaction	person'	won ability.
Support from staff	Great support from staff	Being a better
Better as a mature student	Age balance important	conversationalist – being
Enjoying it so much		able to have conversations.
N. C. 1	First assignment – hard	3.6 10 .1
More confident	Pleased at doing well – milestone	More self worth.
The real me.	illiestone	Enjoying it. Feeling a bit guilty.
Language of first	Benefits for the kids	Appreciating it more as a
assignment a problem –		mature student. Being
Better now.		more focused.
		Importance of social
		aspect.
		Surprised at enthusiasm for education.
		Wish I'd done it years ago.
		Seeing things differently

TABLE OF FIGURES

4.1	Initial Coding Process	Page	77
4.2	Focused Coding	Page	78
4.3	Theoretical Coding	Page	80
5.1	Father's and mother's occupations	Page	87
5.2	Interview participants	Page	96
6.1	Initial and focused coding	Page	112

ACRONMYS

APL Accreditation of prior learning
BTEA Back to Education Allowance

CAO Central Applications Office

CDVEC City of Dublin Vocational Education Committee

CSO Central Statistics Office

DES Department of Education and Skills

FE Further Education

FETAC Further Education and Training Awards Council

GATS General Agreement on Trade and Services

HE Higher Education

HEI Higher Education Institutes

HETAC Higher Education and Training Awards Council

IT Institute of Technology

LC Leaving Certificate

LCA Leaving Certificate Applied ProgrammeNCVA National Council for Vocational AwardsNFQ National Framework of Qualifications

NQAI National Qualifications Authority of Ireland

PLC Post Leaving Certificate Courses

UCAS Universities and Colleges Admissions Service

VET Vocational Education and Training

VTOS Vocational Training Opportunities Scheme

WTO World Trade Organisation